energy&fuels

Recent Advances in Machine Learning Research for Nanofluid-Based Heat Transfer in Renewable Energy System

Prabhakar Sharma,* Zafar Said,* Anurag Kumar, Sandro Nižetić, Ashok Pandey, Anh Tuan Hoang,* Zuohua Huang, Asif Afzal, Changhe Li, Anh Tuan Le, Xuan Phuong Nguyen,* and Viet Dung Tran

Cite This: Energy Fu	els 2022, 36, 6626–6658	Read (Online	
ACCESS	III Metrics & More		E Article Recommendations	

ABSTRACT: Nanofluids have gained significant popularity in the field of sustainable and renewable energy systems. The heat transfer capacity of the working fluid has a huge impact on the efficiency of the renewable energy system. The addition of a small amount of high thermal conductivity solid nanoparticles to a base fluid improves heat transfer. Even though a large amount of research data is available in the literature, some results are contradictory. Many influencing factors, as well as nonlinearity and refutations, make nanofluid research highly challenging and obstruct its potentially valuable uses. On the other hand, data-driven machine learning techniques would be very useful in nanofluid research for forecasting thermophysical features and heat transfer rate, identifying the most influential factors, and assessing the efficiencies of different renewable energy systems. The primary aim of this review study is to look at the features and applications of different machine learning techniques employed in the nanofluid-based renewable energy system, as well as to reveal new developments in machine learning research. A variety of modern machine learning algorithms for nanofluid-based heat transfer studies in renewable and sustainable energy systems are examined, along with their



advantages and disadvantages. Artificial neural networks-based model prediction using contemporary commercial software is simple to develop and the most popular. The prognostic capacity may be further improved by combining a marine predator algorithm, genetic algorithm, swarm intelligence optimization, and other intelligent optimization approaches. In addition to the well-known neural networks and fuzzy- and gene-based machine learning techniques, newer ensemble machine learning techniques such as Boosted regression techniques, K-means, K-nearest neighbor (KNN), CatBoost, and XGBoost are gaining popularity due to their improved architectures and adaptabilities to diverse data types. The regularly used neural networks and fuzzy-based algorithms are mostly black-box methods, with the user having little or no understanding of how they function. This is the reason for concern, and ethical artificial intelligence is required.

1. INTRODUCTION

1.1. Background. The energy consumption and greenhouse gas (GHG) emissions in the world have been growing dramatically in recent decades because of the global population rise, as well as the sharply increased industry and transportation activities.^{1,2} As illustrated in Figure 1, the global GHG emissions can be seen to roughly originate from four main categories, including energy, agriculture, industry, and waste.3 However, energy consumption has been considered as the primary source of GHG emissions with around a 73.2% share,⁴ showing that extensive use of fossil-based energy sources caused unfavorable impacts on the living environment because of the high level of GHG emissions.⁵ Due to this reason, the development of renewable energies has been paid much attention in recent years due to their beneficial properties;^{6,7} this could be observed more clearly during the post-COVID19 pandemic.⁸

In the development progress of energy systems, there has been a large interest in working fluids since they play a critical role in thermal efficiency and heat transfer capacity.^{9,10} Due to their weak heat transfer characteristics, ethylene glycol (EG), water, and other common working fluids cannot provide sufficient heat transfer efficiencies in industrial operations. The ability of conventional fluids to transfer heat can be enhanced by suspending solid particles in them.¹¹ Over the years, researchers have focused their efforts on modifying heat transfer fluids using small-sized dispersed solid particles in the fluid. Indeed, using nanofluids could enhance the overall thermal efficiency significantly, and the total efficiency could be increased by 25% for nanofluids compared to the case of pure water.¹² However, these fluids experienced several issues,

Received:April 4, 2022Revised:May 26, 2022Published:June 13, 2022





Figure 1. Global greenhouse gas emissions by sector 3 (reuse with permission of Elsevier via LN 5315110697148).

including particle swarming, channel blockage, transmission device degradation, increased pressure drop, and finally alluviation.¹³ Particle settling reduced thermal conductivity and created sludge deposits, increased thermal resistance, and decreased the fluid's heat transfer capacity.¹⁴ By 1881, James Clerk Maxwell had produced a significant revolution in heat transmission in fluids by introducing the notion of utilizing nanosized particles for the first time. He presented a new perspective on the liquid–solid suspension using nanosized particles. Okonkwo et al.¹⁵ initially coined the term "nanofluid" to designate a fluid containing suspended particles, and Eshgarf et al. further refined the notion at the Argonne National Laboratory in the United States.¹⁶

These newer generations of heat transfer fluids, known as nanofluids (NFs), were invented based on this concept and breakthroughs in nanotechnology. The NFs are a blend of solid nanoscale materials (metal oxides, metals, nonmetals, and other nanoparticles) and a fluid known as the base fluid (like ethylene glycol, oil, or water).^{17,18} NFs have emerged as a viable alternative to the currently utilized heat transfer fluids due to their superior heat transfer capabilities. NFs have better thermal conductivities and thermophysical characteristics.¹⁹ That is why NFs are preferred for use in thermal management systems and heat transfer devices.²⁰ At present NFs are being used in solar collectors,²¹ hydroelectric rotors,²² geothermal heat exchangers,^{23,24} wind turbines,²⁵ fuel cells,^{26,27} chemical process plants,²⁸ and photovoltaic/thermal (PV/T) systems.^{29,30}

The addition of nanosized particles to base fluids has a greater impact on the thermophysical characteristics of nanofluids. For example, in an experimental investigation by Moldoveanu et al.,³¹ they used hybrid NF to enhance the thermal conductivity of NF by 19.2% in comparison with base fluid (water). Wole-Osho et al.³² explored the effects of particle concentrations on the dynamic viscosity and specific heat of an Al₂O₃-ZnO-H₂O hybrid NF. In the case of the 2:1 ratio, viscosity is enhanced by 96.37%, while specific heat decreases by 30.12% at 25 °C temperature. Xian et al.³³ explored the influences of ultrasonication time and a variety of surfactants on the thermophysical properties of NFs. The experimental study used a hybrid NF of graphene nanoplatelets

with titanium dioxide. The nanoparticles were dispersed in a mixture of EG and water. It was experienced that the thermal conductivity was enhanced up to 23.74%. Kakavandi and Akbari³⁴ reported an enhancement of thermal conductivity up to 33% with hybrid NFs—binary base fluids (MWCNTs-SiC/Water-EG). Several other studies reported the impacts of hybrid NFs in improving thermophysical properties to make heat transfer fluids more suitable for energy systems.^{35–38} In recent years, a novel class of working fluids has been identified and actively investigated, consisting of three solid nanoparticles scattered in a typical fluid. These fluids are known as ternary nanofluids, ternary hybrid nanofluids, or trihybrid nanofluids.^{39,40}

Nanofluids are utilized in sophisticated heat exchange processes and energy systems due to their better thermophysical properties than base fluids. They have nonlinear impacts on fluid flow, heat transfer, optics, stability, and radiative performances. Mapping their effects on complex thermal processes and the effects of different mixing ratios is cumbersome.⁴¹ As a result, artificial intelligence-based machine learning (ML) might be used in nanofluids research. The advent of improved computational facilities with advanced machine learning algorithms has made it possible to predict the thermophysical properties of hybrid NFs.⁴²

1.2. Renewable Energy. Energy is critical to industrialization, economic progress, education, and urbanization in today's world. As per the US EIA briefing, worldwide energy consumption is increasing at a rate of about 2.3% each year. According to the report, petroleum will be in limited supply throughout the planet shortly.⁴³ Fossil fuels are also a major source of environmental contamination, such as air and water pollutions. As a result, renewable sources of energy such as biomass, solar, geothermal, wind, and others have received considerable attention as prospective energy development options.^{44,45} Environmental limitations are frequently violated in the existing global energy system. The basic energy requirements are not fulfilled for the masses, and energy justice is not guaranteed for everyone. In essence, the energy system is unsustainable, and it is on a path to stay unsustainable in the future. This realization spurred the United Nations to establish 17 sustainable development goals (SDGs), one of which (SDG 7) envisions a future where everyone has access to cheap, dependable, sustainable, and modern energy services by 2030.46,47 SDG 7 is not the only goal for which sustainable energy is important. The development of renewable, clean, and sustainable energy sources is important to attaining the United Nations' sustainable development goals (SDGs),48 even though developing energy systems with renewable sources is the first step toward achieving global environmental sustainability in many industrialized economies.

The studies in the past point out that meeting the global energy demands sustainably has been a critical concern for humanity.⁴⁹ As a result, to maximize their sustainability, optimal design, and economic viability, a thorough understanding of sustainable energy systems is required. Sustainable energy systems have previously been characterized as explicit domains that employ energy sources projected to decline in a time frame relevant to the human race. Significant research efforts have been undertaken to enhance renewable energy generation to meet the global target of reducing energy consumption and CO₂ emissions by 20% by the year 2030.⁵⁰ Furthermore, the massive rise in energy consumption in recent

years as a result of fast industrial development and population expansion confirms that developing renewable, clean, and sustainable energy source Research Center in Systems Ecology and Sustainability (RCSES) is now a key priority in many nations across the world,⁵¹ as seen from the creation of the Research Center in Systems Ecology and Sustainability (RCSES). As a result, not only RCSES are developed for achieving SDGs but also their behaviors and performances under various operating situations must be understood.

Figure 2 illustrates the statistics on the number of published research articles on nanofluids (NF), machine learning (ML),



Figure 2. Number of research articles published in renewable energy, nanofluids, machine learning, and their combinations.

and renewable energy (RE) and also a combination of these three domains obtained from keyword searches on the Web of Science on February 3, 2022, for the decade 2012-2021. The data on NF came from a subject search with the key phrases "nanofluid" or "nanofluids". The analysis data on ML came from a variety of searches that included the terms "adaptive neuro-fuzzy", "artificial neural network", "category and regression tree", "artificial intelligence", "support vector machine", and "random forest". The data of published articles on renewable energy were searched using the terms "renewable" and "renewables". The graph shows that publications in these areas have grown considerably in recent years, with renewable energy having the most articles, followed by machine learning and nanofluids. However, nanofluid publications utilizing machine learning in the renewable energy domain (i.e., a combination of these three search queries) are scarce, gradually increasing from a two-digit yearly figure in the period 2012-2013 to approximately 1468 in 2021, accounting for approximately 1.7% of renewable energy research publications in that year. As a result, the nanofluid research community must focus more on machine learning research in the renewable energy sector.

1.3. Novelty of the Review Study. Machine learning technologies based on AI are being utilized to model business and engineering difficulties in virtually every industry. Most energy-producing industries in the renewable energy arena use some form of heat transfer. Numerous research studies indicate the efficacy of nanofluids in boosting the heat transfer rate. The selection of an acceptable nanofluid for a specific renewable energy application, on the other hand, is a crucial challenge. The selection of a suitable nanofluid is based on an assessment

of its thermophysical properties via a thorough characterization technique.⁵² The repeated comprehensive characterization of multiple nanofluids is time consuming and labor intensive and requires a significant investment in specialized testing equipment and raw materials.⁵³ Machine learning approaches are a viable answers to these challenges in nanofluid characterization. Modern machine learning algorithms can effectively construct a prediction model from experimental data, which may then be used to predict features in future investigations. Table 1 highlights key review articles published in the previous five years on the use of machine learning techniques in the use of nanofluids in the renewable energy field.

Table 1. Summary of Review Papers on Use of Machine
Learning Techniques for Utilization of Nanofluids in
Renewable Energy Domain in Last Five Years

Reference	Focus area	Year
Akhter et al. ⁵⁴	Application of machine learning in photovoltaic power generation	2019
Mosavi et al. ⁵⁵	Applications of machine learning system in general energy systems	2019
Ramezanizadeh et al. ⁵⁶	Utilization of machine learning for estimation of nanofluid dynamic viscosity	2019
Diogo et al. ⁵⁷	Machine learning for energy efficiency in industry	2020
Shateri et al. ⁵⁸	Machine learning for nanofluid viscosity estimations	2020
Antonopoulos et al. ⁵⁹	Artificial intelligence in demand-side response of energy	2020
Fathi et al. ⁶⁰	Machine learning in performance forecasting of urban building	2020
Ahmad et al. ⁶¹	Artificial intelligence in sustainable industry	2021
Ma et al. ⁶²	Nanofluids heat transfer applied to renewable energy	2021
Jamei et al. ⁶³	Evaluations of data analysis techniques for estimations of nanofluid specific heat	2021
Aghbhaslo et al. ⁶⁴	Machine learning techniques for biodiesel research	2021
Hoang et al. ⁶⁵	Artificial neural network for behaviors of diesel engine fueled with biodiesel	2021
Wang et al. ⁶⁶	Machine learning methods applied for use of nanofluid in a heat pipe	2021
Adun et al. ⁶⁷	Evaluations of machine learning techniques for forecasting of nanofluid specific heat	2021

An examination of review articles published in the previous five years (2017–2022) revealed that while numerous review studies on the thermophysical properties of nanofluids are available in the open literature only a few on the use of nanofluids in conventional energy systems or demand forecasting are available. Even though machine learning methods are rapidly being employed in nanofluids and renewable energy, no comprehensive research evaluation on machine learning applications in nanofluid-based renewable energy systems exists. The present work seeks to connect the dots to provide a comprehensive analysis of the connected research domains of machine learning, nanofluids, and renewable energy. In Section 2, the review article outlines the approach used throughout the review of the relevant literature. Section 3 provides a detailed analysis of different machine learning approaches, while Section 4 provides a detailed examination of the use of machine learning techniques for modeling and forecasting thermophysical characteristics of nanofluids. Section 6 discusses the use of machine learning approaches in the field of renewable energy. Section 7 discusses the problems and opportunities of employing machine learning techniques in nanofluid-based renewable energy systems, followed by the study's key findings.

2. METHODOLOGY

The present study examines the most recent research on the use of machine learning methods in nanofluid heat transfer. Over the past decade, there has been a surge in research papers on the use of machine learning methods.⁶⁸⁻⁷¹ In the past, a variety of machine learning approaches have been employed to model and forecast the properties of nanofluids.^{41,68,72} With the advent of newer machine learning methods, the everincreasing computational power is the catalyst for the increasing popularity of machine learning techniques.^{73,74} As a result, the current study attempts to compile a comprehensive report including the most recent developments in this subject for easy reference. Because these strategies are constantly changing, examining and portraying just the most recent work is critical. As a result, the research papers over the past three years were examined and a few notable works from prior years. The study's data was gathered from Scopus, Google Scholar, Web of Science, and the Google search engine. Journals referred to included Journal of Cleaner Production, Energy & Fuels, Energy Conversion and Management, Renewable and Sustainable Energy Reviews, Energy, Powder Technology, Journal of Molecular Liquids, International Journal of Heat and Mass Transfer, Fuel, and Journal of Thermal Analysis and Calorimetry. The research studies published in the field of machine learning-based model prediction of engineering applications were examined. The research papers were divided into categories using machine learning methods. Each approach's efficacy was then evaluated in terms of model prediction quality. The subsequent phase investigated the machine learning methodologies utilized for model prediction of thermophysical properties of nanofluids. The literature on a comparative analysis of several machine learning methodologies was beneficial in deciding which techniques were best suited to the nanofluid domain. The methodologies used in machine learning to develop prognostic models for thermal conductivity, density, specific heat, and viscosity were examined. The study was broadened to include the use of machine learning methods for renewable energy sources to present a more complete picture for future researchers.

3. MOST RELEVANT MACHINE LEARNING TECHNIQUES

3.1. Multilayer Perceptron Artificial Neural Network. Two types of studies, experimental and analytical, are utilized to assess the performance of engineering systems in the research field. As the process of achieving satisfactory results may necessitate multiple lab-based trials, the experimental studies become more expensive as they need extended manhours, material resources, and energy.^{75,76} Most engineering systems are becoming more complex with time, so mathematical modeling with the first principle is increasingly becoming complex. Soft computing techniques are useful in this domain to save costs.⁷⁷ In the field of artificial intelligence, the artificial neural network (ANN) is a useful technique for model prediction, because it saves time and yields more precise findings than other approaches.^{78,79} The ANN technique has

risen in popularity in the engineering and energy industries during the past decade. ANN is a simple, easy-to-use, fast computing technique capable of solving complex nonlinear problems.^{56,80,81} The only drawback is the correct design of the network and the requirement of a larger data set for modeling.

The most fundamental neural network architecture is the multilayer perceptron (MLP). Because all neurons in each layer have forward connections to all neurons in the layer above, it is also known as a feed-forward neural network. As illustrated in Figure 3, the MLP-ANN approach comprises



Figure 3. Line diagram depicting MLP-ANN.

three layers: the first input layer, which is used to get input data; certain interlayers known as hidden layers; and the final layer, or output layer, which reveals the ANN's projected outcomes. Every node is allotted a weight vector to link with the following layers' nodes. Every node present in the MLP-ANN model can receive inputs, process them, and provide an output.^{56,68} In a combined form, the input layer's nodes act as input for the next (hidden) layer's nodes. In general, this method attempts to iterate ML on time series and nonlinear data like that of the human brain. Assume that the input vector is "X" such that

$$X = [x_{1,}x_{2,}x_{3}, x_{4}, \dots, x_{n}]^{K}$$
(1)

If n_j is taken as the input for the *j*th node in the middle layer (hidden), then

$$n_j = \sum_{i=1}^n \omega_{ij} x_i + j, \text{ where } j = 1, 2, 3, \cdots, T$$
(2)

$$y_j = f(n_j) = f\left(\sum_{i=1}^n \omega_{ij} x_i + \theta_j\right)$$
, where $j = 1, 2, 3, \dots, T$

(3)

where K, ω_{ij} , and f denote the node indexing in the hidden layer, weight between the *i*th input node and *j*th output node, and the transfer function, respectively.

Each function has its own set of features and may be used in a variety of different engineering problems. The output node is formed by multiplying the output of every hidden node by the output linking weight of that node. The counts of hidden neurons in the input and output layers in the MLP-ANN model are governed by the number of variables in the examined application. However, there is no proven approach for estimating the number or sizes of hidden layers.^{82,83} The number of hidden layers can be modified depending on the scale of the problem, the quality, and the quantity of data. As a result, neurons are regularly adjusted depending on iterations during the training to achieve the optimal amount.^{84,85} The construction of a predictive MLP-ANN model necessitates the establishment of a training stage. When multiple input and output groups are supplied to build a network, the prediction process regulates weight and bias values. Backpropagation (BP) is a training method widely used in the training mode of MLP-ANNs to manage bias and weight values.^{86,87}

3.2. Adaptive Neuro-Fuzzy Inference System. The term "neuro-fuzzy" refers to a hybrid artificial intelligence approach that blends fuzzy logic and artificial neural networks. In ANFIS, the learning capability and relational architecture of artificial neural networks are combined with the logical process of fuzzy logic.⁸⁸ ANFIS, like artificial neural networks, may be trained with a large amount of data. As a consequence, the best optimum ANFIS structure for solving the related issue is discovered.⁸⁹ The resultant ANFIS architecture is tested to observe how it reacts to new samples. The ANFIS network topology is divided into two components, referred to as the premise and effect parts. ANFIS training entails using an optimization method to determine the parameters associated with these components. During training, ANFIS takes advantage of the existing control and response variables data pairings. Following that, fuzzy IF-THEN rules are built to demonstrate that these portions are linked.^{52,90}

The fuzzy inference systems (FIS) are also referred to as fuzzy models, fuzzy rule systems, fuzzy controllers, or fuzzy associative memory in literature. A FIS is made up of five functional parts: a rule base that comprises multiple fuzzy if then rules and also a database that specifies the membership functions of a processing unit that conducts the inference, combined with a unit that performs the inference. This is followed by a fuzzification interface that converts crisp inputs into fuzzy inputs and a defuzzification stage that transforms fuzzy inputs into crisp output. The knowledge base is usually referred to as a combination of the rule base and the database.

ANFIS is made up of five layers, as shown in Figure 4. ANFIS architecture with two numbers of inputs and a single



Figure 4. Typical ANFIS structure.

output is shown in this diagram. As per the ANFIS structure illustrated in Figure 3, the first layer is referred to as the fuzzification layer. To generate fuzzy clusters from input data, the fuzzification layer employs membership functions. The premise parameter is a set of parameters that governs the shape

of the membership function. The premise parameter set is a, b, and c. As listed in eqs 4 and 5, these parameters are used to determine the membership degrees of each membership function.⁹¹ The membership degrees are shown with θ_x and θ_y as

$$\theta_{A_i}(x) = \frac{1}{1 + \left|\frac{x-c}{a}\right|^{2b}}$$

$$\tag{4}$$

$$O_i^1 = \theta_{A_i}(x) \tag{5}$$

The second layer is referred to as the rule layer. Membership values obtained in the fuzzification layer are used to produce firing strengths (ϕ_{ii}) for the rules. The membership values are multiplied to find the ϕ_i values, as shown in eq 6. The third layer is denoted as the normalizing layer. It also determines each rule's normalized firing strengths. The normalized value is the firing power ratio of the *i*th rule to the total of all firing strengths, as determined by eq 7. Defuzzification is the name of the fourth layer. In each node of this layer, weighted values of rules are determined as shown in eq 8. The value is calculated using a first-order polynomial.^{92,93}

$$O_i^2 = \varphi_i = \theta_{A_i}(x) \cdot \theta_{B_i}(y), \text{ where } i = 1, 2$$
(6)

$$O_i^3 = \overline{\varphi_i} = \frac{\varphi_i}{\varphi_1 + \varphi_2 + \varphi_3 + \varphi_i}$$
, where $i = 1, 2, 3, 4$ (7)

$$O_i^4 = \overline{\varphi}_i f_i = \overline{\varphi}_i (p_i x + q_i x + r_i)$$
(8)

$$O_i^5 = \sum_i \overline{\varphi_i} f = \frac{\sum_i \varphi_i f_i}{\sum_i \varphi_i}$$
(9)

The actual ANFIS result is achieved through the last (fifth) layer. This layer is known as the layer of summation. The total outcome is calculated by adding the results of the respective rule inside the defuzzification layer. Equation 9 denotes the ruling expression for this layer.

3.3. Gene Expression Programming. GEP is an established algorithm that uses computer program evolution to solve linear and linear engineering problems. The computer program in GEP is generally encoded by fixed-length gene expression strings that are developed using nature-inspired operators like mutation and crossover. GEP has been proven to help find a precise and concise computer-based model for output prediction. Time-series predictions, classification difficulties, regression issues, data mining, and knowledge discovery are just a few of the real-world applications where it has been used with great success. It uses the historical data set of the problem to offer a solution in the form of an expression tree (ET). Ferreira⁹⁴ created this approach in 1999, and it was publicly launched in 2001. The GEP algorithm combines the major views of the two earlier legacy algorithms, namely, genetic programming (GP) and genetic algorithm (GA), to solve their flaws. In this technique, the chromosomal genotype is similar to that of a GA, whereas the phenotypic of a chromosome is a tree data structure of varying length and size, comparable with that of the GP algorithm.

In the case of several variables, the logical link between them (if any) may be a function of at least a function that can be correctly stated. The linking function can be Boolean logic operators (like IF, OR, and AND) and/or arithmetical operators (+, -, \times , /) or algebraic functions like exponential

and trigonometric. The logical link between variables should, of course, be investigated. Generally, three ETs are formed during the model development, and the sum of these form the overall model. A typical example of ET^{73} is shown in Figure 5. Utilizing the GEP technique, a group of linear type chromosomes is initially generated to discover the connection between variables *a* and *b* and *y*.



One of the variables may be inserted at each location of the genes on these chromosomes. It is time to evaluate each person's fitness (chromosome) in the generation examined once the chromosomes have been produced and their places filled. The chromosomes are represented as an expression tree in the GEP method for this reason (ET). An ET is comparable to a protein found in natural cells that controls the phenotype of a gene.

The conventional GEP technique consists of several operations. The first one is an initialization to create the starting population by producing a set of random chromosomes. Each element of the fixed-length strings is randomly assigned to each chromosome in the initial population based on the element's type. Functions and terminals are allocated to elements belonging to the head portion, whereas terminals are assigned to elements belonging to the tail component. The next operation is the fitness assessment, in which the fitness strengths of all chromosomes in the population are assessed. The problem-specific fitness evaluation function has a noteworthy impact on the performance of the test algorithm. The next stage is selection and replication. This stage selects the population's better chromosomes to create a new group for the upcoming generation. The roulette-wheel selection method and the tournament selection strategy are two examples of selection strategies that can be employed. The roulette-wheel selection with elitism has been demonstrated to produce superior results when tackling complicated issues.

3.4. Least Square Support Vector Machine (LS-SVM). Cortes et al.⁷⁷ originally proposed the support vector machine (abbreviated as SVM) in 1995. To overcome the problematic linear indivisibility, it employs the kernel function and follows Mercer's theorem to abstract features from the source. It has a strong capacity for generalization and can handle real issues such as minuscule samples, nonlinear data, and handling of local minima.^{87,95} It does, however, have the restricted quadratic programming problem, which has a significant calculation complexity. A generalized flowchart of LS-SVM is shown in Figure 6.

The support vector machines (SVM) offer higher generality than other machine learning approaches, so the SVM has gained substantial attention in the past decade as a strong prognostic tool. SVM technology is used in a variety of



Figure 6. Flowchart of LS-SVM process.

pubs.acs.org/EF

industries due to its strong generalization abilities like face identification, feature selection, and function estimation.⁷⁹ The expanded model of conventional SVM, least squares SVM (LS-SVM), transforms a quadratic programming (QP) problem into linear equations.⁹⁶ It is possible to achieve both a fast solution speed and a stable real-time function with this method. If $T = \{(x_i, y_i)\}$, where $(i = 1, 2, 3, \ldots, N)$, T denotes training data, x_i the measured value and y_i the predicted value. Then, LS-VSM can be expressed as

$$y(x) = \omega^T \mu(x_i) + d + e_i$$
(10)

In eq 10, ω denotes the assigned weight vector, $\mu(x_i)$) a nonlinear function, *d* deviation, and e_i residuals in the fitting.

Also, Min F(
$$\omega$$
, e) = $\frac{1}{2}\omega^T \omega(x_i) + \gamma \frac{1}{2} \sum_{i=1}^n e_{i'}^2$ (11)

If
$$y_i = \omega^T \mu(x_i) + d + e_i$$
, and $i = 1, 2, 3, \dots, N$ (12)

This describes Lagrange function L such that

$$L(\omega, b, e, \sigma) = I(\omega, e) - \sum_{i=1}^{n} \sigma_{i} \{ \omega^{T} \mu(x_{i}) + d + e_{i} - y_{i} \}$$
(13)

In eq 13, the Lagrange multiplier is denoted with σ_{ij} in which the partial derivatives of ω , *b*, *e*, are achieved at a point where σ_i becomes 0 to attain the optimization condition. At the optimized condition, ω is removed, and regression is accomplished.⁷⁷

$$y(x) = \sum_{i=1}^{n} \sigma_{i} J(x, x_{i}) + b$$
(14)

The LSSVM model for prediction is developed using the following steps:

- (i) Normalize the experimental data, and split it into two categories: training data and prediction data.
- (ii) The crossover technique is used to search the LS-SVM parameters using the training data, the specified kernel function, and the multiclass coding scheme.
- (iii) Finally, the prognostic model is developed with LS-SVM parameters.

3.5. Other Novel Al Methods. Traditional ANN seeks to construct a straight mapping between historical input data and output prediction data to achieve the forecasting method. However, the ANN-based model cannot characterize the link between data and time because of the absence of consideration of the time correlation in the data sequence, limiting its usefulness in time series forecasting techniques.⁹⁷ Therefore, the recurrent neural network (RNN) is offered as a solution to this problem. RNN may build a sequential mapping between input and output data by establishing cyclical connections to neurons. Thus, the output of each time step is influenced by the preceding time step's input. As a result, RNN realizes the "memory" feature.⁹⁸

A backward and forward are included in the RNN training procedure. The forward pass of an RNN is like a single hidden layer of a MLP. The stimulations from both the current external input and previous time steps' hidden layer responses appear at the hidden layer.⁹⁹ The nuance is that the loss function for RNN is affected by the instigation of the hidden layer not only via its effect on the output layer but also by its impact on the hidden layer at the subsequent step.98 Hochreiter and Schmidhuber¹⁰⁰ proposed the long shortterm memory (LSTM) network in 1997, a kind of RNN that integrates learning and model training without the need for extra domain knowledge. In contrast to traditional RNNs, the enhanced LSTM structure helps to minimize gradient collapse and explosion. As a result, LSTM has benefits in capturing long-term dependency and modeling nonlinear dynamics, and it may be utilized to cope with lengthy sequence data.^{98,101}

The radial basis function (RBF) neural network is also widely used because of its advantages, which include a simple structure, more estimation features, and a rapid training algorithm. The RBF network is a powerful conventional neural network structure.¹⁰² RBF networks are made up of three layers. The input layer's sole purpose is to link the network to its surroundings. The hidden layer comprises several nodes that use a radial basis function, such as the Gaussian function or the thin plate spline function, to apply a nonlinear modification to the input variables. The output layer is a summation unit and is linear.¹⁰³

The structure of an RBF network is selected through trial and error in the standard training method. The network parameters are determined in two stages. On the basis of the kmeans clustering method, the centers of the hidden layer nodes are determined in the first. The connection weights are generated in the second step using basic linear regression.^{104,105} Researchers in classification and model prediction also apply modern machine learning methods such as XGBoost, CatBoost, Boosting, Bagging regression tree, and LightGBM. In nanofluid-based heat transfer applications, the machine learning methods discussed in the previous subsections were mainly reported in the literature. To enhance prediction accuracy and computing efficiency, such machine learning approaches can be coupled with genetic algorithms, particle swarm optimization, and imperialist competitive algorithm.¹⁰⁶ The important historical progress in machine learning application in the domain of nanofluids is shown in Figure 7.



Figure 7. Flowchart of important previous research on a time scale.

3.6. Overfitting Problems in Machine Learning-Based Prediction Models. Although machine learning algorithms are adaptable to the properties of the observable values, overcomplexity in such models might result in overfitting. Overfitting occurs when the predicting algorithm describes the in-sample data too well, integrating not only the information associated with the data process but also the noises particular to that sample. In this scenario, the model is simply "memorizing" the observed patterns from the history, and thus, it is not truly "learning" the relevant patterns.^{107,108,109}

As previously stated, one of the causes of overfitting is that input signals are mixed with noises, resulting in poor accuracy; thus, increasing the data size is one method by which we can avoid the mixing of signals and noises. In machine learning, limiting the iteration is also known as the "early halting" approach; this overfitting avoidance strategy works only when our machine learning model learns repeatedly. Another method for improving the model's accuracy is to merge all of the weak learners into strong learners. This may be accomplished by bagging and/or boosting.^{79,110} Crossvalidation is another machine learning approach that gives a solution to the overfitting problem. Cross-validation, like ensemble learning, splits the data set, but the process is different. Regularization is another effective and widely used machine learning strategy for avoiding overfitting; this method matches the function of the training data set. This procedure causes the coefficient to move toward zero, minimizing mistakes.

3.7. Evaluation and Validation of Models. The precise uses of model assessment, model evaluation, and algorithm selection approaches are crucial in the efficient usage of machine learning techniques.¹¹¹ The main techniques for model evaluation and selection include the holdout method if the data set is small. The bootstrapping procedure is better suited for a larger data set. The bootstrapping technique is a resampling process for calculating population statistics by sampling a data set with replacement. It may be used to produce summary statistics like mean and standard deviation. Cross-validation and hyperparameter adjustment are other

computationally feasible strategies for model assessments. Common cross-validation procedures such as leave-one-out cross-validation and k-fold cross-validation have been employed by machine learning researchers.^{112,113}

Statistical assessment is one of the most common approaches for the evaluation of the prognostic efficacy of the models. The general indicators such as coefficient of determination, coefficient of regression, mean squared error, mean absolute percentage error, and Theil's statistics have been utilized by various published investigations.¹¹⁴ The appeal of statistical approaches for regression issues arises from their simplicity of implementation and flexibility irrespective of the size of the data set. Finally, when comparing machine learning algorithms with limited data sets, additional algorithm selection procedures such as the combined F-test, 5×2 cross-validation, and nested cross-validation are also advocated.¹¹⁵

4. MACHINE LEARNING FOR THERMOPHYSICAL PROPERTIES

The heat transfer in a heat exchanger device or equipment is dependent on the characteristics of the heat transfer fluid; therefore, precise measurement and forecasting of nanofluid's thermophysical characteristics are important.¹¹⁶ However, numerous factors impact the thermophysical characteristics of a nanofluid, including nanoparticle type, concentration, shape, size, shape, temperature, rate of shear, preparation method, kind of base fluid, flow condition, and so on. For example, hundreds of high thermal conductivity nanoparticles, such as $Al_2O_{3,}^{81} Ag$,¹¹⁷ CuO,¹¹⁸ graphene,¹⁹ SiO₂,¹⁸ MgO,¹¹⁷ Ni,¹¹⁹ TiO₂,¹²⁰ Fe₃O₄,¹²¹ and SiC¹²², have been investigated for producing nanofluids. As a result, determining the thermophysical characteristics of a nanofluid is exceedingly difficult and remains a key challenge in nanofluid research. Experimental property measurement costs are high, and the parametric data range is limited.⁶² Analytical or numerical models need values of different nanofluid properties and interactions between nanoparticles and base fluids to be understood. On the other hand, machine learning is capable of reducing in several experimental studies. Once the prognostic model is developed, it can be used for forecasting the thermophysical properties with high accuracy.

4.1. Thermal Conductivity. The basic aim of developing nanofluids was to enhance the thermal conductivity of the base fluid by adding nanoparticles. A significant number of experimental studies have been carried out to test the thermal conductivity of nanofluids during the last two decades using various technologies such as the temperature oscillation method, transient hot-wire method, and steady-state parallel plate method. The thermal conductivity of a test nanofluid is determined by numerous parameters, including nanoparticle size, concentration, synthesis technique, and temperature. The majority of the literature revealed that nanofluids might increase heat conductivity to various degrees.^{123,124} The basic characteristics of nanoparticles, base fluid, and their temperature, are macroscopic parameters that can impact the thermal conductivities of nanofluids.¹²⁵

One of the most significant factors for theoretical as well as numerical analyses of heat transport systems employing nanofluids as coolants is the selection of nanofluid's thermophysical characteristics. Given this, understanding how to create an appropriate thermophysical characteristic model is quite beneficial and important. The analytical model-based

method and data-driven machine learning approach are the two main approaches used to model and forecast the characteristics of nanofluids.^{62,126} Compared to the analytical model-based method, the data-driven machine learning approaches represented by the ANN, 42,127 ANFIS,¹ GEP,¹²⁹ for example, have attracted a lot of interest in recent years because of their superior mapping, modeling, and forecast capabilities. According to the literature study, the predictive models' accuracies are related to their structures, functions used, input variables used, and algorithms used.^{66,68} Maleki et al.¹³⁰ conducted a comparison of prediction performances for three machine learning algorithms For model prediction of the TC of nanofluids containing ZnO nanoparticles, the multivariate adaptive regression splines (MARS), ANN, and group method of data handling (GMDH) were utilized. To compare the models' confidences, several statistical indicators were utilized. The most accurate model has been determined to be ANN, followed by GMDH and MARS. ANN, GMDH, and MARS had R² values of 0.998, 0.998, and 0.987, respectively. The sensitivity analysis found that the thermal conductivity of the base fluid was the most critical input component. The thermal conductivities of nanofluids containing CuO nanoparticles were model predicted by Komeilibirjandi et al.,¹³¹ using ANN and GMDH. Both models employed thermal conductivities, temperature, mass fraction, and nanoparticle size as inputs. When polynomial and ANN were utilized, the R-squared values were 0.986 and 0.9995, respectively. On the basis of these statistical findings, it is possible to conclude that utilizing an ANN-based regression to predict the TC of nanofluids containing CuO nanoparticles produces a more trustworthy model. Rostami et al.¹³² explored the effects of adding MWCNT on the TC of paraffin using a comparative study between ANN and the response surface methodology (RSM). A nanofluid temperature (25–70 °C) and mass fraction (0.005-5%) were used as input variables, while the nanofluid thermal conductivity was used as an output parameter. Finally, both methods' mean squared errors (MSEs), R, and maximum margin of deviation (MMD) were compared. The R indices for ANN and RSM models were 0.993 and 0.972, correspondingly, implying that the ANN approach was more accurate than the RSM. Maleki et al.¹³³ applied GMDH, ANN with scaled conjugate gradient (SCG), and the Levenberg-Marquardt (LM) training algorithm for prediction of thermal conductivity of silica nanoparticles in different base fluids. The model-based prediction results were quite close to the results of experimental research. The R² values for GMDH, LM, and SCG in optimal circumstances were 0.9996, 0.9990, and 0.9998, individually. Furthermore, the MSE using these three techniques were 0.000010, 0.000032, and 0.0000078 for

GMDH, LM, and SCG, respectively. In a study by Alrashed et al.,¹³⁴ carboxylic diamond nanoparticles and multiwalled carbon nanotubes were dispersed in water without the use of any chemicals or surfactants. The viscosities, thermal conductivities, and densities of the nanofluids were then measured at temperatures ranging from 20 to 50 °C, with a maximum volume fraction of 0.2 vol.%. Using this experimental data, a nonlinear regression, ANN, and ANFIS models were utilized to anticipate the TC of nanofluids. On the basis of mean absolute percentage error (MAPE), ANN was shown to be the best model prediction strategy among these three models. Razavi et al.¹³⁵ developed a prediction model of thermal conductivities of nanofluids using ANFIS and LSSVM. The prognostic models based on the LSSVM and ANFIS algorithms were compared using 15 types of correlation-based statistical indices. The results showed that the LSSVM method outperformed the ANFIS model. In a study by Said et al.,¹⁹ the synthesis of a rGO/Co_3O_4 nanocomposite was accomplished, and the thermophysical characteristics were evaluated. The data were acquired under various experimental settings of concertation and temperature. In this study, the authors used a novel algorithm, namely, marine predators, for determining the optimal values of input controlling parameters (concentration ratio and temperature) that optimally minimizes two response variables (density and viscosity) at the same time while maximizing the other two response variables (specific heat and thermal conductivity). Said et al.¹³⁶ synthesized functionalized carbon nanofiber (F-CNF), carbon nanofiber (CNF), and rGO coated F-CNF (F-CNF/rGO). Furthermore, fuzzy logic techniques were used to develop an accurate prediction model based on the experimental data set. The observed values of thermal conductivity and the fuzzy model were found to be wellfitting. For thermal conductivity, the proposed numbers for fuzzy model rules were 16. The fuzzy model was trained for a total of 50 epochs. The thermal conductivity model had quite a low mean squared error (MSE) of 1.0709×10^{-5} , indicating superior prognostic efficiency.

In a study by Said et al.,¹³⁷ the heat transfer performance of a nanofluid combination was improved by utilizing a hybrid approach of fuzzy-based modeling with PSO. The TiO₂ and Al₂O₃ nanoparticles distributed in distilled ethylene glycol and water with 50:50 volumetric proportions are studied. Several volume fractions (0.05 and 0.3 vol%) and temperatures ranging from 25-70 °C were used to measure the nanofluid characteristics. The thermal conductivities at 25 °C improved by 9.52% and 26.52% for the Al₂O₃ nanofluid at 0.05% and 0.3% concentrations, respectively. At a 25 °C temperature of TiO₂ nanofluid with concentrations of 0.05% and 0.3%, the thermal conductivities could be improved by 7.77% and 18.05%, respectively.

Motlagh et al.¹³⁸ used GEP for model prediction of the TC of nanofluids including CuO and Al₂O₃ nanoparticles-water nanofluids. The derived novel model is dependent on nanoparticle size, volume fraction, and temperature. The model was built using the train data set, and the results were compared with the test data set. The findings showed that the GEP can accurately estimate and predict the TC of nanofluids and that it may be utilized to simulate engineering challenges. Pourrajab et al.¹²⁹ used GEP, linear genetic programming (LGP), and local weighted linear regression (LWLR) to develop a rigorous predictive model for ethylene glycol-based hybrid nanofluids. According to performance criteria, the LWLR model outperformed LGP (RMSE = 0.0259, R = (0.964) and GEP (RMSE = 0.0474, R = 0.865) in predicting thermal conductivities of hybrid nanofluids at RMSE = 0.014 and R = 0.988. Sensitivity analysis found that the temperature, volume fraction, and nanoparticle size were among the most important parameters. Table 2 lists some of the key research studies that used machine learning approaches to estimate the thermal conductivities of nanofluids.

4.2. Specific Heat. The specific heat capacity (SHC) of nanofluids is the degree of the heat retentive ability of the nanofluid. As a result, it determines the nanofluid's thermal behavior. Furthermore, the SHC capacity measurement can aid in estimating other thermally related characteristics such as

Table 2. Machine L	earning Techniques in Model Predic	tion of Thermal Conductivities o	of Nanofluids		
Reference	NF used	ML technique	Input	Performance parameter	Outcome
Maleki et al. ¹³⁰	NFs of ZnO particles	Multivariate adaptive regression splines (MARS), ANN, group method of han- dling data (GMDH)	TC of base fluids, volume percent of solid phase, particle size, and temperature	$ \begin{array}{l} R^2 \ (ANN) = 0.9987, \\ R^2 \ (GMDH) = 0.9980, \\ R^2 \ (MARS) = 0.9879 \end{array} $	Sensitivity analysis revealed thermal conductivity of base fluids being most important factor
Komeilibirjandi et al. ¹³¹	NFs of ZnO particles	ANN, GMDH	TC of base fluid, temperature, mass fraction, and nanoparticle size	R^{2} (ANN) = 0.9996, R^{2} (GMDH) = 0.9826	ANN superior in model prediction of TC
Rostami et al. ¹³²	Paraffin/MWCNT	ANN, RSM	NF temperature and concentration	R (ANN) = 0.993 R (RSM) = 0.972	ANN better than RSM in model prediction of TC
Maleki et al. ¹³³	Silica nanoparticles in different base fluids	GMDH, ANN, scaled conjugate gradient (SCG)	TC of base fluid, temperature, mass fraction, and size	$ \begin{array}{l} R^2 \ (ANN) = 0.9997, \\ R^2 \ (GMDH) = 0.9991, \\ R^2 \ (SCG) = 0.9998 \end{array} \end{array} $	All three techniques developed robust TC models
Alrashed et al. ¹³⁴	Carboxylic diamond and MWCNT dispersed in water	ANN, ANFIS	TC of base fluid, temperature, mass fraction, and size	MAPE $(ANN) = 0.00842$ MAPE $(ANFIS) = 0.16294$	ANN superior to ANFIS in model prediction of TC
Said et al. ¹⁹	rGO/Co ₃ O ₄ nanocomposite	Marine predator algorithm	Concentration ratio and temper- ature	I	For optimization of TC and SHC
Said et al. ¹³⁶	functionalized carbon nanofiber (F-CNF), car- bon nanofiber (CNF), rGO coated F-CNF (F-CNF/rGO)	ANFIS	Concentration ratio and temper- ature	MSE = 0.0000107	ANFIS superior prognostic efficiency

thermal diffusivity, Nusselt number, and heat transfer rate.^{63,139,140} In addition, the SHC is an important parameter for the proper assembly and design of heat transfer systems to minimize the loss of energy and increase the conservation of energy.^{141,142} Several scholars have studied the specific heat capacity (SHC) of various nanofluids and their factors. The physicochemical characteristics of the nanofluid, such as the sizes of nanoparticles, volume fraction, temperature of the nanofluid, types of nanoparticles, and specific heat capacity of the base fluid affect the SHC of nanofluids.^{63,143-145} Whereas numerical methods have provided an alternate solution to experimental investigation for evaluating the SHC of nanofluids, their precision has been limited by several factors, including presumptions controlling the relationship among variables, selectivity and an incomplete synopsis of the atypical nature of nanofluids.^{41,56} Considering the aforementioned factors, computational technologies such as machine learning (ML) techniques are employed since they give a more accurate approximations.

Alade et al.¹⁴¹ employed a support vector machine and ANN to prognosticate the specific heat capacity (SHC) of CuO/ water nanofluids. The models were created using experimentally observed SCH data on CuO/water nanofluids at a volume % range from 0.4% to 2%, with temperatures ranging from 293 to 338 K. The findings showed that the SVR model was slightly superior to the ANN model. However, as compared to existing theoretical models, both the SVR and ANN models exhibit improved prediction abilities for the SHC of CuO/water nanofluids. Adun et al.40 employed ANN for prognosticating the specific heat of a Fe₃O₄-alumina-zinc oxide/water ternary type hybrid nanofluid. The R² value indicating the coefficient of determination between observed and ANN model predicted values for specific heat was 0.9555. This indicates that the observed and projected specific heat values were in good agreement. Shi et al.¹⁴⁶ also used ANN for the development of a prognostic model for thermophysical properties of a carbonmagnetic type nanofluid. The objective of this research was to use experimental data on viscosity, thermal conductivity, and specific heat to build an ANN that could predict the thermophysical characteristics of magnetic nanofluids. The ANN model could predict the specific heat values within 5% of observed values, indicating an efficient predictive model.

To forecast the SHC of nanofluids, Jamei et al.¹⁴⁷ presented a Gaussian process regression (GPR)-based prognostic model. It was compared to the random forest (RF) technique and the generalized type regression neural network (GRNN). The GPR model (RMSE = 0.015 J/K.g, R = 0.9997) outperformed the RF (RMSE = 0.045 J/K.g, R = 0.9976) and GRNN (RMSE = 0.0608 J/K.g, R = 0.995) models in terms of performance. The results demonstrated that the proposed model could properly predict the SHCs of the nanofluids under study. Jamei et al.⁶³ used GEP for model prediction of specific heat using large data sets of diverse nanofluids. The SHC data of nanofluids, namely, Al2O3, ZnO, TiO2, SiO2, MgO, and CuO dispersed in various base fluids, were acquired from published studies. The GEP-based model was robust and efficient having an R value of 0.9570, RMSE value of 0.0650, and MAPE of only 5.5134. A summary of recent research works on machine learning-based model prediction of specific heat capacity is shown in Table 3.

4.3. Viscosity. Nanotechnology has attracted the interest of scientists, academics, and engineers during the last two decades because of rapid advancement. One of the unexpected

Table 3. Machine l	Learning Techniques in Model 1	Prediction of Specific Heat Capacities of	Nanofluids		
Reference	NF used	ML technique	Input	Performance parameter	Outcome
Alade et al. ¹⁴¹	CuO/water nanofluids	Support vector regression (SVR), ANN	Volume percent of solid phase, temperature	RMSE (ANN) = 0.0025 $RMSE (SVR) = 0.0023$	SVR slightly superior to ANN in model prediction
Adun et al. ⁴⁰	Fe ₃ O ₄ -alumina-zinc oxide/water ternary type hybrid nanofluid	ANN	Temperature, mass fraction, nanoparticle size	R^{2} (ANN) = 0.9555	Observed and projected SHC values almost identical
Shi et al. ¹⁴⁶	Carbon-magnetic type nanofluid	ANN	Volume percent of solid phase, temperature	R = 0.95	Observed and projected SHC values almost identical
Jamei et al. ¹⁴⁷	Metal oxide and carbon-based nanoparticles in different base fluids	Gaussian process regression (GPR), random forest (RF), generalized regression neural network (GRNN)	Solid volume fraction, temperature, size of nanoparticles, SHC of base fluids	R (GPR) = 0.9997 , R (RF) = 0.9976 , R (GRNN) = 0.9956	All three techniques developed robust SHC models
Jamei et al. ⁶³	Several nanofluids	GEP	Solid volume fraction, temperature, size of nanoparticles, SHC of base fluids	R = 0.9570, RMSE = 0.065 MAPE = 5.5134	GEP proved an efficient model prediction technique
Hasan and Banerjee ¹⁴⁸	Molten nitrate salt mixture seeded with silica, alumina, and titania nanoparticles	ANN	Mass fractions, temperature	R^{2} (ANN) = 0.9998, MAPE = 2%	Observed and projected SHC values in good agreement

pubs.acs.org/EF

consequences of this technology is nanofluids, which can dramatically improve the efficiency of thermal systems. Because solid nanoparticle-containing nanofluids have a greater viscosity than ordinary working fluids, evaluating the viscosity is important for building thermal systems and calculating pumping power.¹⁴⁹ Viscosity is one of the most significant thermophysical characteristics that influence how nanofluids transmit momentum and heat. The heat transfer performance of nanofluids requires an accurate estimate of this parameter.⁸⁴ Their dynamic viscosity heavily influences the thermal behavior and heat transmission capabilities of nanofluids. According to experimental investigations in this sector, the viscosity of nanofluids rises as the volume concentration of nanoparticles increases. According to studies, a nanofluid's viscosity decreases as the fluid's temperature rises.¹⁵⁰ This phenomenon results from the decline in intermolecular interaction between the nanoparticles and the base fluids due to temperature rise. The base fluids with high viscosities tend to minimize Brownian motion, resulting in less agglomeration as nanoparticles are less likely to collide. Nanofluids with high base fluid viscosities become steadily more stable than nanofluids with lower base fluid viscosities. 151,152

Numerous forecasting models have been proposed for predicting dynamic viscosity based on influencing factors such as the sizes, types, and volume fractions of nanoparticles and also their temperature, using various methods such as ANN, ANFIS, SVM, PSO, GEP, GPR, and mathematical correlations.^{153–155} In the recent decade, various models based using intelligent methods for the prediction of hybrid nanofluid viscosity have been proposed. The modeling technique employed, the data utilized for training the model, the choice of the input parameters, and so on all impact the efficacies and accuracies of these models. The models produced by various distinct machine learning methods are evaluated in this review article. Toghraei et al. 156 explored the viscosity of an ethyleneglycol/Ag nanofluid in temperatures ranging from 25 to 55 °C, having a volume percentage of nanoparticles ranging from 0.2% to 2%. A data set of 42 samples obtained from the experiment was used to create an ANN model to predict the dynamic viscosity. ANN model outcomes were compared with correlation findings. In comparison to the correlation approach, the ANN could accurately estimate the viscosity of the Ag/ethylene glycol nanofluid. The predictive model's MSE was 0.0012, with the highest amount of error as 0.0858. The dynamic viscosities of oxide nanoparticles suspended in water/ ethylene glycol were model predicted using an ANN model by Longo et al.¹⁵⁷ This research used nanoparticle concentrations, temperatures, diameters of nanoparticles, average sizes of nanoparticle clusters, and base fluid characteristics for modeling. The ANN-based model predicted experimental data quite well, with a MAPE value of 4.15%. Chen et al.¹⁵⁸ also used ANN to model and forecast the viscosity of a hybrid type of nanolubricant (SAE50/MWCNTs-TiO₂). The performance of the ANN-based predictive model was compared with conventional curve fitting. The study concluded that ANN performs far superior to the conventional curve fitting model. Sedaghat and Yousefi¹⁵⁹ also used ANN to model and predict the viscosities of innovative graphene quantum dot nanofluids. With an $R^2 = 0.99915$ for the viscosity model, the overall predicted viscosities of nanofluids were in excellent agreement with experimental results. Baghban et al.¹⁶⁰ used ANFIS to prognosticate the viscosity of nanofluids. The voluminous experimental data (1277 data set) for model

development were collected from published research. The temperatures, densities of nanofluids, nanoparticle diameters, and viscosities of base fluids were used to correlate the viscosities of nanofluids. The model's robustness was demonstrated by a high R² of 0.99997. Furthermore, the absolute and maximum average relative deviations of 0.42% and 6.45% were reported. Hemmat Esfe et al.¹⁶¹ presented a comprehensive study comparing three different artificial intelligence-based methods to forecast the viscosity of a nanolubricant (TiO₂/SAE 50). Genetic algorithm type radial basis function (GA-RBF) neural networks, LS-SVM, and GEP were used in the study. The models were evaluated on a statistical basis, and GA-RBF was found to be the most accurate among them. Researchers in the recent past have reported several other nonconventional and modern AI-based modeling methods. Bardool et al.¹⁶² used friction theory for predicting nanofluid viscosity data. The repulsion and attraction pressure factors in the residual viscosity term of the friction model were included using the Esmaeilzadeh-Roshanfekr and Peng-Robinson equations of state. A total of 711 data sets obtained by experiments using a variety of nanofluids were used in modeling. The R-squared values for the Peng-Robinson and Esmaeilzadeh-Roshanfekr equations of state, respectively, were 0.9978 and 0.9979, showing extremely high performances of the proposed model. Yan et al¹⁶³ conducted a comparative study between ANFIS and SVM to model and predict the viscosities of MWCNTs-TiO₂/EG hybrid nanofluids over a range of temperatures and volume concentrations. Both approaches can accurately predict the rheological behaviors of MWCNTs-TiO₂/EG nanofluids based on their error calculations. The ANFIS technique, however, beats the SVM method in terms of accuracy. Shahsavar et al. 164 employed a GMDH type of NN to develop a prediction model for viscosity. The study reported a robust model to predict the viscosities of liquid paraffin and Fe₃O₄ nanofluid accurately. Ansari et al.¹⁶⁵ used various types of neural networks based on different training algorithms such as Bayesian regulation backpropagation (BR), scaled conjugate gradient (SCG), resilient backpropagation, and Levenberg-Marquardt (LM), including several transfer functions. Among these, the network architecture with one hidden layer and 23 neurons in the output and hidden layers with tan-sigmoid and purelin transfer functions was found to have the best performance. With an overall MSE value of 0.00901 and R² of 0.9954, the suggested network was robust enough to properly correlate and predict relative viscosity. Olumegbon et al.¹⁶⁶ employed a SVM-based model to forecast the viscosities of various carbon nanomaterials distributed in diesel oil. MWCNTs-graphene nanoplatelets and their hybrid were tested in the lab to acquire 120 experimental data points. The fluid temperatures, mass fraction of nanoparticles, and viscosity of diesel oil were chosen as model inputs. The training data set had an R of 0.9998 and an RMSE of 0.0076, whereas the testing data set had an R of 0.9999 and an RMSE of 0.0026. The suggested model was shown to predict the viscosities of carbon-based nanomaterialsdiesel oil nanofluids with good accuracy. Table 4 depicts a summary of recent research studies on machine learning-based model prediction of viscosity.

4.4. Density. The literature review revealed that most of the emphasis had been dedicated to viscosity and thermal conductivity among the different thermophysical characteristics of nanofluids. The specific heat and density could not receive that much attention. In heat transfer applications as

Reference	NF used	ML technique	Input	Performance parameter	Outcome
loghraei et al. ¹⁵⁶	Ag/wthylene-glycol nanofluid	ANN	Volume percent of solid phase, temperature	MSE (ANN) = 0.0858	ANN accurately estimates viscosity of Ag/ethylene glycol NF
ongo et al. ¹⁵⁷	Metal oxide nanoparticles suspended in water/ethylene glycol	ANN	Volume percent of solid phase, temperature	R^{2} (ANN) = 0.9878	Observed and projected viscosity values in good agreement
Chen et al. ¹⁵⁸	Hybrid nanolubricant (SAE50/MWCNTs-TiO ₂)	ANN	Concentrations, temperature	Compared with conventional curve fitting	ANN superior to conventional curve fitting
3aghban et al. ¹⁶⁰	Several nanofluids (1277 data sets)	ANFIS	Temperature, density of nanofluid, nanoparticle diameter, base fluid viscosity	\mathbb{R}^2 (ANN) = 0.9999, Absolute average relative deviation (AAD) = 0.42%	ANFIS produced robust predictive model
Hemmat Esfe et al. ¹⁶¹	TiO ₂ /SAE 50 nanolubricant	GA-RBF neural networks (GA-RBF), LS-SVM, GEP	Solid volume fraction, size, temperature of nanoparticles, SHC of base fluids	RMSE (GA-RBF) = 0.58 RMSE (LS-SVM) = 1.28 RMSE (GFP) = 6.59	GA-RBF performed superior to LS-SVM and GEP
⁄an et al. ¹⁶³	MWCNTs-TiO ₂ /EG hybrid nanofluids	ANFIS, SVM	Mass fractions, temperature	R^{2} (ANFIS) = 0.992, R^{2} (SVM) = 0.987	ANFIS performed superior to SVM
hahsavar et al. ¹⁶⁴	Liquid paraffin, Fe ₃ O ₄ nanofluid	GMDH	Mass fractions, temperature	$R^2 = 0.96$, RMSE = 0.0179	GMDH type ANN efficient in model prediction of viscosity

applicable to renewable energy also, the heat transfer coefficient calculation requires the value of density.¹⁶⁷ Density also affects the nanofluid flow, so this parameter should be properly determined.^{19,111} The addition of a modest number of solid nanoparticles to a base liquid would increase the density of the combination over time. This is because solids have a higher density than liquids. Estimating the material densities and volume concentrations of constituents is required for modeling densities in nanofluids.¹⁶⁸

Several studies have been reported in recent years on labbased measurement of nanofluid density. In an experimental investigation by Said et al.,¹³⁶ the densities of single and hybrid nanofluids were measured. The study found that the densities of nanofluids and base fluids decrease when the temperature rises, because of the liquid's thermal expansion. Compared to the base fluid, the average density rise for all nanofluids tested was less than 0.11%. Żyła et al.¹⁶⁹ measured the density of ethylene glycol (EG)-based nanofluids comprising three kinds of nitride nanoparticles: aluminum nitride (AlN), silicon nitride (Si₃N₄), and titanium nitride (TiN). Measurements were taken at three distinct temperatures: 308.15, 298.15, and 288.15 K. It was discovered that the proportion of nanoparticles has a considerable impact, but particle size does not affect density. The determination of a nanofluid's thermophysical characteristics is a tedious and onerous process in nanofluid research. Density measurement via experiment is expensive, and the parametric data range is limited. Numerical and analytical models need property values and interactions between base fluids and nanoparticles to be understood.¹⁷⁰ On the other hand, machine learning is capable of efficiently model predicting the densities of nanofluids over a large range of operating parameters.¹⁷¹ Said et al.¹³⁷ used fuzzy to model predict the density using a range of operating conditions (vol % concentration and temperature). The study used the Al_2O_3 and TiO₂ nanoparticles distributed in 50:50 volumetric quantities of ethylene glycol and distilled water as nanofluids. Using the suggested technique, the optimum condition that generates the ideal output that minimizes the density while also maximizing thermal conductivity was obtained. Montazer et al.¹⁷² used response surface methodology (RSM) to present a novel correlation for the estimation of carbon-based nanofluid density. The several machine learning methods applied for modeling thermophysical properties demonstrate that high prognostic efficiency can be achieved. Most machine learning techniques can accurately predict the outcomes. However, the efficacy of any machine learning technique depends upon the quality of data, the quantity of data, human efficiency in the measurement of thermophysical properties, human efficiency in data collection, the technique of data collection, and uncertainty in measurement. Nevertheless, the machine learning techniques are excellent means of model prediction, resulting in saving costs and human efforts. A summary of recent research studies employing machine learning for model prediction of density is illustrated in Table 5.

5. MACHINE LEARNING FOR RE

The renewable energy system does suffer from a certain degree of uncertainty in demand and supply balancing. Demand response (DR) has gained popularity in recent years as a costeffective approach to provide flexibility, thus enhancing the energy network dependability. However, because of the high complexity of DR activities, the frequent requirement for near ANN developed robust predictive model for density

ANN produced robust predictive model for density

Maximum error in density

Temperature, volume

fractions

Temperature, volume fractions

component analysis (PCA)

ANN

Alumina/distilled water (DW), MWCNT)/DW, graphene

Alumina/distilled water (DW), MWCNT)/DW

nanoparticle/DW

Sb₂O₃, ZnO/(EG+H₂O), SnO₂/(EG+H₃O), ZnO/(PEG +H₂O), Al₂O₃/(EG+H₂O), TiO₂/EG, ZnO/PEG

Yousefi and Amoozandeh¹⁷⁴

Singh and Ghosh¹⁷⁵

Giwa et al.¹⁷⁶

Mirsaeidi and Yousefi¹⁷³

Kumar et al.⁴²

ANN + Principal

measurement < 0.2%

AAD (ANN) = 0.01%

AAD = 0.01%

MSE < 0.001 $R^2 = 0.9999$

phase, temperature phase, temperature

Concentrations, temperature

R = 0.999

ANN performed superior to ANFIS

RMSE (ANFIS) = 0.154,RMSE (ANN) = 0.133

Base fluid, mass fractions.

ANFIS, ANN

temperature

ANN produced robust predictive model for density

Observed and projected density

values in good agreement

ANFIS accurately estimates densities of nanofluids

Outcome

Performance parameter

 R^{2} (ANN) = 0.9982

Jolume percent of solid Volume percent of solid

Input

ML technique

ANFIS

 $\rm Al_2O_3$ and TiO_2 nanoparticles dispersed in deionized water and EG MWCNT-water-based Al₂O₃, TiO₂, ZnO, CeO₂ nanofluids

NF used

Reference

Said et al.¹³⁷

ANN

ANN

Water–ethylene glycol mixture and carbon quantum dots (CQDs) nanofluids

real-time choices, and the utilization of large-scale data, machine learning methods have lately been developed as essential technologies. AI may be employed to solve a variety of problems such as predicting demand, raw material supply, seasonal cost variations, dynamic pricing, scheduling, and control.59

5.1. Heat Exchangers. The heat exchanger is essential equipment in the renewable and sustainable energy sectors. The hydrodynamic and heat transport properties of heat exchangers significantly impact the performance of renewable and sustainable energy systems.^{177,178} There are various methods for increasing a heat exchanger's efficiency. The use of working fluids with improved thermal characteristics would be the most efficient of these strategies.¹⁷⁹ Nanofluids are a viable alternative for improving heat transmission in heat exchangers due to their outstanding thermophysical characteristics.¹⁸⁰ Because of the relevance of nanofluids, numerous researchers explored their performance and use in various heat exchangers.

The double pipe heat exchanger (DPHE) has been used extensively in industrial applications such as power plants, air conditioning, refrigeration systems, water heaters, and petrochemicals. Thermal conductivity is a significant factor in enhancing the heat transfer rate of fluids.¹⁸¹⁻¹⁸³ Many attempts have been made to introduce fine solid particles in the base fluid to enhance the TC.^{184–187} Ali et al.¹⁸⁸ explored the heat transfer properties of a MgO oil-based nanofluid used in a double-pipe heat exchanger. The study discovered a considerable improvement in the heat transfer rate and total heat transfer coefficient. Alumina nanoparticles were utilized with a base fluid in the majority of the works.^{189–191} Chun et al.¹⁸⁹ reported that the use of alumina nanoparticles in the DPHE significantly increases the coefficient of heat transfer. In an experimental study, Abd Elhafez et al.¹⁸⁶ found the enhanced thermal performance of a DPHE with an Al₂O₃ nanofluid in specific ranges with no significant pressure drop penalty. Mansoury et al.¹⁹¹ studied the performance of the DPHE, plate type heat exchanger, and STHEx, in which Al₂O₃/water nanofluid (0.2%, 0.5%, and 1% vol %) concentrations were employed. The study revealed that a DPHE exhibited the best performance with an improvement of heat transfer coefficient by 60%. Saeedan et al.¹⁹² used the water-based nanofluids of Cu, CuO, and CNT nanoparticles in the DPHE. Effects of Reynolds number and volume concentration were also investigated. It was observed in the study that rises in the Reynolds number and volume concentration enhance the heat transfer coefficient. Sarafraz and Hormozi¹⁹³ explored the effect of rate of flow rate and vol % concentration on the coefficient of heat transfer. Water/ ethylene-glycol (50:50 by volume) was used as a base, whereas particles were extracted from green tea leaves and silver nitrate. An enhancement of 67% was reported in the heat transfer coefficient at vol % = 1. Bahiraei et al.¹⁹⁴ explored the efficacy of silver particle-based NF in the DPHE. The study revealed that particle migration increased the heat transfer rate, whereas rises in the concentration and Reynolds number resulted in improved heat transfer. Ravi Kumar et al.¹⁹⁵ used a Fe₃O₄ nanofluid in the DPHE. The nanoparticles utilized in this work have volume concentrations of 0.005%, 0.01%, 0.03%, and 0.06%, with Reynolds numbers in the range from 15,000 to 30,000. Nusselt number enhancement by 14.7%, compared to the base fluid, was reported in the study for the concentration of 0.06% and Reynolds number of 30,000. Shakiba and

Table 5. Machine Learning Techniques in Model Prediction of Densities of Nanofluids

6638

pubs.acs.org/EF

Vahedi¹⁹⁶ studied the influence of a magnetic field over hydrothermal characteristics of a ferrofluid in a DPHE and found that there were increases in Nu number, pressure drop, and friction factor. This was attributed to the transverse magnetic field changing the axial velocity profile. Sözen et al.¹⁹⁷ experimentally showed that using alumina and fly ash nanofluids in heat exchangers improved the efficiency. As a result of the existing literature's research, it has been established that using nanofluids in heat exchangers greatly improves their thermal performance.

To improve the performance of the plate-type heat exchanger (PHE), two major variables must be considered: the amount of pressure drop and heat transfer. The degree of heat transfer must be raised while the pressure drop must be lowered. Several optimization studies have been carried out for PHEs.¹⁹⁸⁻²⁰² Pandya et al.²⁰³ reviewed the enhancement of heat transfer enhancement using nanofluids in PHEs. Several authors reviewed and identified the various heat transfer enhancement techniques in PHEs, such as louver fin, corrugated fin, perforated fin, artificial roughness, extended surface, winglets, and staggered wavy fin.²⁰⁴⁻²⁰⁶ Inserts and vortex generators were also studied as heat transport improvement strategies.²⁰⁷ A modification to the plate was also made to improve heat transfer.^{208,209} The chevron angle was found as the significant geometrical parameter in heat transfer improvement.²¹⁰ Besides the above approaches, nanofluids are also attractive solutions to improve the efficiencies of heat exchangers. Because of the growing importance of nanofluids, many researchers explored the performances of PHEs employing nanofluids.^{211,212} In general, performances of PHEs employing nanonulds. In general, researchers studied the pressure drop and heat transfer characteristics.^{213,214} Many scholars numerically investigated the applications of nanofluids in the PHEs.^{215–217} Various nanoparticles like SiO,^{218,219} TiO₂,²²⁰ CNT,^{221,222} Al₂O₃,^{223–225} Ag,^{226–228} CuO,²²⁹ ZnO,²³⁰ and hybrid nanoparticles^{231–234} have been studied to improve the performance of the PHEs. Kumar et al.²³⁵ investigated the effect of nanofluid flow on the performance of the square channel. Attalla and Maghrabie $^{\rm 224}$ employed an $\rm Al_2O_3/water$ nanofluid to evaluate the performance of the PHE. Experiments were conducted for various surface roughness values and different volume concentrations with Re ranging from 500 to 5000. Estimates were made for the friction factor, Nu number, and heat transfer enhancement factor. It was concluded that enhancing the vol % of nanoparticles improves the heat transfer rates and pressure drops in PHEs. Despite having a larger pressure drop and friction factor than the base fluid, MWCNT/water nanofluids showed improved overall thermal performances.²²¹ Similarly, Goodarzi et al.²²² also reported that the performances of PHEs could be improved by using MWCNT. Fouling of the nanoparticles over the inner body of the heat exchanger generally enhances the roughness, thermal resistance, friction factor, and drop in pressure. Sarafraz et al.²²⁹ used a low-frequency vibration, which improved the heat exchanger's thermal performance. The latest research work in STHEx revealed that nanofluids containing different types of nanoparticles such as Ag,²³⁶ CuO,²³⁷ Al₂O₃,²³⁸ TiO₂,²³⁹ and graphene oxide²⁴⁰ outperformed their respective base fluids in terms of thermal performance. Nazarzade et al.²³⁶ experimentally explored the forced convective heat transfer in STHEx with a silver/water nanofluid. Silver nanoparticles were first synthesized using a reduction process and then dispersed in water to make the nanofluid. The results showed that

nanofluid had a higher overall heat transfer coefficient compared to water as a base fluid at similar mass flow rates. In a similar finding, Said et al.²³⁷ reported an increase of 7% in the overall heat transfer coefficient and an increase of 11.39% in convective heat transfer and achieved a reduction of 6.81% in the area. Somasekhar et al.²³⁸ conducted a numerical study to know the heat transfer characteristics and pressure drop of an Al₂O₃/water nanofluid and distilled water in a shell and tube heat exchanger. An Al₂O₃/water nanofluid was revealed to be a better cooling medium than distilled water in the study; however, the use of an Al₂O₃/water nanofluid as a cooling medium increased the pressure drop on the tube side in comparison to the distilled water. Ullah et al.²³⁹ numerically investigated the heat transfer properties of TiO_2 /water and γ -Al₂O₃/water nanofluids in the STHEx. The influences of the Reynolds's number of suspended nanoparticle vol % and particle type were examined on the thermal properties. The results discovered the peak improvement of 41.8% in the convective heat transfer coefficient for Al₂O₃/water, whereas TiO_2 /water exhibits a maximum increase of 37%. According to the study, the introduction of nanofluids increases the pressure drop, and the explanation for this was attributed to the nanofluid's higher viscosity than the base fluid. Esfahani and Languri²⁴⁰ used graphene oxide as a nanofluid in the shell and tube heat exchanger and performed an exergy analysis. Exergy analysis revealed that raising the graphene oxide concentration from 0.01 to 0.1 wt % increased heat conductivity by 8.7% and 18.9% at 25 and 40 °C, respectively. Investigation showed a reduced level of exergy loss in the STHEx in both laminar and turbulent conditions by employing graphene oxide nanofluids. Lotfi et al.²⁴¹ experimentally studied the enhancement in heat transfer by using the MWCNT/water nanofluid in a horizontal STHEx. In comparison to the base fluid, the results show that heat transfer was improved because of the presence of MWCNT.

According to the findings of the preceding research, various nanoparticles had varying influences on the performances of the heat exchangers. Furthermore, the thermal performance of heat exchangers is affected by a variety of elements such as nanofluid characteristics, and structural-geometrical features.²⁴² In this case, machine learning might assist in resolving the difficulties of heat transfer and nanofluid flow, as well as providing realistic thermal design assistance to increase the efficiency of various types of heat exchangers. Various machine learning techniques have been employed to analyze the thermal and hydrodynamic performances of heat exchangers with nanofluids. Saeedan et al.¹⁹² proposed a neural network model to forecast the pressure drop and Nusselt number. The value of the Nusselt number was assessed concerning the maximum relative error and MSE values of 1.2596% and 0.409%, respectively. However, the pressure gradient value was predicted concerning maximum relative error and MSE values of 0.25% and 0.08%, respectively. Xie et al.²⁴³ applied the ANN for analyzing the heat transfer in the STHEx. For training and testing neural network configurations, lab-based experimental data were gathered. Overall heat transfer rates and temperature variations at the outlets on either side were anticipated. Different network designs were investigated to determine the best network for prediction, and ANN was shown to be superior. The maximum variation between experimental and predicted data was determined to be less than 2%. In a similar study, Wang et al.²⁴⁴ reported that the ANN approach is better than empirical correlations for predicting heat transfer rates

Table 6. Summar	y of Recent	Research on Heat Excl	hangers Utilizing	Nanofluids and Models Anti	icipated Using M	achine Learning Approaches	
References	Type of heat exchanger studied	NF used	ML technique used	Input	Output	Performance parameters	Outcome
Xie et al. ²⁴³	Shell-and- tube heat exchanger	I	ANN	Reynolds number, inlet temperature, total no. of tubes, diameter of center tube, total no. of baffles, baffle pitch	Ttemperature differ- ence, heat transfer rate	R = 0.9954	Prediction from ANN found to be much more precise than correla- tion
Hojjat ²⁴⁶	Shell-and- tube heat exchanger	TiO ₂ /water Al ₂ O ₃ /water	ANN	Nanoparticle volume concentration, Reynolds number, nanoparticle thermal conductivity, Prandtl number	Nusselt number, pressure difference	For Nusselt number, R = 0.9944, mean squared error (MSE) = 0.0009, For pressure drop, R = 0.9991, MSE 0.0002	9% average difference found be- tween experimental data and predicted values of Nusselt num- bers by ANN
Maddah et al. ²⁴⁷	Double-pipe heat ex- changer	SiO ₂ /water	ANN	Re, Pr, particle concentration, Tr, GPR	Exergetic efficiency	$R^{2} = 0.9779$	ANN determined optimal plan of operation for heat transfer proc- ess
Bahiraei et al. ²⁴⁸	Concentric tube heat exchanger	Cu nanoparticles and base fluids, carboxymethyl cel- lulose (CMC), water (0.4 wt %)	ANN-genetic algo- rithm (GA)	Radius ratio, volume concentration, particle size	Convective heat transfer coefficient, pressure drop	For heat transfer coefficient of inner wall, $\mathbb{R}^2 = 0.999$ MRE = 0.0110	ANN model estimated data accurately
Safikhani et al. ²⁴⁹	Flat tubes	$\rm Al_2O_3/water$	For modeling GMDH type ANN. NSGA II for optimization	Particle concentration and size, heat flux, volumetric flow rate	Convective heat transfer coefficient, pressure drop	For heat transfer coefficient R ² = 0.9802 MAPE = 0.2108	Comparison of results between ANN and CFD gave error less than 5%
Zdaniuk et al. ²⁵⁰	Straight tubes with inter- nal helical fins	I	ANN	Number of fins starts, helix angle, Re	Colburn j-factors, Fanning friction factors	For j, MSE = $7.7760 \text{ x}10^{-9}$ For f, MSE = $7.7760 \text{ x}10^{-9}$	ANN found superior to corre- sponding algebraic power law regressions
Wijayasekara et al. ²⁵¹	Compact heat ex- changer	I	EBaLM-OTR	Mass flow rate, hot and cold side inlet temperature, pressure	Pressure drop, heat transfer	Error less than $(10^{-5} \text{ to } 10^{-3})$	EBaLM-OTR used for overtraining resilience

and recommended that ANN be used to model heat exchangers and simulate thermal systems. Xie et al.²⁴⁵ The Nusselt numbers and friction factors of three types of fin-andtube heat exchangers were correlated using the ANN. Twelve geometric parameters were fed into ANN, and the outputs were the Nusselt number and friction factor. NN was trained, and the weights were adjusted using the well-known feedforward backpropagation method. The disparity between forecasts and experimental data was discovered to be less than 4%. Hojjat²⁴⁶ developed the ANN to forecast the thermal and hydrodynamic performance of nanofluids used in STHEx. Volume concentration, Reynolds number, nanoparticle thermal conductivity, and Prandtl number were used as the inputs in the ANN model. Deviations in experimental data and ANN results were almost 9% and 9.6% for Nusselt Number and pressure drop, respectively. In another study, Maddah et al.²⁴⁷ performed the exergetic efficiency and used the different concentrations of nanofluids with twisted tape in double-pipe heat exchangers. Exergetic efficiency predictions from experimental processes were difficult and time consuming. ANN was utilized to identify the correlation between the thermal and flow parameters and the exergetic efficiency. Bahiraei et al.²⁴⁸ investigated the convective heat transfer in STHEx. It was reported that the impact of altering concentration on pressure drop was greater than the impact of radius ratio and particle size. ANN was used to create a forecast model for convective heat transfer. GA in a combination compromise programming technique was used to determine the optimum conditions with the most heat transfer and the least pressure loss. Safikhani et al.²⁴⁹ employed the multiobjective optimization for $Al_2O_3/$ water nanofluid in the flat type tubes by using ANN, computational fluid dynamics (CFD), and nondominated sorting genetic algorithms (NSGA II). To obtain the correlation between the Fanning friction factors and Colburn j-factors for the flow of water in straight-type tubes, Zdaniuk et al.²⁵⁰ used the ANN. It was concluded that ANN was superior to its corresponding algebraic power law regressions. Wijayasekara et al.²⁵¹ highlighted the risk of overtraining or overlearning the network. Generalizing is the main goal of the network. The authors used the EBaLM-OTR (error backpropagation and Levenberg-Marquardt algorithms for overtraining resilience) method to overcome this problem. In this method, neural network architecture evaluates alternative network topologies based on mean square error (MSE) and standard deviation of MSE using a separate validation set. The approach has proven to be a valuable tool for identifying the flaws and benefits of various network topologies. The above literature survey showed that ANN is the most common machine learning technique used to predict different parameters in the heat exchangers. The survey also reveals that the training database may be enlarged to incorporate experimental and numerical data for improved prediction. Table 6 summarizes current work on heat exchangers utilizing nanofluids and models predicted using machine learning approaches.

5.2. Solar Energy Systems. A solar-based photovoltaic (PV) energy system is the most common form of renewable energy source, having a huge presence in energy markets.^{252,253} This is because the Sun's continuous solar energy of 1367 W/m² is distributed throughout the atmosphere. The total amount of energy absorbed by the planet from solar radiation at any instance is estimated as 1.8×10^{11} MW.⁵⁴ Solar's enormous potential for generating sustainable energy has piqued the

interest of policymakers, economists, governments, and environmental engineers.^{254,255} PV energy thus offers enormous promise for urban as well as rural electrification.^{256,257}

On the other hand, PV power generation is impacted by a variety of factors like humidity, weather, solar radiation, wind pressure, and ambient and module temperatures.²⁵⁸ Natural climatic fluctuations may change these factors, affecting the quantity of electricity generated. The power system's dependability, stability, and planning are all disrupted by a sudden shift in solar power production.²⁵⁹ To avoid such situations, accurate and exact solar power output forecasting is needed to maintain the power system's reliability, stability, and quality.²⁶⁰ It has the potential to decrease the grid's effect from power unpredictability. One of the most essential problems for the near future global energy supply will be the considerable integration of renewable energy sources (particularly solar) into existing or future energy supply architecture.²⁶¹ An electrical operator must maintain a perfect balance between power output and consumption at all times. In practice, the operator frequently faces challenges in maintaining this balance with traditional and controlled energy production systems, particularly in small or isolated electrical grids (as found in islands).²⁶²

Machine learning techniques such as SVM, ANN, extreme learning machine (ELM), GA, and GEP can model predict the generation and uncertainties in the solar energy domain. These models are useful for pattern recognition, classification, data mining, and forecasting because they can build a connection between inputs and outputs even when representation is difficult.^{263,264} Meenal and Selvakumar²⁶⁵ employed SVM, ANN, and empirical methods for the prediction of solar radiation. Modeling inputs were month, longitude, latitude, day length, brilliant sunlight hours, relative humidity, and temperature range. It was observed that month, peak temperature, latitude, and bright sunlight hours were the most affecting input parameters, whereas relative humidity is the least influential. SVM was superior to ANN and empirical models in prognostic modeling. Fan et al.²⁶⁶ also used SVM to study the influence of air pollution on solar radiation. Six common constituents of air pollution (NOx, SO2, O3, CO, $PM_{2.5}$, and PM_{10}) and air quality index (AQI) were used as input for modeling the diffuse and global solar radiations. The SVM model predicted that the AQI was the most influencing factor affecting both diffuse and global solar radiations. Eseye et al.²⁶⁷ used a hybrid approach of the PSO-SVM-wavelet method to forecast the photovoltaic (PV) solar power. The interactions of the PV system's real power record with numerical weather prediction (NWP) meteorological data over one year using a 1 h time increment were used for modeling. The hybrid approach demonstrated superior performance to standalone model prediction techniques. Fan et al.²⁶⁸ also used a hybrid approach to study air pollution's influence on daily diffuse solar radiation. SVM with PSO, BAT, and the whale optimization algorithm (WOA) were recommended for diffuse daily solar radiation prediction in airpolluted locations. SVM models outperformed extreme gradient boosting (XGBoost) and MARS models, while the BAT method was the most effective in improving the performance of solo SVM models. Lee et al.²⁶⁹ developed a prediction model for solar irradiance using ensemble machine learning techniques. A summary of recent studies on applications of nanofluids in solar thermal systems and

model prediction with machine learning techniques is depicted in Table 7.

5.3. Geothermal Energy. The shift from fossil-fuel-based systems to renewable-energy systems is becoming critical for sustainable energy development and environmental protection. Geothermal energy is a promising renewable type of energy and has the potential to reduce fossil fuel usage and minimize environmental impacts.²⁷⁵⁻²⁷⁷ Because of the growing importance of geothermal energy systems, many researchers reviewed the various aspects.^{278,279} In general, geothermal power plants have low operating costs^{280,281} but on the other hand have lower energy conversion efficiencies than typical thermal power plants.^{282,283} The reason is attributed to the relatively low temperature of geothermal resources.²⁸⁴ This is the major technological challenge that restricts the widespread application of geothermal energy.²⁸⁵ The efficiency of heat conversion could be raised by improving the heat transfer between the ground and heat-carrying fluid. Due to better thermophysical properties than the base fluids, nanofluids could be a good alternative for a heat carrier fluid.²⁸⁶

Daneshipour and Rafee²⁸⁷ applied Al₂O₃/water and CuO/ water nanofluids as the working fluids of a geothermal borehole heat exchanger. The Reynolds averaged Navier-Stokes (RANS) equations were numerically solved with the SST k-turbulence model to simulate the flow. The results reveal that the heat exchanger's overall water flow pressure loss is minimized at a particular diameter ratio. The findings also reveal that the CuO-water nanofluid extracts more heat than the alumina-water nanofluid, albeit at the expense of larger pressure losses and pumping powers. Diglio et al.²⁸⁸ numerically investigated the use of different nanofluids in the borehole heat exchanger. The optimal nanofluid is determined using a computational model based on energy and momentum balances. It is found in the study that copper has the biggest borehole thermal resistance decrease, reaching around 3.8% when a nanoparticle's volumetric concentration is 1%, but it also has the second-highest pressure drop. A summary of recent studies on applications of nanofluids in solar thermal systems and model prediction with machine learning techniques is depicted in Table 8.

5.4. Wind Energy. Wind power is a significant renewable energy source, and its growth is helpful in terms of energy supply security and lowering greenhouse gas emissions.^{293,294} The wind power curve plays an important role in estimating wind energy potential and wind turbine selection.²⁹⁵⁻²⁹⁷ Wind turbines must disperse a substantial quantity of heat during operation. The produced heat must be effectively distributed; otherwise, the temperature of the mechanical and electrical components would rise, lowering overall energy efficiency. The cooling system is used to dissipate the heat. Raj et al.²⁹⁸ employed nanofluid in the cooling system of the wind turbine. The use of nanofluid reduces the heat transfer surface and the flow rate of heat transfer fluid due to the greater thermal conductivity. De Risi et al.²⁹⁹ employed Al₂O₃/water nanofluid to increase the performance of the cooling system. This cooling system limits the thermal stresses in the electrical and mechanical components of the wind turbine. In their study, Álvarez-Regueiro et al.³⁰⁰ determind that the wind turbines having a capacity above 10 MW require a sophisticated cooling system. Empirical relation functions for the Nu number and Darcy friction factor were developed. The use of nanoenhanced industrial coolant enhanced the convective coefficient of heat transfer by 7% for the nanofluid with

Reference	Type of solar energy application and NF	ML technique used	Input	Output	Performance parameters	Outcome
Jamei et al. ¹⁴³	Generic solar thermal with molten (nitrate) salt-based nanofluids	AdaBoost, Extra Tree Regres- sion	Solid mass fraction, SHC of base fluids, temperature, density, mean diameter	SHC of NF	R= 0.9964, RMSE = 0.1566	SHC of BFs was most vital parameter in estimation of NF SHC
Jamei et al. ¹⁴⁷	Generic solar thermal with carbon and metal oxide based nanofluids	Gaussian process regression	Temperature and SHC of base fluids, mean diameter and solid volume fraction of nanoparticles	SHC of NF	R = 0.9997 RMSE = 0.01506	Mean diameter of nanoparticles and SHC of BFs was most vital parameter in SHC of NFs estimation
Cao et al. ²⁷⁰	Efficiency of thermal/PV collectors with nanocoo- lants	ANN, least square support vector regression, ANFIS	NFs properties, flow rate of coolant, solar radiation	Electric efficiency	$R^2 = 0.9534$, MSE = 2.548	Performance could be simulated efficiently
Ranjan Parida et al. ²⁷¹	Generic solar thermal with molten (nitrate) salt-based NFs	Principal component analysis (PCA), hierarchical cluster analysis (HCA)	Concentration, density ratio, temper- ature, nanoparticles size	SHC of NF	81.3% variations in entire data set	Mean diameter of nanoparticles was most vital parameter in SHC of NFs estima- tion
Ebrahimi-Moghadam et al. ²⁷²	Parabolic trough solar collec- tor with Al ₂ O ₃ and ethyl- ene glycol	ANN	Reynolds number, mean flow temper- ature, particle size	Optimal volume fraction	$R^2 = 0.9992,$ MSE = 5.92 × 10 ⁻⁹	Optimal volume fraction not influenced by Re number
Delfani et al. ²⁷³	Direct absorption solar col- lector with graphene oxide/ deionized water	ANN	Solar collector length and depth, fluid flow rate, concentration, temperature difference	Nusselt number, collector effi- ciency	For efficiency MAPE = 1.47%, for Nusselt number MAPE = 2.567%	Highly precise predictor model developed using ANN
Bahiraei et al. ²⁷⁴	Solar still with Cu ₂ O nano- particles	ANN, PSO-ANN, ANFIS	Fan power, nanoparticle volume frac- tion, glass, basin, water temperature	Energy efficiency	$R^2 = 0.9884$ for PSO-ANFIS	PSO-ANFIS hybrid approach superior in model prediction

Table 7. Summary of Recent Research Conducted on Solar Thermal Systems Using Nanofluids and Models Predicted with Machine Learning Techniques

Energy	&	Fue	ls
	-		

5	Jy c	x Fu	eis			
	Outcome	Helped in bore length reduction by 1.3%	GMDH-ANN could develop a robust predictive model	SVR performed better compared to ANN	ANN with NSGA II creates a robust meta model	Average Nusselt number enhancements up to 64%
	Performance parameters	1	$R^{2} = 0.99$	RMSE (ANN) = $0.002S$ RMSE (SVR) = 0.0023	MSE = 0.0008	RMSE = 0.924 $R^2 = 0.9921$
	Output	Convective heat transfer	SHC of NF	SHC of NF	TC and dynamic viscosity of NF	Nusselt number
	Input	Effective thermal conductivity, viscosity	Temperature, solid vol %, SHC of base fluids, mean diameter of nanoparticles	Solid volume fraction, temperature	Concentration, temperature	Angular rotational velocity, Richardson number, nanoparticle volume fraction
	ML technique used	Multiobjective flower pollination algorithm	GMDH-ANN	ANN, SVR	ANN + NSGA II	ANFIS
	Nanofluids	Al ₂ O ₃ + water	Metallic nanoparticles in water	CuO + water	Al ₂ O ₃ + water	CNT + water
	Reference	Narei et al. ²⁸⁹	Ramezanizadeh and Nazari ²⁹⁰	Alade et al. ¹⁴¹	Hemmat Esfe et al. ²⁹¹	Tiwari et al. ²⁹²

Table 8. Recent Studies Conducted on Geothermal Energy Systems Using Nanofluids and Models Predicted with Machine Learning Techniques

nanoadditive loading of 0.25 wt %. Rostamzadeh and Rostami²⁵ employed the waste heat captured from the cooling system of the wind turbine for the desalination process. Cu, CuO, TiO₂, Al₂O₃, and SiO₂ nanoparticles were employed with a water base fluid. A Cu/water mixture had the best performance of all the nanoparticles used because it produces more freshwater, whereas the SiO₂/water mixture has the lowest efficiency. A summary of recent work in the domain of model prediction/numeric simulations for wind energy systems using nanofluids is listed in Table 9.

5.5. Other Renewable Energy Systems. Modern machine learning methods powered by high-speed computational machines are making huge differences in several renewable energy fields. Biomass is currently the world's most extensively utilized renewable energy source. It is mostly utilized as a solid fuel, with liquid fuels and gas being used to a lesser amount.^{302,303} The utilization of biomass for energy generation has risen slowly in recent years. Rural and off-grid regions rely heavily on biomass for energy.³⁰⁴ Biomass is utilized to satisfy a wide range of energy requirements, including producing electricity, powering cars, heating homes, and supplying process heat for industrial operations. Wood, animal, and plant wastes all have biomass potential.^{305,306} Biomass can be directly used as fuel, although direct combustion is highly inefficient and polluting. The biomass can be converted to fuel in liquid or gaseous.^{307,308} Among the biomass conversion methods, pyrolysis has a lot of promise.^{309,310} Thermochemical conversion, which includes combustion, pyrolysis, torrefaction, hydrothermal liquefaction, and also gasification, is the most practical method for converting solid biomass into biofuel.^{311,312} By optimizing process parameters, the fundamental objective of thermochemical conversion is to remove unwanted byproducts. Pyrolysis is a potential technology for converting biomass into biofuel in an inert environment at high temperatures (250-600 °C).^{313,314} Pyrolysis technology is being used to produce biobased fuels and chemicals from biomass, which is a relatively new technique. Bio-oil, biochar, and syngas are the end products of biomass pyrolysis, and they might be used for energy generation or other applications with exceptional characteristics such as environmental friendliness, cheap prices, and biodegradability.^{310,315}

The process of biomass conversion is highly nonlinear. Several researchers applied machine learning methods such as ANN³¹⁶ and GEP³¹⁷ to model predict the biomass conversion to usable fuel, biofuels, and its byproducts. Safarian et al.³¹⁸ employed the ANN technique for model prediction of a downdraft biomass gasification-integrated power generating unit. The aim was to estimate the net output power of systems using biomass feedstocks under diverse operating conditions and atmospheric pressure. Compositions of feedstock, proximate type analysis (ash, moisture, fixed carbon, and volatile material), and operational parameters were utilized as input parameters. The ANN model exhibited robust prediction with observed data with R² greater than 0.999. Mutlu and Yucel³¹⁹ used machine learning classifiers for the prediction of gasification products. Two types of classifiers were used: binary least squares support vector machine and multiclass random forests classifiers. The suggested methods were developed and validated using 10-fold cross-validation on 5237 data samples, with binary and multiclass classifiers achieving prediction accuracy values of over 96% and 89%, respectively.

Energy & Fuels

Reference	Nanofluids	ML technique used	Input	Output	Performance parameters	Outcome
Álvarez-Regueiro et al. ³⁰⁰	Havoline XLC Premixed 50/50	Dimensionless analysis- based estimation	Nanoadditive loading, temperature, Reynolds number	Convective heat transfer coefficient, pressure drop	Nu number, Darcy function	Peak coefficient of heat transfer increased by 7% for 0.25% nanofluid concentration
Maleki et al. ¹³⁰	ZnO + water	ANN, multivariate adaptive regression spline (MARS)	Base fluid TC, concentration, temperature, nanoparticle size	TC of NF	R^{2} (ANN) = 0.9987, R^{2} (MARS) = 0.9879	ANN performed better compared to MARS
Komeilibirjandi et al. ¹³¹	CuO with water	GMDH-ANN	Base fluid TC, concentration, temperature, nanoparticle size	TC of NF	$R^2 = 0.9996$, AARD = 0.881%	GMDH-ANN could develop a robust predictive model
Adun et al. ³⁰¹	Al ₂ O ₃ -ZnO_Fe ₃ O ₄	Gaussian process regression	Concentration, mixture ratio, temperature,	TC, dynamic viscosity of NF	R^{2} (TC model = 0.9987, R^{2} (viscosity model) = 0.034	ANN with NSGA II creates robust metamodel

Baruah et al.³²⁰ developed an ANN-based biomass gasification model using ANN to forecast the product gas composition in terms of concentrations of four key gas species: CH₄, CO₂, CO, and H₂ percentwise. Biomass composition and reduction zone temperature were utilized as input parameters in the models. The output of the ANN models was found to agree with experimental data with absolute fractions of variance (R^2) more than 0.99 for CH₄ and CO models and greater than 0.98 for CO₂ and H₂ models. An ANN model was created by Kargbo et al.³²¹ utilizing experimental data to forecast the gasification process. The study's objective was to use AI-based model prediction, resulting in saving time and money in the development and testing process. With a correlation of $R^2 > 0.99$, the ANNbased model could reliably forecast gas composition and yield in response to output changes. Aghbashlo et al.³²² used a hybrid ANFIS-PSO approach to predict lignocellulose pyrolysis kinetic constants. The PSO technique was employed to fine-tune the membership function parameters of the ANFIS model. With an $R^2 > 0.970$ and a MAPE = 3.27%, the generated models were able to accurately predict the pyrolysis behaviors of three distinct biomass feedstocks with $R^2 > 0.91$, indicating their fidelity. Elmaz et al.³²³ employed four regression techniques: support vector machine, polynomial regression, decision tree regression, and ANN to forecast CO, H₂, CO₂, CH4, and high heating values from the biomass gasification process. To avoid multicollinearity and enhance the computing efficiency, the extracted features are subjected to principal component analysis. The ANN and decision tree regression outperformed others by attaining $R^2 > 0.9$ for the majority of outputs.

In the field of renewable energy and, more specifically, the solar energy domain, battery energy storage systems are the basic necessity. The uncertainties in optimal scheduling and operation of a renewable energy system need efficient machine learning techniques for modeling.³²⁴ Masoumi et al.³²⁵ successfully used an ANN-PSO hybrid approach to model and predict the uncertainty in optimal renewable energy allocations combined with battery energy storage. The researchers integrated the modeling of solar-, wave-, and wind-based renewable energy systems. A battery storage system was employed to compensate for any uncertainty caused by the renewable energy system. It was concluded that the use of electric vehicles and their aggregators could assist system operators in maximizing their benefits in terms of power quality and market control. In another study by Gayen and Jana,³²⁶ the ANFIS-based control strategy improves the overall control capabilities for microgrid-connected battery energy storage systems. Several simulation studies were done on a 230 V, 50 Hz single-phase ac grid linked 100 Ah, 120 V battery using MATLAB-SIMULINK 2014b software to test the enhanced control approach. Using ANFIS, the efficiency of a coordinated regulating strategy for transferring power between a battery and single-phase type electrical grid was enhanced. A summary of recent work in model prediction for the various renewable energy systems is listed in Table 10.

6. CHALLENGES AND PERSPECTIVES

The application of machine learning has grown in importance as a technique for studying system behavior and evaluating system performance. It is increasingly being applied in nanofluids and renewable energy systems, although its potential and prospects are yet to be completely realized.

Reference	ML technique used	Input	Output	Performance parameters	Outcome
Sharma ³¹⁶	ANN, GEP	Engine load, fuel blending ratio, injection parameters	Brake thermal efficiency, brake specific energy consumption,	R (GEP) = 0.997 R (ANN) = 0.9998	ANN marginally superior to GEP in model prediction
Sharma ³¹⁷	GEP	Engine load, fuel blending ratio, injection parameters	Brake thermal efficiency, brake specific energy consumption, emission	$R = 0.9926$ $R^2 = 0.9854, RMSE = 0.0048$	GEP-based model robust enough to predict engine output
Safarian et al. ³¹⁸	ANN	Ash, moisture, volatile material, and fixed carbon, operational parameters	Net power output	$R^2 = 0.999$	ANN model exhibited agreement with observed data
Mutlu and Yucel ³¹⁹	Binary LS-SVM, multiclass RF	Biomass type, temperature	Syngas composition	Prediction accuracy (LS-SVM) = 96%, prediction accuracy (RF) = 89%	Binary LS-SVM superior to RF in prediction of syngas composition
Baruah et al. ³²⁰	ANN	Biomass composition, reduction zone temperature	Producer gas composition	$R^2 = 0.9991$	ANN model exhibited agreement with observed data
Kargbo et al. ³²¹	ANN	Biomass composition, operating conditions	Yield and gas composition	$R^2 > 0.992$	ANN model exhibited agreement with observed data
Aghbashlo et al. ³²²	ANFIS with PSO	Pyrolysis kinetics, process heating rate	Pyrolysis behavior	$R^2 > 0.97$	Lignocellulose pyrolysis behavior could be reliably estimated using ANFIS-PSO
Elmaz et al. ³²³	Polynomial regression, SVM, ANN, decision trees	Waste composition, operating conditions	CO, CH ₄₁ , CO ₂ , H ₂ , higher heating value of output gas	$R^{2} > 0.9$	MLP-ANN superior to other three machine learning techniques
Masoumi et al. ³²⁵	ANN with PSO	Uncertainty in renewable energy system	Renewable energy allocation combined with battery energy storage	$R^{2} > 0.99$	Metamodel of uncertainty in renewable energy and battery storage developed
Gayen and Jana ³²⁶	ANFIS	Charging and discharging of battery storage system	Control of charging and discharging of battery storage system	$R^{2} > 0.9$	ANFIS improved overall control capability

Table 10. Recent Research Conducted on Various Renewable Energy Systems Model Predicted with Machine Learning Techniques

The challenges of modeling and simulation are briefly discussed in this section and how they may be overcome.

Data quality and lack of data, machine learning parameter tuning, technical infrastructure problems, lack of trained specialists, integration issues, hazards, compliance issues, and legal concerns are some of the bottleneck barriers to AI adoption in the smart energy sector.³²⁷ Low-quality controllers, sensors, and controlled devices are used in energy system operation and data estimates.^{328,329} The solution to most of the problems here can involve trained manpower from data collection to efficient modeling and result presentations. Data quality can be improved by the application of enhanced sensors and modern data acquisition systems. Technical infrastructure and integration require investments that are just a small fraction of renewable energy project costs.^{330,331} Compliance and legal hurdles need implementation of longterm sustainable policy in this sector.³³² There may be challenges in this sector, but their mitigation can result in much more economic gains for the stakeholders.³³³

The shortcoming of adequate scientific focus while building more extensive integrated modeling systems sometimes makes integrating energy models challenging. Additionally, as energy systems grow decentralized, they become more complex and linked, relying on a variety of energy sources with increasingly interwoven boundaries. Because of these increases in complexity, certain existing energy models can no longer adequately address all energy optimization issues.^{334,335} As a result, the complexity of the modeling domain in RSES must be addressed as soon as possible. Complex energy systems do not lend themselves to concise representations. As a result, the complexity of energy models in RSES is generally countered by the fact that the correctness of a model is determined by the basic assumptions incorporated into its design.³³⁶ It was discovered that research on energy system modeling and simulation has continuously utilized modeling techniques with big data and hourly profiles of energy use, making RSES models challenging to solve in instances when substantial system and process data are not available.^{337,338}

Many experts have a thorough understanding of technical issues. However, skilled people to create trustworthy AI-powered apps with real-world advantages are exceedingly difficult to come by. Even though power companies collect and retain data, it is difficult to digitize it with modern management software. Data loss, improper configuration, device failure, and illegal access are all risks that are linked.³³⁹ The most significant impediment to the energy sector's modernization is outdated infrastructure. Utility businesses are currently engulfed in a plethora of data that they generate and have no clue how or when to deal with it. Despite having more data than anybody else, the data are scattered, unorganized, distributed across several formats, and kept locally. While the business makes a lot of money, it is also vulnerable to antiquated systems.^{61,114}

It is evident from the literature that challenges and bottlenecks exist in effective implementations of machine learning methods in nanofluid-based renewable energy systems. However, ever-increasing computational power, the advent of intelligent machine learning techniques, and the availability of an AI trained workforce demonstrate the high growth aspect of this domain.³⁴⁰ The voluminous research published in the last five years in various reputable energy journals substantiates this fact. The recently published

literature review shows that researchers are meeting challenges in modeling complex energy systems with greater fierceness.

7. CONCLUSIONS

Several machine learning algorithms have been used in recent times to evaluate the performances of renewable energy systems working with nanofluids. This work focuses on a literature review to highlight current advances in machine learning research in the use of various machine learning algorithms for nanofluid-based renewable energy systems. Even though hundreds of archival papers are published on this subject each year, many essential elements, together with their nonlinear effects, complicate nanofluid research to new heights and restrict its practical usefulness. Surprisingly, sophisticated artificial intelligence has not been used to solve complicated challenges. Nonetheless, machine learning has the potential to be beneficial and cost effective in nanofluid research, enabling potentially dependable practice in solar collectors, solar-based desalination system, biomass gasification, wind energy, and PV/T systems. This review also acknowledges the increased application of the hybrid approach by combining the machine learning-based modeling and optimization approach in the last five years. Some noteworthy conclusions of this study are as follows:

- (1) Several factors influence nanofluid thermophysical properties. Evaluating the thermophysical characteristics of various nanofluids across a wide range of parameters and flow conditions takes time and effort. Theoretical and empirical correlations may be quite ambiguous due to underlying assumptions. Many studies have shown that machine learning approaches like ANN can accurately predict nanofluid characteristics.
- (2) Nanofluids improve heat transmission in renewable energy sources, including biomass, solar, geothermal, and wind. Machine learning approaches can accurately anticipate the performance of a renewable energy system based on nanofluids.
- (3) The ANN and ANFIS predictions using modern commercial software are easy to implement and popular. The prognostic capability can be further enhanced using a hybrid approach with additional intelligent optimization techniques such as RSM, PSO, marine predator algorithm, genetic algorithm, swarm intelligence optimization, and more.

Newer ensemble machine learning algorithms, including boosted regression, K-means, K-nearest neighbor (KNN), CatBoost, and XGBoost, are gaining popularity due to their improved architectures and adaptabilities to a wide range of data sets. The ANN, ANFIS, and GEP techniques are mainly black-box methods, with the user having no idea how they work. This is a source of concern, and ethical artificial intelligence is required.

AUTHOR INFORMATION

Corresponding Authors

- Anh Tuan Hoang Institute of Engineering, HUTECH University, Ho Chi Minh City 717000, Vietnam; orcid.org/0000-0002-1767-8040; Email: hatuan@ hutech.edu.vn
- Zafar Said Department of Sustainable and Renewable Energy Engineering, University of Sharjah, 27272 Sharjah, United Arab Emirates; U.S.-Pakistan Center for Advanced

Studies in Energy (USPCAS-E), National University of Sciences and Technology (NUST), Islamabad 44000, Pakistan; Email: zsaid@sharjah.ac.ae

- Prabhakar Sharma School of Engineering Sciences, Delhi Skill and Entrepreneurship University, Delhi 110089, India; orcid.org/0000-0002-7585-6693; Email: psharmahal@ gmail.com
- Xuan Phuong Nguyen PATET Research Group, Ho Chi Minh City University of Transport, Ho Chi Minh City 717000, Vietnam; Email: phuong@ut.edu.vn

Authors

- Anurag Kumar School of Engineering Sciences, Delhi Skill and Entrepreneurship University, Delhi 110089, India Sandro Nižetić – University of Split, FESB, 21000 Split, Croatia
- Ashok Pandey Centre for Innovation and Translational Research, CSIR-Indian Institute of Toxicology Research, Lucknow 226 001 Uttar Pradesh, India; Sustainability Cluster, School of Engineering, University of Petroleum and Energy Studies, Dehradun 248 007 Uttarakhand, India; Centre for Energy and Environmental Sustainability, Lucknow 226 029 Uttar Pradesh, India; Orcid.org/0000-0003-1626-3529
- **Zuohua Huang** State Key Laboratory of Multiphase Flow in Power Engineering, Xi'an Jiaotong University, Xi'an 710049, P. R. China
- Asif Afzal Department of Mechanical Engineering, P. A. College of Engineering (Affiliated to Visvesvaraya Technological University, Belgavi), Mangaluru 574153, India
- Changhe Li School of Mechanical and Automotive Engineering, Qingdao University of Technology, Qingdao 266520, China
- Anh Tuan Le − School of Mechanical Engineering, Hanoi University of Science and Technology, Hanoi 10999, Vietnam; [©] orcid.org/0000-0003-4609-0382
- Viet Dung Tran PATET Research Group, Ho Chi Minh City University of Transport, Ho Chi Minh City 717000, Vietnam

Complete contact information is available at: https://pubs.acs.org/10.1021/acs.energyfuels.2c01006

Notes

The authors declare no competing financial interest.

Biographies

Prabhakar Sharma graduated from the Birla Institute of Technology, Mesra, India, and received a Ph.D. from the DCR University of Science and Technology, Murthal, India. At present, he is an assistant professor with the Delhi Skill and Entrepreneurship University, Delhi, India. He has research interests in applied machine learning for hybrid nanofluid-based renewable energy systems. He has been working on modeling optimization using machine learning biomass gasification, hybrid nanofluids, and dual-fuel engines.

Zafar Said is an associate professor with the Department of Sustainable Renewable Energy Engineering, University of Sharjah. He works in the fields of renewable energy, solar energy, heat transfer, nanofluids, PCM, and AI. He has published over 170+ papers as per Scopus, including in *Progress in Energy and Combustion, Physics Reports, Nano*, and *Renewable and Sustainable Energy Reviews*, and has two books/book chapters and 25 conference papers, with about 6400+ citations and an H-index of 45. Anurag Kumar completed his bachelor's degree in mechanical engineering and master of technology from the Indian Institute of Technology, Roorkee. He has worked in the System Engineering Department at Kirloskar Brothers Limited. He is presently an assistant professor in the Mechanical Engineering Department of Delhi Skill Entrepreneurship University, Delhi. His core research areas are renewable energy, machine learning, and hydroenergy.

Sandro Nižetić is a professor at the Faculty of Electrical Engineering, Mechanical Engineering and Naval Architecture (FESB) at the University of Split, Croatia. He has experience in theoretical and applied thermodynamics, HVAC systems, energy efficiency in buildings, rational usage of energy, and utilization of renewable energy sources.

Ashok Pandey is currently a distinguished scientist of India. His major research and technological development interests are industrial and environmental biotechnology and energy biosciences. Professor Pandey is an adjunct/visiting professor/scientist in universities in many countries in the world and also in several universities in India. He has ~1600 publications/communications, which include 16 patents, 107 books, and ~900 papers and book chapters, with an H-index of 119 and ~62,000 citations (Google scholar).

Anh Tuan Hoang is an associate professor at the Ho Chi Minh City University of Technology (HUTECH), Vietnam. He received his Ph. D in 2016. His research interests lie in the fields of energy conservation, renewable energy resources, alternative fuels, and internal combustion engines. He has published over 140 peerreviewed papers in international journals. He received the 2018 Highly Cited Review Article Award in Applied Energy, Elsevier.

Zuohua Huang is a Professor at Xi'an Jiaotong University, China. He received his Ph.D. in 1994. He is the vice-director of the Combustion Section of the Chinese Society of Engineering Thermo-physics. He is the principal editor of *FUEL* and Fellow of the Combustion Institute, and Director of the Board of the Combustion Institute. His research interests are combustion and energy science. He has published over 400 papers in international journals and is a highly cited scholar in engineering.

Asif Afzal is a researcher at the P. A. College of Engineering, Mangalore, India. He works in the fields of CFD analysis, battery thermal management, optimization, parallel CFD, and biofuels.

Changhe Li is a professor at the Qingdao University of Technology, China. He is a special expert of Taishan Scholars in Shandong Province, China. He received his Ph.D. degree from Northeastern University in 2006. His current research interests focus on intelligent and clean precision manufacturing. He has published 153 SCI/EI papers and won the China Patent Excellence Award (two items) and Shandong Province Technical Invention first prize.

Le Anh Tuan obtained a PhD. degree in 2005 from TU Graz, majoring in mechanical engineering. Currently, he is a full professor, a senior lecturer at Hanoi University of Science and Technology (HUST), and Chairman of the HUST university council. His research interests include fuels, fuel additives and emissions of vehicles, ICE performance, air pollution, electric vehicle technology, and related policy on low-carbon transport.

Xuan Phuong Nguyen is an associate professor of Ho Chi Minh City University of Transport, Viet Nam. He received a Ph.D degree in 2011. His research interests lie in the fields of energy conservation, renewable energy resources, maritime safety, and engineering transport. He has been leading the PATET Research Group at the University since 2019. He is the author of close to 70 publications withan H-index of 18 (Scopus) and several patents. Viet Dung Tran obtained a PhD. degree in 2022 from Ho Chi Minh City University of Transport, Vietnam. His research interests include energy conservation, renewable energy resources, transport management, alternative fuels, and internal combustion engines.

ABBREVIATIONS

EG Ethylene glycol Al₂O₃ Alumina MWCNT Multiwalled carbon nanotubes US EIA United States Energy Information Administration RCSES Research Centre in Systems Ecology and Sustainability ANN Artificial neural network BP Backpropagation

GEP Gene expression programming ET Expression trees LS-SVM Least square support vector machine QP Quadratic programming RBF Radial basis function SHC Specific heat capacity MARS Multivariate adaptive regression splines MSE Mean squared error SCG Scaled conjugate gradient MAPE Mean absolute percentage error CNF Carbon nanofiber LWLR Local weighted linear regression RMSE Root mean squared error SVR Support vector regression STHEx Shell and tube heat exchanger ELM Extreme learning machine PM Particulate matter WOA Whale optimization algorithm NFs Nanofluids ZnO Zinc oxide ML Machine learning SDGs Sustainable development goals RE Renewable energy MLP Multilayer perceptron ANFIS Adaptive neuro-fuzzy inference system GA Genetic algorithm GP Gene programming SVM Support vector machine RNN Recurrent neural network TC Thermal conductivity GMDH Group method of data handling MMD Maximum margin of deviation RSM Response surface methodology LM Levenberg–Marquardt F-CNF Functionalized carbon nanofiber PSO Particle swarm optimization LGP Linear genetic programming GPR Gaussian process regression DPHE Double pipe heat exchanger PHE Plate type heat exchanger AQI Air quality index NWP Numerical weather prediction XGBoost Extreme gradient boosting

REFERENCES

(1) Hoang, A. T.; Foley, A. M.; Nizetic, S.; Huang, Z.; Ong, H. C.; Olcer, A. I.; Pham, V. V.; Nguyen, X. P. Energy-Related Approach for Reduction of CO2 Emissions: A Strategic Review on the Port-to-Ship Pathway. *J. Clean. Prod.* **2022**, *355*, 131772. (2) Nguyen, X. P.; Hoang, A. T.; Ölçer, A. I.; Huynh, T. T. Record Decline in Global CO 2 Emissions Prompted by COVID-19 Pandemic and Its Implications on Future Climate Change Policies. *Energy Sources, Part A Recover. Util. Environ. Eff.* **2021**, 1–4.

(3) Bakır, H.; Ağbulut, Ü.; Gürel, A. E.; Yıldız, G.; Güvenç, U.; Soudagar, M. E. M.; Hoang, A. T.; Deepanraj, B.; Saini, G.; Afzal, A. Forecasting of Future Greenhouse Gas Emissions Trajectory for India Using Energy and Economic Indexes with Various Metaheuristic Algorithms. J. Clean. Prod. **2022**, 360, 131946.

(4) A Global Breakdown of Greenhouse Gas Emissions by Sector. *Transport Energy Strategies*. https://www.transportenergystrategies. com/2020/11/09/a-global-breakdown-of-greenhouse-gas-emissions-by-sector/ (accessed Jan 13, 2022).

(5) Ahmadi, M. H.; Ghazvini, M.; Sadeghzadeh, M.; Alhuyi Nazari, M.; Ghalandari, M. Utilization of Hybrid Nanofluids in Solar Energy Applications: A Review. *Nano-Structures & Nano-Objects* **2019**, *20*, 100386.

(6) Ranganathan, P.; Pandey, A. K.; Sirohi, R.; Tuan Hoang, A.; Kim, S.-H. Recent Advances in Computational Fluid Dynamics (CFD) Modelling of Photobioreactors: Design and Applications. *Bioresour. Technol.* **2022**, 350, 126920.

(7) Hoang, A. T.; Pham, V. V.; Nguyen, X. P. Integrating Renewable Sources into Energy System for Smart City as a Sagacious Strategy towards Clean and Sustainable Process. *J. Clean. Prod.* **2021**, *305*, 127161.

(8) Hoang, A. T.; Nizetic, S.; Olcer, A. I.; Ong, H. C.; Chen, W.-H.; Chong, C. T.; Thomas, S.; Bandh, S. A.; Nguyen, X. P. Impacts of COVID-19 Pandemic on the Global Energy System and the Shift Progress to Renewable Energy: Opportunities, Challenges, and Policy Implications. *Energy Policy* **2021**, *154*, 112322.

(9) Tiwari, D.; Sherwani, A. F.; Kumar, N. Optimization and Thermo-Economic Performance Analysis of Organic Rankine Cycles Using Mixture Working Fluids Driven by Solar Energy. *Energy Sources, Part A Recover. Util. Environ. Eff.* **2019**, *41* (15), 1890–1907.

(10) Gupta, V. K.; Kumar, S.; Kukreja, R. Comparative Thermal Performance Enhancement Study of a Reverse-Irradiated and Direct-Irradiated Direct Absorption Solar Collector Using Silver Nanofluid. *Energy Sources, Part A Recover. Util. Environ. Eff.* **2022**, 44 (2), 3823–3835.

(11) Mahian, O.; Bellos, E.; Markides, C. N.; Taylor, R. A.; Alagumalai, A.; Yang, L.; Qin, C.; Lee, B. J.; Ahmadi, G.; Safaei, M. R.; Wongwises, S. Recent Advances in Using Nanofluids in Renewable Energy Systems and the Environmental Implications of Their Uptake. *Nano Energy* **2021**, *86*, 106069.

(12) Ghafurian, M. M.; Niazmand, H.; Ebrahimnia-Bajestan, E.; Elhami Nik, H. Localized Solar Heating via Graphene Oxide Nanofluid for Direct Steam Generation. *J. Therm. Anal. Calorim.* **2019**, *135* (2), 1443–1449.

(13) Yang, L.; Ji, W.; Mao, M.; Huang, J.-n. An Updated Review on the Properties, Fabrication and Application of Hybrid-Nanofluids along with Their Environmental Effects. *J. Clean. Prod.* **2020**, *257*, 120408.

(14) Xiong, Q.; Altnji, S.; Tayebi, T.; Izadi, M.; Hajjar, A.; Sundén, B.; Li, L. K. B. A Comprehensive Review on the Application of Hybrid Nanofluids in Solar Energy Collectors. *Sustain. Energy Technol. Assessments* **2021**, *47* (May), 101341.

(15) Okonkwo, E. C.; Wole-Osho, I.; Almanassra, I. W.; Abdullatif, Y. M.; Al-Ansari, T. An Updated Review of Nanofluids in Various Heat Transfer Devices. *J. Therm. Anal. Calorim.* **2021**, *145*, 2817–2872.

(16) Eshgarf, H.; Kalbasi, R.; Maleki, A.; Shadloo, M. S.; Karimipour, A. A Review on the Properties, Preparation, Models and Stability of Hybrid Nanofluids to Optimize Energy Consumption. *J. Therm. Anal. Calorim.* **2021**, *144* (5), 1959–1983.

(17) Farooq, U.; Patil, A.; Panjwani, B.; Simonsen, G. Review on Application of Nanotechnology for Asphaltene Adsorption, Crude Oil Demulsification, and Produced Water Treatment. *Energy Fuels* **2021**, 35 (23), 19191–19210. (18) Hemmat Esfe, M.; Bahiraei, M.; Mir, A. Application of Conventional and Hybrid Nanofluids in Different Machining Processes: A Critical Review. *Adv. Colloid Interface Sci.* 2020, 282, 102199.

(19) Said, Z.; Sundar, L. S.; Rezk, H.; Nassef, A. M.; Ali, H. M.; Sheikholeslami, M. Optimizing Density, Dynamic Viscosity, Thermal Conductivity and Specific Heat of a Hybrid Nanofluid Obtained Experimentally via ANFIS-Based Model and Modern Optimization. *J. Mol. Liq.* **2021**, 321, 114287.

(20) Tang, W.; Zou, C.; Peng, H.; Wang, Y.; Shi, L. Influence of Nanoparticles and Surfactants on Stability and Rheological Behavior of Polymeric Nanofluids and the Potential Applications in Fracturing Fluids. *Energy Fuels* **2021**, *35* (10), 8657–8671.

(21) Zheng, D.; Wang, J.; Chen, Z.; Baleta, J.; Sundén, B. Performance Analysis of a Plate Heat Exchanger Using Various Nanofluids. *Int. J. Heat Mass Transfer* **2020**, *158*, 119993.

(22) Bao, B.; Wang, Q. Bladeless Rotational Piezoelectric Energy Harvester for Hydroelectric Applications of Ultra-Low and Wide-Range Flow Rates. *Energy Convers. Manag.* **2021**, *227*, 113619.

(23) Jakhar, S.; Paliwal, M. K.; Purohit, N. Assessment of Alumina/ Water Nanofluid in a Glazed Tube and Sheet Photovoltaic/Thermal System with Geothermal Cooling. *J. Therm. Anal. Calorim.* **2022**, 147, 3901–3918.

(24) Khoshgoftar Manesh, M. H.; Mousavi Rabeti, S. A.; Nourpour, M.; Said, Z. Energy, Exergy, Exergoeconomic, and Exergoenvironmental Analysis of an Innovative Solar-Geothermal-Gas Driven Polygeneration System for Combined Power, Hydrogen, Hot Water, and Freshwater Production. *Sustain. Energy Technol. Assessments* **2022**, *51*, 101861.

(25) Rostamzadeh, H.; Rostami, S. Performance Enhancement of Waste Heat Extraction from Generator of a Wind Turbine for Freshwater Production via Employing Various Nanofluids. *Desalination* **2020**, *478*, 114244.

(26) Peera, S. G.; Koutavarapu, R.; Akula, S.; Asokan, A.; Moni, P.; Selvaraj, M.; Balamurugan, J.; Kim, S. O.; Liu, C.; Sahu, A. K. Carbon Nanofibers as Potential Catalyst Support for Fuel Cell Cathodes: A Review. *Energy Fuels* **2021**, *35* (15), 11761–11799.

(27) Khalid, S.; Zakaria, I.; Azmi, W. H.; Mohamed, W. A. N. W. Thermal-Electrical-Hydraulic Properties of Al2O3-SiO2 Hybrid Nanofluids for Advanced PEM Fuel Cell Thermal Management. *J. Therm. Anal. Calorim.* **2021**, *143* (2), 1555–1567.

(28) Mehdipour, M.; Keshavarz, P.; Rahimpour, M. R. Rotating Liquid Sheet Contactor: A New Gas-Liquid Contactor System in CO2 Absorption by Nanofluids. *Chem. Eng. Process. - Process Intensif.* **2021**, *165*, 108447.

(29) Sajid, M. U.; Bicer, Y. Nanofluids as Solar Spectrum Splitters: A Critical Review. *Sol. Energy* **2020**, 207 (July), 974–1001.

(30) Tyagi, P. K.; Kumar, R.; Said, Z. Recent Advances on the Role of Nanomaterials for Improving the Performance of Photovoltaic Thermal Systems: Trends, Challenges and Prospective. *Nano Energy* **2022**, *93*, 106834.

(31) Moldoveanu, G. M.; Minea, A. A.; Huminic, G.; Huminic, A. Al2O3/TiO2 Hybrid Nanofluids Thermal Conductivity: An Experimental Approach. J. Therm. Anal. Calorim. 2019, 137 (2), 583-592.

(32) Wole-Osho, I.; Okonkwo, E. C.; Kavaz, D.; Abbasoglu, S. An Experimental Investigation into the Effect of Particle Mixture Ratio on Specific Heat Capacity and Dynamic Viscosity of Al2O3-ZnO Hybrid Nanofluids. *Powder Technol.* **2020**, *363*, 699–716.

(33) Xian, H. W.; Sidik, N. A. C.; Saidur, R. Impact of Different Surfactants and Ultrasonication Time on the Stability and Thermophysical Properties of Hybrid Nanofluids. *Int. Commun. Heat Mass Transfer* **2020**, *110*, 104389.

(34) Kakavandi, A.; Akbari, M. Experimental Investigation of Thermal Conductivity of Nanofluids Containing of Hybrid Nanoparticles Suspended in Binary Base Fluids and Propose a New Correlation. *Int. J. Heat Mass Transfer* **2018**, *124*, 742–751.

(35) Mashhadian, A.; Heyhat, M. M.; Mahian, O. Improving Environmental Performance of a Direct Absorption Parabolic Trough Collector by Using Hybrid Nanofluids. *Energy Convers. Manag.* 2021, 244, 114450.

(36) Ahmed, W.; Kazi, S.N.; Chowdhury, Z.Z.; Johan, M. R. B.; Mehmood, S.; Soudagar, M. E. M.; Mujtaba, M.A.; Gul, M.; Ahmad, M. S. Heat Transfer Growth of Sonochemically Synthesized Novel Mixed Metal Oxide ZnO+Al2O3+TiO2/DW Based Ternary Hybrid Nanofluids in a Square Flow Conduit. *Renew. Sustain. Energy Rev.* 2021, 145, 111025.

(37) Tiwari, A. K.; Kumar, V.; Said, Z.; Paliwal, H.K. A Review on the Application of Hybrid Nanofluids for Parabolic Trough Collector: Recent Progress and Outlook. *J. Clean. Prod.* **2021**, *292*, 126031.

(38) Mourad, A.; Aissa, A.; Said, Z.; Younis, O.; Iqbal, M.; Alazzam, A. Recent Advances on the Applications of Phase Change Materials for Solar Collectors, Practical Limitations, and Challenges: A Critical Review. *J. Energy Storage* **2022**, *49*, 104186.

(39) Adun, H.; Kavaz, D.; Dagbasi, M. Review of Ternary Hybrid Nanofluid: Synthesis, Stability, Thermophysical Properties, Heat Transfer Applications, and Environmental Effects. *J. Clean. Prod.* **2021**, 328, 129525.

(40) Adun, H.; Kavaz, D.; Wole-Osho, I.; Dagbasi, M. Synthesis of Fe3O4-Al2O3-ZnO/Water Ternary Hybrid Nanofluid: Investigating the Effects of Temperature, Volume Concentration and Mixture Ratio on Specific Heat Capacity, and Development of Hybrid Machine Learning for Prediction. *J. Energy Storage* **2021**, *41*, 102947.

(41) Ma, T.; Guo, Z.; Lin, M.; Wang, Q. Recent Trends on Nanofluid Heat Transfer Machine Learning Research Applied to Renewable Energy. *Renew. Sustain. Energy Rev.* **2021**, *138* (July), 110494.

(42) Kumar, V.; Pare, A.; Tiwari, A. K.; Ghosh, S. K. Efficacy Evaluation of Oxide-MWCNT Water Hybrid Nanofluids: An Experimental and Artificial Neural Network Approach. *Colloids Surfaces A Physicochem. Eng. Asp.* **2021**, *620*, 126562.

(43) Galushchinskiy, A.; González-Gómez, R.; McCarthy, K.; Farràs, P.; Savateev, A. Progress in Development of Photocatalytic Processes for Synthesis of Fuels and Organic Compounds under Outdoor Solar Light. *Energy Fuels* **2022**, *36* (9), 4625–4639.

(44) Bond, A.; Pope, J.; Morrison-Saunders, A.; Retief, F. Taking an Environmental Ethics Perspective to Understand What We Should Expect from EIA in Terms of Biodiversity Protection. *Environ. Impact Assess. Rev.* **2021**, *86*, 106508.

(45) Loomis, J. J.; de Oliveira, C. M. R.; Dziedzic, M. Environmental Federalism in EIA Policy: A Comparative Case Study of Paraná, Brazil and California, US. *Environ. Sci. Policy* **2021**, *122*, 75–82.

(46) Fuso Nerini, F.; Tomei, J.; To, L. S.; Bisaga, I.; Parikh, P.; Black, M.; Borrion, A.; Spataru, C.; Castán Broto, V.; Anandarajah, G.; Milligan, B.; Mulugetta, Y. Mapping Synergies and Trade-Offs between Energy and the Sustainable Development Goals. *Nat. Energy* **2018**, *3*, 10.

(47) Madurai Elavarasan, R.; Pugazhendhi, R.; Jamal, T.; Dyduch, J.; Arif, M. T.; Manoj Kumar, N.; Shafiullah, G. M.; Chopra, S. S.; Nadarajah, M. Envisioning the UN Sustainable Development Goals (SDGs) through the Lens of Energy Sustainability (SDG 7) in the Post-COVID-19 World. *Appl. Energy* **2021**, *292*, 116665.

(48) Shyu, C.-W. A Framework for 'Right to Energy' to Meet UN SDG7: Policy Implications to Meet Basic Human Energy Needs, Eradicate Energy Poverty, Enhance Energy Justice, and Uphold Energy Democracy. *Energy Res. Soc. Sci.* **2021**, *79*, 102199.

(49) Fiore, M.; Magi, V.; Viggiano, A. Internal Combustion Engines Powered by Syngas: A Review. *Appl. Energy* **2020**, 276 (March), 115415.

(50) Salvia, A. L.; Leal Filho, W.; Brandli, L. L.; Griebeler, J. S. Assessing Research Trends Related to Sustainable Development Goals: Local and Global Issues. *J. Clean. Prod.* **2019**, 208, 841–849.

(51) Su, W.; Ye, Y.; Zhang, C.; Baležentis, T.; Štreimikienė, D. Sustainable Energy Development in the Major Power-Generating Countries of the European Union: The Pinch Analysis. *J. Clean. Prod.* **2020**, *256*, 120696.

(52) Sharma, P.; Said, Z.; Memon, S.; Elavarasan, R. M.; Khalid, M.; Nguyen, X. P.; Arıcı, M.; Hoang, A. T.; Nguyen, L. H. Comparative Evaluation of AI-Based Intelligent GEP and ANFIS Models in Prediction of Thermophysical Properties of Fe3O4-Coated MWCNT Hybrid Nanofluids for Potential Application in Energy Systems. *Int. J. Energy Res.* **2022**, na DOI: 10.1002/er.8010.

(53) Han, S.; Li, X.; Guo, L.; Sun, H.; Zheng, M.; Ge, W. Refining Fuel Composition of RP-3 Chemical Surrogate Models by Reactive Molecular Dynamics and Machine Learning. *Energy Fuels* **2020**, *34* (9), 11381–11394.

(54) Akhter, M. N.; Mekhilef, S.; Mokhlis, H.; Mohamed Shah, N. Review on Forecasting of Photovoltaic Power Generation Based on Machine Learning and Metaheuristic Techniques. *IET Renew. Power Gener.* **2019**, *13* (7), 1009–1023.

(55) Mosavi, A.; Salimi, M.; Faizollahzadeh Ardabili, S.; Rabczuk, T.; Shamshirband, S.; Varkonyi-Koczy, A. State of the Art of Machine Learning Models in Energy Systems, a Systematic Review. *Energies* **2019**, *12* (7), 1301.

(56) Ramezanizadeh, M.; Ahmadi, M. H.; Nazari, M. A.; Sadeghzadeh, M.; Chen, L. A Review on the Utilized Machine Learning Approaches for Modeling the Dynamic Viscosity of Nanofluids. *Renew. Sustain. Energy Rev.* **2019**, *114*, 109345.

(57) Narciso, D. A. C.; Martins, F. G. Application of Machine Learning Tools for Energy Efficiency in Industry: A Review. *Energy Reports* **2020**, *6*, 1181–1199.

(58) Shateri, M.; Sobhanigavgani, Z.; Alinasab, A.; Varamesh, A.; Hemmati-Sarapardeh, A.; Mosavi, A.; S, S. Comparative Analysis of Machine Learning Models for Nanofluids Viscosity Assessment. *Nanomaterials* **2020**, *10* (9), 176.

(59) Antonopoulos, I.; Robu, V.; Couraud, B.; Kirli, D.; Norbu, S.; Kiprakis, A.; Flynn, D.; Elizondo-Gonzalez, S.; Wattam, S. Artificial Intelligence and Machine Learning Approaches to Energy Demand-Side Response: A Systematic Review. *Renew. Sustain. Energy Rev.* **2020**, *130* (April), 109899.

(60) Fathi, S.; Srinivasan, R.; Fenner, A.; Fathi, S. Machine Learning Applications in Urban Building Energy Performance Forecasting: A Systematic Review. *Renew. Sustain. Energy Rev.* **2020**, *133* (July), 110287.

(61) Ahmad, T.; Zhang, D.; Huang, C.; Zhang, H.; Dai, N.; Song, Y.; Chen, H. Artificial Intelligence in Sustainable Energy Industry: Status Quo, Challenges and Opportunities. *J. Clean. Prod.* **2021**, *289*, 125834.

(62) Ma, T.; Guo, Z.; Lin, M.; Wang, Q. Recent Trends on Nanofluid Heat Transfer Machine Learning Research Applied to Renewable Energy. *Renew. Sustain. Energy Rev.* **2021**, *138*, 110494.

(63) Jamei, M.; Ahmadianfar, I.; Olumegbon, I. A.; Asadi, A.; Karbasi, M.; Said, Z.; Sharifpur, M.; Meyer, J. P. On the Specific Heat Capacity Estimation of Metal Oxide-Based Nanofluid for Energy Perspective - A Comprehensive Assessment of Data Analysis Techniques. *Int. Commun. Heat Mass Transfer* **2021**, *123*, 105217.

(64) Aghbashlo, M.; Peng, W.; Tabatabaei, M.; Kalogirou, S. A.; Soltanian, S.; Hosseinzadeh-Bandbafha, H.; Mahian, O.; Lam, S. S. Machine Learning Technology in Biodiesel Research: A Review. *Prog. Energy Combust. Sci.* **2021**, *85*, 100904.

(65) Tuan Hoang, A.; Nizetic, S.; Chyuan Ong, H.; Tarelko, W.; Viet Pham, V.; Hieu Le, T.; Quang Chau, M.; Phuong Nguyen, X. A Review on Application of Artificial Neural Network (ANN) for Performance and Emission Characteristics of Diesel Engine Fueled with Biodiesel-Based Fuels. *Sustain. Energy Technol. Assessments* **2021**, 47, 101416.

(66) Wang, X.; Luo, L.; Xiang, J.; Zheng, S.; Shittu, S.; Wang, Z.; Zhao, X. A Comprehensive Review on the Application of Nanofluid in Heat Pipe Based on the Machine Learning: Theory, Application and Prediction. *Renew. Sustain. Energy Rev.* **2021**, *150*, 111434.

(67) Adun, H.; Wole-Osho, I.; Okonkwo, E. C.; Kavaz, D.; Dagbasi, M. A Critical Review of Specific Heat Capacity of Hybrid Nanofluids for Thermal Energy Applications. *J. Mol. Liq.* **2021**, *340*, 116890.

(68) Maleki, A.; Haghighi, A.; Mahariq, I. Machine Learning-Based Approaches for Modeling Thermophysical Properties of Hybrid Nanofluids: A Comprehensive Review. *J. Mol. Liq.* **2021**, *322*, 114843. (69) Salameh, T.; Kumar, P. P.; Sayed, E. T.; Abdelkareem, M. A.; Rezk, H.; Olabi, A. G. Fuzzy Modeling and Particle Swarm Optimization of Al2O3/SiO2 Nanofluid. *Int. J. Thermofluids* **2021**, *10*, 100084.

(70) Subramani, S.; Govindasamy, R.; Rao, G. L. N. Predictive Correlations for NOx and Smoke Emission of DI CI Engine Fuelled with Diesel-Biodiesel-Higher Alcohol Blends-Response Surface Methodology Approach. *Fuel* **2020**, *269*, 117304.

(71) Veza, I.; Karaoglan, A. D.; Ileri, E.; Kaulani, S.A.; Tamaldin, N.; Latiff, Z.A.; Muhamad Said, M. F.; Hoang, A. T.; Yatish, K.V.; Idris, M. Grasshopper Optimization Algorithm for Diesel Engine Fuelled with Ethanol-Biodiesel-Diesel Blends. *Case Stud. Therm. Eng.* **2022**, *31*, 101817.

(72) Said, Z.; Arora, S.; Farooq, S.; Sundar, L. S.; Li, C.; Allouhi, A. Recent Advances on Improved Optical, Thermal, and Radiative Characteristics of Plasmonic Nanofluids: Academic Insights and Perspectives. *Sol. Energy Mater. Sol. Cells* **2022**, *236*, 111504.

(73) Sharma, P.; Sharma, A. K. AI-Based Prognostic Modeling and Performance Optimization of CI Engine Using Biodiesel-Diesel Blends. Int. J. Renew. Energy Resour. 2021, 11 (2), 701–708.

(74) Jamei, M.; Karbasi, M.; Mosharaf-Dehkordi, M.; Adewale Olumegbon, I.; Abualigah, L.; Said, Z.; Asadi, A. Estimating the Density of Hybrid Nanofluids for Thermal Energy Application: Application of Non-Parametric and Evolutionary Polynomial Regression Data-Intelligent Techniques. *Meas. J. Int. Meas. Confed.* **2022**, *189*, 110524.

(75) Kanti, P.; Sharma, K. V.; Yashawantha, K. M.; Jamei, M.; Said, Z. Properties of Water-Based Fly Ash-Copper Hybrid Nanofluid for Solar Energy Applications: Application of RBF Model. *Sol. Energy Mater. Sol. Cells* **2022**, *234*, 111423.

(76) Ramalho, F. M. G.; Carvalho, G. S.; Hein, P. R. G.; Napoli, A.; Wojcieszak, R.; Guilherme, L. R. G. Artificial Neural Networks To Distinguish Charcoal from Eucalyptus and Native Forests Based on Their Mineral Components. *Energy Fuels* **2020**, *34* (8), 9599–9608. (77) Cortes, C.; Vapnik, V. Support-Vector Networks. *Mach. Learn.* **1995**, *20* (3), 273–297.

(78) Sun, Y.; Zhai, C.; Zhao, Y.; Xu, J.; Cong, Y.; Zheng, Y.; Tang, W. Multifractal Analysis and Neural Network Prediction of Pore Structures in Coal Reservoirs Based on NMR T2Spectra. *Energy Fuels* **2021**, *35* (14), 11306–11318.

(79) Said, Z.; Cakmak, N. K.; Sharma, P.; Sundar, L. S.; Inayat, A.; Keklikcioglu, O.; Li, C. Synthesis, Stability, Density, Viscosity of Ethylene Glycol-Based Ternary Hybrid Nanofluids: Experimental Investigations and Model -Prediction Using Modern Machine Learning Techniques. *Powder Technol.* **2022**, 400, 117190.

(80) Peng, Y.; Parsian, A.; Khodadadi, H.; Akbari, M.; Ghani, K.; Goodarzi, M.; Bach, Q. V. Develop Optimal Network Topology of Artificial Neural Network (AONN) to Predict the Hybrid Nanofluids Thermal Conductivity According to the Empirical Data of Al2O3 - Cu Nanoparticles Dispersed in Ethylene Glycol. *Phys. A Stat. Mech. its Appl.* **2020**, *549*, 124015.

(81) Pare, A.; Ghosh, S. K. A Unique Thermal Conductivity Model (ANN) for Nanofluid Based on Experimental Study. *Powder Technol.* **2021**, 377, 429–438.

(82) Sharma, P.; Sahoo, B. B. Precise Prediction of Performance and Emission of a Waste Derived Biogas-Biodiesel Powered Dual-Fuel Engine Using Modern Ensemble Boosted Regression Tree: A Critique to Artificial Neural Network. *Fuel* **2022**, *321*, 124131.

(83) Schweidtmann, A. M.; Rittig, J. G.; Konig, A.; Grohe, M.; Mitsos, A.; Dahmen, M. Graph Neural Networks for Prediction of Fuel Ignition Quality. *Energy Fuels* **2020**, *34* (9), 11395–11407.

(84) Gholami, E.; Vaferi, B.; Ariana, M. A. Prediction of Viscosity of Several Alumina-Based Nanofluids Using Various Artificial Intelligence Paradigms - Comparison with Experimental Data and Empirical Correlations. *Powder Technol.* **2018**, *323*, 495–506.

(85) Sharma, P.; Sahoo, B. B.; Said, Z.; Hadiyanto, H.; Nguyen, X. P.; Nižetić, S.; Huang, Z.; Hoang, A. T.; Li, C. Application of Machine Learning and Box-Behnken Design in Optimizing Engine Characteristics Operated with a Dual-Fuel Mode of Algal Biodiesel and Waste-

6650

Derived Biogas. Int. J. Hydrogen Energy 2022, na DOI: 10.1016/j.ijhydene.2022.04.152.

(86) Ramezanizadeh, M.; Ahmadi, M. H.; Nazari, M. A.; Sadeghzadeh, M.; Chen, L. A Review on the Utilized Machine Learning Approaches for Modeling the Dynamic Viscosity of Nanofluids. *Renew. Sustain. Energy Rev.* **2019**, *114*, 109345.

(87) Hemmati-Sarapardeh, A.; Varamesh, A.; Husein, M. M.; Karan, K. On the Evaluation of the Viscosity of Nanofluid Systems: Modeling and Data Assessment. *Renew. Sustain. Energy Rev.* **2018**, *81*, 313–329.

(88) Bendary, A. F.; Abdelaziz, A. Y.; Ismail, M. M.; Mahmoud, K.; Lehtonen, M.; Darwish, M. M. F. Proposed Anfis Based Approach for Fault Tracking, Detection, Clearing and Rearrangement for Photovoltaic System. *Sensors* **2021**, *21* (7), 2269.

(89) Mehrabi, M.; Sharifpur, M.; Meyer, J. P. Application of the FCM-Based Neuro-Fuzzy Inference System and Genetic Algorithm-Polynomial Neural Network Approaches to Modelling the Thermal Conductivity of Alumina-Water Nanofluids. *Int. Commun. Heat Mass Transfer* **2012**, *39* (7), 971–977.

(90) Sharma, P.; Sahoo, B. B. An ANFIS-RSM Based Modeling and Multi-Objective Optimization of Syngas Powered Dual-Fuel Engine. *Int. J. Hydrogen Energy* **2022**, *47*, 19298.

(91) Thakur, A. K.; Kaviti, A. K.; Singh, R.; Gehlot, A. Modelling the Performance and Emissions of Ethanol-Gasoline Blend on a Gasoline Engine Using ANFIS. *Int. J. Ambient Energy* **2021**, 1.

(92) Betiku, E.; Osunleke, A. S.; Odude, V. O.; Bamimore, A.; Oladipo, B.; Okeleye, A. A.; Ishola, N. B. Performance Evaluation of Adaptive Neuro-Fuzzy Inference System, Artificial Neural Network and Response Surface Methodology in Modeling Biodiesel Synthesis from Palm Kernel Oil by Transesterification. *Biofuels* **2021**, *12* (3), 339–354.

(93) Karaboga, D.; Kaya, E. Adaptive Network Based Fuzzy Inference System (ANFIS) Training Approaches: A Comprehensive Survey. *Artif. Intell. Rev.* **2019**, *52* (4), 2263–2293.

(94) Ferreira, C. Gene Expression Programming: A New Adaptive Algorithm for Solving Problems. *Complex Systems* **2001**, *13*, 87–129. (95) Yilmaz, N.; Ileri, E.; Atmanli, A.; Deniz Karaoglan, A.; Okkan, U.; Sureyya Kocak, M. Predicting the Engine Performance and Exhaust Emissions of a Diesel Engine Fueled With Hazelnut Oil Methyl Ester: The Performance Comparison of Response Surface Methodology and LSSVM. *J. Energy Resour. Technol.* **2016**, *138* (5), 052206.

(96) Meybodi, M. K.; Naseri, S.; Shokrollahi, A.; Daryasafar, A. Prediction of Viscosity of Water-Based Al2O3, TiO2, SiO2, and CuO Nanofluids Using a Reliable Approach. *Chemom. Intell. Lab. Syst.* **2015**, *149*, 60–69.

(97) Luo, X.; Li, D.; Yang, Y.; Zhang, S. Spatiotemporal Traffic Flow Prediction with KNN and LSTM. *J. Adv. Transp.* **2019**, 2019, 4145353.

(98) Wang, F.; Xuan, Z.; Zhen, Z.; Li, K.; Wang, T.; Shi, M. A Day-Ahead PV Power Forecasting Method Based on LSTM-RNN Model and Time Correlation Modification under Partial Daily Pattern Prediction Framework. *Energy Convers. Manag.* **2020**, *212*, 112766.

(99) Hajiabotorabi, Z.; Kazemi, A.; Samavati, F. F.; Maalek Ghaini, F. M. Improving DWT-RNN Model via B-Spline Wavelet Multiresolution to Forecast a High-Frequency Time Series. *Expert Syst. Appl.* **2019**, *138*, 112842.

(100) Hochreiter, S.; Schmidhuber, J. Long Short-Term Memory. Neural Comput. **1997**, 9 (8), 1735–1780.

(101) Rao, G.; Huang, W.; Feng, Z.; Cong, Q. LSTM with Sentence Representations for Document-Level Sentiment Classification. *Neurocomputing* **2018**, 308 (May), 49–57.

(102) Park, J.; Sandberg, I. W. Universal Approximation Using Radial-Basis-Function Networks. *Neural Comput.* **1991**, 3 (2), 246–257.

(103) Karamichailidou, D.; Kaloutsa, V.; Alexandridis, A. Wind Turbine Power Curve Modeling Using Radial Basis Function Neural Networks and Tabu Search. *Renew. Energy* **2021**, *163*, 2137–2152.

(104) Stoffel, M.; Gulakala, R.; Barner, F.; Markert, B. Artificial Neural Networks in Structural Dynamics: A New Modular Radial Basis Function Approach vs. Convolutional and Feedforward Topologies. *Comput. Methods Appl. Mech. Eng.* **2020**, 364, 112989.

(105) Pesce, V.; Silvestrini, S.; Lavagna, M. Radial Basis Function Neural Network Aided Adaptive Extended Kalman Filter for Spacecraft Relative Navigation. *Aerosp. Sci. Technol.* **2020**, *96*, 105527. (106) Rajamoorthy, R.; Arunachalam, G.; Kasinathan, P.;

Devendiran, R.; Ahmadi, P.; Pandiyan, S.; Muthusamy, S.; Panchal, H.; Kazem, H. A; Sharma, P. A Novel Intelligent Transport System Charging Scheduling for Electric Vehicles Using Grey Wolf Optimizer and Sail Fish Optimization Algorithms Electric Vehicles Using Grey Wolf Optimizer and Sail Fish Optimization Algorithms. *Energy Sources, Part A Recover. Util. Environ. Eff.* **2022**, *44* (2), 3555–3575.

(107) Bejani, M. M.; Ghatee, M. A Systematic Review on Overfitting Control in Shallow and Deep Neural Networks. *Artif. Intell. Rev.* 2021, 54 (8), 6391–6438.

(108) Bejani, M. M.; Ghatee, M. A Systematic Review on Overfitting Control in Shallow and Deep Neural Networks. *Artif. Intell. Rev.* 2021, *54*, 6391–6438.

(109) Peng, Y.; Nagata, M. H. An Empirical Overview of Nonlinearity and Overfitting in Machine Learning Using COVID-19 Data. *Chaos, Solitons & Fractals* **2020**, *139*, 110055.

(110) Nakaura, T.; Higaki, T.; Awai, K.; Ikeda, O.; Yamashita, Y. A Primer for Understanding Radiology Articles about Machine Learning and Deep Learning. *Diagn. Interv. Imaging* **2020**, *101* (12), 765–770. (111) Sharifpur, M.; Yousefi, S.; Meyer, J. P. A New Model for Density of Nanofluids Including Nanolayer. *Int. Commun. Heat Mass Transfer* **2016**, *78*, 168–174.

(112) Raschka, S. Model Evaluation, Model Selection, and Algorithm Selection in Machine Learning. *arXiv Preprint*, arXiv:1811.12808, 2018. DOI: 10.48550/arXiv.1811.12808.

(113) Ahmed, R.; Sreeram, V.; Mishra, Y.; Arif, M. D. A Review and Evaluation of the State-of-the-Art in PV Solar Power Forecasting: Techniques and Optimization. *Renew. Sustain. Energy Rev.* **2020**, *124*, 109792.

(114) Sharma, P.; Chhillar, A.; Said, Z.; Memon, S. Exploring the Exhaust Emission and Efficiency of Algal Biodiesel PowereCompression Ignition Engine: Application of Box-Behnken and Desirability Based Multi-Objective Response Surface Methodology. *Energies* **2021**, *14*, 5968.

(115) Kelter, R. Bayesian Model Selection in the M-Open Setting — Approximate Posterior Inference and Subsampling for Efficient Large-Scale Leave-One-out Cross-Validation via the Difference Estimator. *J. Math. Psychol.* **2021**, *100*, 102474.

(116) Li, H.; Zhang, Y.; Li, C.; Zhou, Z.; Nie, X.; Chen, Y.; Cao, H.; Liu, B.; Zhang, N.; Said, Z.; Debnath, S.; Jamil, M.; Ali, H. M.; Sharma, S. Cutting Fluid Corrosion Inhibitors from Inorganic to Organic: Progress and Applications. *Korean J. Chem. Eng.* **2022**, 39 (5), 1107–1134.

(117) Goudarzi, S.; Shekaramiz, M.; Omidvar, A.; Golab, E.; Karimipour, A.; Karimipour, A. Nanoparticles Migration Due to Thermophoresis and Brownian Motion and Its Impact on Ag-MgO/ Water Hybrid Nanofluid Natural Convection. *Powder Technol.* **2020**, 375, 493–503.

(118) Ma, M.; Zhai, Y.; Yao, P.; Li, Y.; Wang, H. Synergistic Mechanism of Thermal Conductivity Enhancement and Economic Analysis of Hybrid Nanofluids. *Powder Technol.* **2020**, *373*, 702–715.

(119) Saleh, B.; Sundar, L. S. Experimental Study on Heat Transfer, Friction Factor, Entropy and Exergy Efficiency Analyses of a Corrugated Plate Heat Exchanger Using Ni/Water Nanofluids. *Int. J. Therm. Sci.* **2021**, *165*, 106935.

(120) Asadi, A.; Alarifi, I. M.; Foong, L. K. An Experimental Study on Characterization, Stability and Dynamic Viscosity of CuO-TiO2/ Water Hybrid Nanofluid. *J. Mol. Liq.* **2020**, 307, 112987.

(121) Tong, Y.; Boldoo, T.; Ham, J.; Cho, H. Improvement of Photo-Thermal Energy Conversion Performance of MWCNT/Fe3O4 Hybrid Nanofluid Compared to Fe3O4 Nanofluid. *Energy* **2020**, *196*, 117086.

(122) Li, X.; Wang, H.; Luo, B. The Thermophysical Properties and Enhanced Heat Transfer Performance of SiC-MWCNTs Hybrid Nanofluids for Car Radiator System. Colloids Surfaces A Physicochem. Eng. Asp. 2021, 612, 125968.

(123) Ghazvini, M.; Maddah, H.; Peymanfar, R.; Ahmadi, M. H.; Kumar, R. Experimental Evaluation and Artificial Neural Network Modeling of Thermal Conductivity of Water Based Nanofluid Containing Magnetic Copper Nanoparticles. *Phys. A Stat. Mech. its Appl.* **2020**, 551, 124127.

(124) Tahmooressi, H.; Kasaeian, A.; Tarokh, A.; Rezaei, R.; Hoorfar, M. Numerical Simulation of Aggregation Effect on Nanofluids Thermal Conductivity Using the Lattice Boltzmann Method. *Int. Commun. Heat Mass Transfer* **2020**, *110*, 104408.

(125) Kanti, P. K.; Sharma, K. V.; Said, Z.; Jamei, M.; Yashawantha, K. M. Experimental Investigation on Thermal Conductivity of Fly Ash Nanofluid and Fly Ash-Cu Hybrid Nanofluid: Prediction and Optimization via ANN and MGGP Model. *Part. Sci. Technol.* **2022**, 40 (2), 182–195.

(126) Jamei, M.; Olumegbon, I. A.; Karbasi, M.; Ahmadianfar, I.; Asadi, A.; Mosharaf-Dehkordi, M. On the Thermal Conductivity Assessment of Oil-Based Hybrid Nanofluids Using Extended Kalman Filter Integrated with Feed-Forward Neural Network. *Int. J. Heat Mass Transfer* **2021**, *172*, 121159.

(127) Meijuan, C. Application of ANN Technique to Predict the Thermal Conductivity of Nanofluids: A Review. J. Therm. Anal. Calorim. 2021, 145 (4), 2021–2032.

(128) Said, Z.; Sundar, L. S.; Rezk, H.; Nassef, A. M.; Chakraborty, S.; Li, C. Thermophysical Properties Using ND/Water Nanofluids: An Experimental Study, ANFIS-Based Model and Optimization. *J. Mol. Liq.* **2021**, 330, 115659.

(129) Pourrajab, R.; Ahmadianfar, I.; Jamei, M.; Behbahani, M. A Meticulous Intelligent Approach to Predict Thermal Conductivity Ratio of Hybrid Nanofluids for Heat Transfer Applications. *J. Therm. Anal. Calorim.* **2020**, na DOI: 10.1007/s10973-020-10047-9.

(130) Maleki, A.; Elahi, M.; Assad, M. E. H.; Alhuyi Nazari, M.; Safdari Shadloo, M.; Nabipour, N. Thermal Conductivity Modeling of Nanofluids with ZnO Particles by Using Approaches Based on Artificial Neural Network and MARS. J. Therm. Anal. Calorim. 2021, 143 (6), 4261–4272.

(131) Komeilibirjandi, A.; Raffiee, A. H.; Maleki, A.; Alhuyi Nazari, M.; Safdari Shadloo, M. Thermal Conductivity Prediction of Nanofluids Containing CuO Nanoparticles by Using Correlation and Artificial Neural Network. *J. Therm. Anal. Calorim.* **2020**, *139* (4), 2679–2689.

(132) Rostami, S.; Kalbasi, R.; Sina, N.; Goldanlou, A. S. Forecasting the Thermal Conductivity of a Nanofluid Using Artificial Neural Networks. J. Therm. Anal. Calorim. **2021**, 145 (4), 2095–2104.

(133) Maleki, A.; Haghighi, A.; Irandoost Shahrestani, M.; Abdelmalek, Z. Applying Different Types of Artificial Neural Network for Modeling Thermal Conductivity of Nanofluids Containing Silica Particles. J. Therm. Anal. Calorim. **2021**, 144 (4), 1613–1622.

(134) Alrashed, A. A. A. A.; Gharibdousti, M. S.; Goodarzi, M.; de Oliveira, L. R.; Safaei, M. R.; Bandarra Filho, E. P. Effects on Thermophysical Properties of Carbon Based Nanofluids: Experimental Data, Modelling Using Regression, ANFIS and ANN. *Int. J. Heat Mass Transfer* **2018**, *125*, 920–932.

(135) Razavi, R.; Sabaghmoghadam, A.; Bemani, A.; Baghban, A.; Chau, K.-w.; Salwana, E. Application of ANFIS and LSSVM Strategies for Estimating Thermal Conductivity Enhancement of Metal and Metal Oxide Based Nanofluids. *Eng. Appl. Comput. Fluid Mech.* **2019**, *13* (1), 560–578.

(136) Said, Z.; Abdelkareem, M. A.; Rezk, H.; Nassef, A. M.; Atwany, H. Z. Stability, Thermophysical and Electrical Properties of Synthesized Carbon Nanofiber and Reduced-Graphene Oxide-Based Nanofluids and Their Hybrid along with Fuzzy Modeling Approach. *Powder Technol.* **2020**, *364*, 795–809.

(137) Said, Z.; Abdelkareem, M. A.; Rezk, H.; Nassef, A. M. Fuzzy Modeling and Optimization for Experimental Thermophysical Properties of Water and Ethylene Glycol Mixture for Al2O3 and TiO2 Based Nanofluids. *Powder Technol.* **2019**, *353*, 345–358. (138) Motlagh, S. Y.; Sharifi, A.; Ahmadi, M.; Badfar, H. Presentation of New Thermal Conductivity Expression for Al 2 O 3 -Water and CuO -Water Nanofluids Using Gene Expression Programming (GEP). J. Therm. Anal. Calorim. 2019, 135 (1), 195– 206.

(139) Carrillo-Berdugo, I.; Midgley, S. D.; Grau-Crespo, R.; Zorrilla, D.; Navas, J. Understanding the Specific Heat Enhancement in Metal-Containing Nanofluids for Thermal Energy Storage: Experimental and Ab Initio Evidence for a Strong Interfacial Layering Effect. ACS Appl. Energy Mater. 2020, 3 (9), 9246–9256.

(140) Carrillo-Berdugo, I.; Grau-Crespo, R.; Zorrilla, D.; Navas, J. Interfacial Molecular Layering Enhances Specific Heat of Nanofluids: Evidence from Molecular Dynamics. J. Mol. Liq. **2021**, 325, 115217.

(141) Alade, I. O.; Abd Rahman, M. A.; Abbas, Z.; Yaakob, Y.; Saleh, T. A. Application of Support Vector Regression and Artificial Neural Network for Prediction of Specific Heat Capacity of Aqueous Nanofluids of Copper Oxide. *Sol. Energy* **2020**, *197*, 485–490.

(142) Cherecheş, E. I.; Bejan, D.; Ibanescu, C.; Danu, M.; Minea, A. A. Ionanofluids with [C2mim][CH3SO3] Ionic Liquid and Alumina Nanoparticles: An Experimental Study on Viscosity, Specific Heat and Electrical Conductivity. *Chem. Eng. Sci.* **2021**, *229*, 116140.

(143) Jamei, M.; Karbasi, M.; Adewale Olumegbon, I.; Mosharaf-Dehkordi, M.; Ahmadianfar, I.; Asadi, A. Specific Heat Capacity of Molten Salt-Based Nanofluids in Solar Thermal Applications: A Paradigm of Two Modern Ensemble Machine Learning Methods. *J. Mol. Liq.* **2021**, 335, 116434.

(144) Cherecheş, E. I.; Prado, J. I.; Ibanescu, C.; Danu, M.; Minea, A. A.; Lugo, L. Viscosity and Isobaric Specific Heat Capacity of Alumina Nanoparticle Enhanced Ionic Liquids: An Experimental Approach. J. Mol. Liq. **2020**, 317, 114020.

(145) Alade, I. O.; Abd Rahman, M. A.; Saleh, T. A. Modeling and Prediction of the Specific Heat Capacity of Al2 O3/Water Nanofluids Using Hybrid Genetic Algorithm/Support Vector Regression Model. *Nano-Structures & Nano-Objects* **2019**, *17*, 103–111.

(146) Shi, L.; Zhang, S.; Arshad, A.; Hu, Y.; He, Y.; Yan, Y. Thermo-Physical Properties Prediction of Carbon-Based Magnetic Nanofluids Based on an Artificial Neural Network. *Renew. Sustain. Energy Rev.* **2021**, *149*, 111341.

(147) Jamei, M.; Ahmadianfar, I.; Olumegbon, I. A.; Karbasi, M.; Asadi, A. On the Assessment of Specific Heat Capacity of Nanofluids for Solar Energy Applications: Application of Gaussian Process Regression (GPR) Approach. *J. Energy Storage* **2021**, *33*, 102067.

(148) Hassan, M. A.; Banerjee, D. A Soft Computing Approach for Estimating the Specific Heat Capacity of Molten Salt-Based Nanofluids. J. Mol. Liq. 2019, 281, 365–375.

(149) Murshed, S. M. S.; Estellé, P. A State of the Art Review on Viscosity of Nanofluids. *Renew. Sustain. Energy Rev.* **2017**, *76* (March), 1134–1152.

(150) Kazemi, I.; Sefid, M.; Afrand, M. A Novel Comparative Experimental Study on Rheological Behavior of Mono & Hybrid Nanofluids Concerned Graphene and Silica Nano-Powders: Characterization, Stability and Viscosity Measurements. *Powder Technol.* **2020**, *366*, 216–229.

(151) Jabbari, F.; Rajabpour, A.; Saedodin, S. Thermal Conductivity and Viscosity of Nanofluids: A Review of Recent Molecular Dynamics Studies. *Chem. Eng. Sci.* **2017**, *174*, 67–81.

(152) Moldoveanu, G. M.; Ibanescu, C.; Danu, M.; Minea, A. A. Viscosity Estimation of Al2O3, SiO2 Nanofluids and Their Hybrid: An Experimental Study. *J. Mol. Liq.* **2018**, 253, 188–196.

(153) Gholizadeh, M.; Jamei, M.; Ahmadianfar, I.; Pourrajab, R. Prediction of Nanofluids Viscosity Using Random Forest (RF) Approach. *Chemom. Intell. Lab. Syst.* **2020**, *201*, 104010.

(154) Adun, H.; Wole-Osho, I.; Okonkwo, E. C.; Ruwa, T.; Agwa, T.; Onochie, K.; Ukwu, H.; Olusola, B.; Dagbasi, M. Estimation of Thermophysical Property of Hybrid Nanofluids for Solar Thermal Applications: Implementation of Novel Optimizable Gaussian Process Regression (O-GPR) Approach for Viscosity Prediction. *Neural Comput. Appl.* **2022**, 2022, 1–22.

(155) Adun, H.; Wole-Osho, I.; Okonkwo, E. C.; Bamisile, O.; Dagbasi, M.; Abbasoglu, S. A Neural Network-Based Predictive Model for the Thermal Conductivity of Hybrid Nanofluids. *Int. Commun. Heat Mass Transfer* **2020**, *119*, 104930.

(156) Toghraie, D.; Sina, N.; Jolfaei, N. A.; Hajian, M.; Afrand, M. Designing an Artificial Neural Network (ANN) to Predict the Viscosity of Silver/Ethylene Glycol Nanofluid at Different Temperatures and Volume Fraction of Nanoparticles. *Phys. A Stat. Mech. its Appl.* **2019**, 534, 122142.

(157) Longo, G. A.; Zilio, C.; Ortombina, L.; Zigliotto, M. Application of Artificial Neural Network (ANN) for Modeling Oxide-Based Nanofluids Dynamic Viscosity. *Int. Commun. Heat Mass Transfer* **2017**, *83*, 8–14.

(158) Chen, Z.; Ashkezari, A. Z.; Tlili, I. Applying Artificial Neural Network and Curve Fitting Method to Predict the Viscosity of SAE50/MWCNTs-TiO2 Hybrid Nanolubricant. *Phys. A Stat. Mech. its Appl.* **2020**, 549, 123946.

(159) Sedaghat, F.; Yousefi, F. Synthesizes, Characterization, Measurements and Modeling Thermal Conductivity and Viscosity of Graphene Quantum Dots Nanofluids. *J. Mol. Liq.* **2019**, 278, 299–308.

(160) Baghban, A.; Jalali, A.; Shafiee, M.; Ahmadi, M. H.; Chau, K.-W. Developing an ANFIS-Based Swarm Concept Model for Estimating the Relative Viscosity of Nanofluids. *Engineering Applications of Computational Fluid Mechanics* **2019**, *13* (1), 26–39.

(161) Hemmat Esfe, M.; Tatar, A.; Ahangar, M. R. H.; Rostamian, H. A Comparison of Performance of Several Artificial Intelligence Methods for Predicting the Dynamic Viscosity of TiO2/SAE 50 Nano-Lubricant. *Phys. E Low-dimensional Syst. Nanostructures* **2018**, *96*, 85–93.

(162) Bardool, R.; Bakhtyari, A.; Esmaeilzadeh, F.; Wang, X. Nanofluid Viscosity Modeling Based on the Friction Theory. *J. Mol. Liq.* **2019**, *286*, 110923.

(163) Yan, S. R.; Kalbasi, R.; Nguyen, Q.; Karimipour, A. Rheological Behavior of Hybrid MWCNTs-TiO2/EG Nanofluid: A Comprehensive Modeling and Experimental Study. *J. Mol. Liq.* **2020**, 308, 113058.

(164) Shahsavar, A.; Khanmohammadi, S.; Karimipour, A.; Goodarzi, M. A Novel Comprehensive Experimental Study Concerned Synthesizes and Prepare Liquid Paraffin-Fe3O4Mixture to Develop Models for Both Thermal Conductivity & Viscosity: A New Approach of GMDH Type of Neural Network. *Int. J. Heat Mass Transfer* **2019**, *131*, 432–441.

(165) Ansari, H. R.; Zarei, M. J.; Sabbaghi, S.; Keshavarz, P. A New Comprehensive Model for Relative Viscosity of Various Nanofluids Using Feed-Forward Back-Propagation MLP Neural Networks. *Int. Commun. Heat Mass Transfer* **2018**, *91*, 158–164.

(166) Olumegbon, I. A.; Alade, I. O.; Sahaluddin, M.; Oyedeji, M. O.; Sa'ad, A. U. Modelling the Viscosity of Carbon-Based Nanomaterials Dispersed in Diesel Oil: A Machine Learning Approach. *Journal* of Thermal Analysis and Calorimetry **2021**, 145, 1769–1777.

(167) Sharma, P.; Said, Z.; Memon, S.; Elavarasan, R. M.; Khalid, M.; Nguyen, X. P.; Arıcı, M.; Hoang, A. T.; Nguyen, L. H. Comparative Evaluation of AI-based Intelligent GEP and ANFIS Models in Prediction of Thermophysical Properties of Fe3O4-coated MWCNT Hybrid Nanofluids for Potential Application in Energy Systems. *Int. J. Energy Res.* **2022**, na DOI: 10.1002/er.8010.

(168) Said, Z.; Hachicha, A. A.; Aberoumand, S.; Yousef, B. A. A.; Sayed, E. T.; Bellos, E. Recent Advances on Nanofluids for Low to Medium Temperature Solar Collectors: Energy, Exergy, Economic Analysis and Environmental Impact. *Prog. Energy Combust. Sci.* **2021**, *84*, 100898.

(169) Żyła, G.; Vallejo, J. P.; Lugo, L. Isobaric Heat Capacity and Density of Ethylene Glycol Based Nanofluids Containing Various Nitride Nanoparticle Types: An Experimental Study. *J. Mol. Liq.* **2018**, 261, 530–539.

(170) Ghodbane, M.; Said, Z.; Tiwari, A. K.; Syam Sundar, L.; Li, C.; Boumeddane, B. 4E (Energy, Exergy, Economic and Environmental) Investigation of LFR Using MXene Based Silicone Oil Nanofluids. Sustain. Energy Technol. Assessments 2022, 49, 101715.

(171) Ji, W.; Yang, L.; Chen, Z.; Mao, M.; Huang, J.-n. Experimental Studies and ANN Predictions on the Thermal Properties of TiO2-Ag Hybrid Nanofluids: Consideration of Temperature, Particle Loading, Ultrasonication and Storage Time. *Powder Technol.* **2021**, 388, 212–232.

(172) Montazer, E.; Salami, E.; Yarmand, H.; Chowdhury, Z. Z.; Dahari, M.; Kazi, S. N.; Badarudin, A. Development of a New Density Correlation for Carbon-Based Nanofluids Using Response Surface Methodology. *J. Therm. Anal. Calorim.* **2018**, *132* (2), 1399–1407.

(173) Mirsaeidi, A. M.; Yousefi, F. Viscosity, Thermal Conductivity and Density of Carbon Quantum Dots Nanofluids: An Experimental Investigation and Development of New Correlation Function and ANN Modeling. *Journal of Thermal Analysis and Calorimetry* **2021**, 143, 351–361.

(174) Yousefi, F.; Amoozandeh, Z. A New Model to Predict the Densities of Nanofluids Using Statistical Mechanics and Artificial Intelligent plus Principal Component Analysis. *Chin. J. Chem. Eng.* **2017**, 25 (9), 1273–1281.

(175) Singh, S.; Ghosh, S. K. A Unique Artificial Intelligence Approach and Mathematical Model to Accurately Evaluate Viscosity and Density of Several Nanofluids from Experimental Data. *Colloids Surfaces A Physicochem. Eng. Asp.* **2022**, *640*, 128389.

(176) Giwa, S. O.; Sharifpur, M.; Goodarzi, M.; Alsulami, H.; Meyer, J. P. Influence of Base Fluid, Temperature, and Concentration on the Thermophysical Properties of Hybrid Nanofluids of Alumina-Ferrofluid: Experimental Data, Modeling through Enhanced ANN, ANFIS, and Curve Fitting. J. Therm. Anal. Calorim. 2021, 143 (6), 4149–4167.

(177) Ahmed, S. F.; Saha, S. C.; Debnath, J. C.; Liu, G.; Mofijur, M.; Baniyounes, A.; Chowdhury, S.; Vo, D.-V. N. Data-Driven Modelling Techniques for Earth-Air Heat Exchangers to Reduce Energy Consumption in Buildings: A Review. *Environ. Chem. Lett.* **2021**, *19* (6), 4191–4210.

(178) Ahmed, S. F.; Khan, M. M. K.; Amanullah, M. T. O.; Rasul, M. G.; Hassan, N. M. S. A Parametric Analysis of the Cooling Performance of Vertical Earth-Air Heat Exchanger in a Subtropical Climate. *Renew. Energy* **2021**, *172*, 350–367.

(179) Hoang, A. T. Waste Heat Recovery from Diesel Engines Based on Organic Rankine Cycle. *Appl. Energy* **2018**, *231*, 138–166.

(180) Yıldız, G.; Ağbulut, Ü.; Gürel, A. E. A Review of Stability, Thermophysical Properties and Impact of Using Nanofluids on the Performance of Refrigeration Systems. *Int. J. Refrig.* **2021**, *129*, 342– 364.

(181) Omidi, M.; Farhadi, M.; Jafari, M. A Comprehensive Review on Double Pipe Heat Exchangers. *Appl. Therm. Eng.* **2017**, *110*, 1075–1090.

(182) Mohammed, H. A.; Hasan, H. A.; Wahid, M. A. Heat Transfer Enhancement of Nanofluids in a Double Pipe Heat Exchanger with Louvered Strip Inserts. *Int. Commun. Heat Mass Transfer* **2013**, *40* (1), 36–46.

(183) Ahmed, S. F.; Liu, G.; Mofijur, M.; Azad, A. K.; Hazrat, M. A.; Chu, Y.-M. Physical and Hybrid Modelling Techniques for Earth-Air Heat Exchangers in Reducing Building Energy Consumption: Performance, Applications, Progress, and Challenges. *Sol. Energy* **2021**, 216, 274–294.

(184) Bahmani, M. H.; Sheikhzadeh, G.; Zarringhalam, M.; Akbari, O. A.; Alrashed, A. A. A. A.; Shabani, G. A. S.; Goodarzi, M. Investigation of Turbulent Heat Transfer and Nanofluid Flow in a Double Pipe Heat Exchanger. *Adv. Powder Technol.* **2018**, *29* (2), 273–282.

(185) Fathi, H. I.; El-Shazly, A. H.; El-Kady, M. F.; Madih, K. Assessment of New Technique for Production Cellulose Nanocrystals from Agricultural Waste. *Mater. Sci. Forum* **2018**, *928*, 83–88.

(186) Abd Elhafez, S. E.; Abo-Zahhad, E. M.; El-Shazly, A. H.; Elkady, M. F. Experimental Investigate of Heat Transfer for Graphene/Water Nanofluid in Micro Heat Exchanger. *AIP Conf. Proc.* 2016, 1814, 020014. (187) Ali, A. Y. M.; El-Shazly, A. H.; El-Kady, M. F.; Abdelhafez, S. E. Evaluation of Surfactants on Thermo-Physical Properties of Magnesia-Oil Nanofluid. *Mater. Sci. Forum* **2018**, *928*, 106–112.

(188) Ali, A. Y. M.; El-Shazly, A. H.; El-Kady, M.F.; Fathi, H. I.; El-Marghany, M. R. Effect of Using MgO-Oil Nanofluid on the Performance of a Counter-Flow Double Pipe Heat Exchanger. *Key Eng. Mater.* **2019**, *801*, 193–198.

(189) Chun, B. H.; Kang, H. U.; Kim, S. H. Effect of Alumina Nanoparticles in the Fluid on Heat Transfer in Double-Pipe Heat Exchanger System. *Korean J. Chem. Eng.* **2008**, *25* (5), 966–971.

(190) Darzi, A. A. R.; Farhadi, M.; Sedighi, K. Heat Transfer and Flow Characteristics of AL2O3-Water Nanofluid in a Double Tube Heat Exchanger. *Int. Commun. Heat Mass Transfer* **2013**, *47*, 105–112.

(191) Mansoury, D.; Doshmanziari, F. I.; Kiani, A.; Chamkha, A. J.; Sharifpur, M. Heat Transfer and Flow Characteristics of Al2O3/ Water Nanofluid in Various Heat Exchangers: Experiments on Counter Flow. *Heat Transfer Eng.* **2020**, *41* (3), 220–234.

(192) Saeedan, M.; Solaimany Nazar, A. R.; Abbasi, Y.; Karimi, R. CFD Investigation and Neutral Network Modeling of Heat Transfer and Pressure Drop of Nanofluids in Double Pipe Helically Baffled Heat Exchanger with a 3-D Fined Tube. *Appl. Therm. Eng.* **2016**, *100*, 721–729.

(193) Sarafraz, M. M.; Hormozi, F. Intensification of Forced Convection Heat Transfer Using Biological Nanofluid in a Double-Pipe Heat Exchanger. *Exp. Therm. Fluid Sci.* **2015**, *66*, 279–289.

(194) Bahiraei, M.; Naghibzadeh, S. M.; Jamshidmofid, M. Efficacy of an Eco-Friendly Nanofluid in a Miniature Heat Exchanger Regarding to Arrangement of Silver Nanoparticles. *Energy Convers. Manag.* **2017**, *144*, 224–234.

(195) Ravi Kumar, N. T.; Bhramara, P.; Addis, B. M.; Sundar, L. S.; Singh, M. K.; Sousa, A. C. M. Heat Transfer, Friction Factor and Effectiveness Analysis of Fe3O4/Water Nanofluid Flow in a Double Pipe Heat Exchanger with Return Bend. *Int. Commun. Heat Mass Transfer* **2017**, *81*, 155–163.

(196) Shakiba, A.; Vahedi, K. Numerical Analysis of Magnetic Field Effects on Hydro-Thermal Behavior of a Magnetic Nanofluid in a Double Pipe Heat Exchanger. *J. Magn. Magn. Mater.* **2016**, *402*, 131–142.

(197) Sözen, A.; Variyenli, H. İ.; Özdemir, M. B.; Gürü, M.; Aytaç, İ. Heat Transfer Enhancement Using Alumina and Fly Ash Nanofluids in Parallel and Cross-Flow Concentric Tube Heat Exchangers. *J. Energy Inst.* **2016**, 89 (3), 414–424.

(198) Sahin, B.; Demir, A. Thermal Performance Analysis and Optimum Design Parameters of Heat Exchanger Having Perforated Pin Fins. *Energy Convers. Manag.* **2008**, *49* (6), 1684–1695.

(199) Yakut, K.; Alemdaroglu, N.; Sahin, B.; Celik, C. Optimum Design-Parameters of a Heat Exchanger Having Hexagonal Fins. *Appl. Energy* **2006**, 83 (2), 82–98.

(200) Park, K.; Choi, D. H.; Lee, K. S. Optimum Design of Plate Heat Exchanger with Staggered Pin Arrays. *Numer. Heat Transfer Part A Appl.* **2004**, 45 (4), 347–361.

(201) Galeazzo, F. C. C.; Miura, R. Y.; Gut, J. A. W.; Tadini, C. C. Experimental and Numerical Heat Transfer in a Plate Heat Exchanger. *Chem. Eng. Sci.* **2006**, *61* (21), 7133–7138.

(202) Kanaris, A. G.; Mouza, A. A.; Paras, S. V. Optimal Design of a Plate Heat Exchanger with Undulated Surfaces. *Int. J. Therm. Sci.* **2009**, 48 (6), 1184–1195.

(203) Pandya, N. S.; Shah, H.; Molana, M.; Tiwari, A. K. Heat Transfer Enhancement with Nanofluids in Plate Heat Exchangers: A Comprehensive Review. *Eur. J. Mech. B/Fluids* **2020**, *81*, 173–190.

(204) Alam, T.; Kim, M. H. A Comprehensive Review on Single Phase Heat Transfer Enhancement Techniques in Heat Exchanger Applications. *Renew. Sustain. Energy Rev.* **2018**, *81*, 813–839.

(205) Xue, Y.; Ge, Z.; Du, X.; Yang, L. On the Heat Transfer Enhancement of Plate Fin Heat Exchanger. *Energies* **2018**, *11* (6), 1398.

(206) Maradiya, C.; Vadher, J.; Agarwal, R. The Heat Transfer Enhancement Techniques and Their Thermal Performance Factor. *Beni-Suef Univ. J. Basic Appl. Sci.* **2018**, 7 (1), 1–21.

(207) Samadifar, M.; Toghraie, D. Numerical Simulation of Heat Transfer Enhancement in a Plate-Fin Heat Exchanger Using a New Type of Vortex Generators. *Appl. Therm. Eng.* 2018, 133, 671–681.
(208) Al zahrani, S.; Islam, M. S.; Saha, S. C. Heat Transfer Enhancement Investigation in a Novel Flat Plate Heat Exchanger. *Int.*

J. Therm. Sci. 2021, 161, 106763. (209) Al zahrani, S.; Islam, M. S.; Saha, S. C. Heat Transfer Enhancement of Modified Flat Plate Heat Exchanger. *Appl. Therm. Eng.* 2021, 186, 116533.

(210) Zhang, J.; Zhu, X.; Mondejar, M. E.; Haglind, F. A Review of Heat Transfer Enhancement Techniques in Plate Heat Exchangers. *Renew. Sustain. Energy Rev.* **2019**, *101*, 305–328.

(211) Pandey, S. D.; Nema, V. K. Experimental Analysis of Heat Transfer and Friction Factor of Nanofluid as a Coolant in a Corrugated Plate Heat Exchanger. *Exp. Therm. Fluid Sci.* **2012**, *38*, 248–256.

(212) Khoshvaght-Aliabadi, M.; Hormozi, F.; Zamzamian, A. Experimental Analysis of Thermal-Hydraulic Performance of Copper-Water Nanofluid Flow in Different Plate-Fin Channels. *Exp. Therm. Fluid Sci.* **2014**, *52*, 248–258.

(213) Unverdi, M.; Islamoglu, Y. Characteristics of Heat Transfer and Pressure Drop in a Chevron-Type Plate Heat Exchanger with Al2O3/Water Nanofluids. *Therm. Sci.* **2017**, *21*, 2379–2391.

(214) Tiwari, A. K.; Ghosh, P.; Sarkar, J. Heat Transfer and Pressure Drop Characteristics of CeO2/Water Nanofluid in Plate Heat Exchanger. *Appl. Therm. Eng.* **2013**, *57* (1–2), 24–32.

(215) Jokar, A.; O'Halloran, S. P. Heat Transfer and Fluid Flow Analysis of Nanofluids in Corrugated Plate Heat Exchangers Using Computational Fluid Dynamics Simulation. *J. Therm. Sci. Eng. Appl.* **2013**, 5 (1), 1–10.

(216) Sheikholeslami, M.; Ganji, D. D. Heat Transfer of Cu-Water Nanofluid Flow between Parallel Plates. *Powder Technol.* **2013**, *235*, 873–879.

(217) Ham, J.; Kim, J.; Cho, H. Theoretical Analysis of Thermal Performance in a Plate Type Liquid Heat Exchanger Using Various Nanofluids Based on LiBr Solution. *Appl. Therm. Eng.* **2016**, *108*, 1020–1032.

(218) Anoop, K.; Cox, J.; Sadr, R. Thermal Evaluation of Nanofluids in Heat Exchangers. *Int. Commun. Heat Mass Transfer* **2013**, *49*, 5–9. (219) Anoop, K.; Sadr, R.; Yu, J.; Kang, S.; Jeon, S.; Banerjee, D. Experimental Study of Forced Convective Heat Transfer of Nanofluids in a Microchannel. *Int. Commun. Heat Mass Transfer* **2012**, *39* (9), 1325–1330.

(220) Taghizadeh-Tabari, Z.; Zeinali Heris, S.; Moradi, M.; Kahani, M. The Study on Application of TiO2/Water Nanofluid in Plate Heat Exchanger of Milk Pasteurization Industries. *Renew. Sustain. Energy Rev.* 2016, *58*, 1318–1326.

(221) Sarafraz, M. M.; Hormozi, F. Heat Transfer, Pressure Drop and Fouling Studies of Multi-Walled Carbon Nanotube Nano-Fluids inside a Plate Heat Exchanger. *Exp. Therm. Fluid Sci.* **2016**, 72, 1–11. (222) Goodarzi, M.; Amiri, A.; Goodarzi, M. S.; Safaei, M. R.; Karimipour, A.; Languri, E. M.; Dahari, M. Investigation of Heat Transfer and Pressure Drop of a Counter Flow Corrugated Plate Heat Exchanger Using MWCNT Based Nanofluids. *Int. Commun. Heat Mass Transfer* **2015**, 66, 172–179.

(223) Sun, B.; Peng, C.; Zuo, R.; Yang, D.; Li, H. Investigation on the Flow and Convective Heat Transfer Characteristics of Nanofluids in the Plate Heat Exchanger. *Experimental Thermal and Fluid Science* **2016**, *76*, 75–86.

(224) Attalla, M.; Maghrabie, H. M. An Experimental Study on Heat Transfer and Fluid Flow of Rough Plate Heat Exchanger Using Al2O3/Water Nanofluid. *Exp. Heat Transfer* **2020**, *33* (3), 261–281. (225) Elias, M. M.; Saidur, R.; Ben-Mansour, R.; Hepbasli, A.; Rahim, N. A.; Jesbains, K. Heat Transfer and Pressure Drop Characteristics of a Plate Heat Exchanger Using Water Based Al2O3 Nanofluid for 30° and 60° Chevron Angles. *Heat Mass Transfer und Stoffuebertragung* **2018**, 54 (10), 2907–2916.

(226) Pourhoseini, S. H.; Naghizadeh, N.; Hoseinzadeh, H. Effect of Silver-Water Nanofluid on Heat Transfer Performance of a Plate Heat Exchanger: An Experimental and Theoretical Study. *Powder Technol.* **2018**, 332, 279–286.

(227) Pourhoseini, S. H.; Naghizadeh, N. An Experimental Study on Optimum Concentration of Silver-Water Microfluid for Enhancing Heat Transfer Performance of a Plate Heat Exchanger. *J. Taiwan Inst. Chem. Eng.* **2017**, *75*, 220–227.

(228) Behrangzade, A.; Heyhat, M. M. The Effect of Using Nano-Silver Dispersed Water Based Nanofluid as a Passive Method for Energy Efficiency Enhancement in a Plate Heat Exchanger. *Appl. Therm. Eng.* **2016**, *102*, 311–317.

(229) Sarafraz, M. M.; Nikkhah, V.; Madani, S. A.; Jafarian, M.; Hormozi, F. Low-Frequency Vibration for Fouling Mitigation and Intensification of Thermal Performance of a Plate Heat Exchanger Working with CuO/Water Nanofluid. *Applied Thermal Engineerin* **2017**, *121*, 388–399.

(230) Kumar, V.; Tiwari, A. K.; Ghosh, S. K. Effect of Chevron Angle on Heat Transfer Performance in Plate Heat Exchanger Using ZnO/Water Nanofluid. *Energy Convers. Manag.* **2016**, *118*, 142–154. (231) Suresh, S.; Venkitaraj, K. P.; Selvakumar, P.; Chandrasekar, M. Effect of Al 2O 3-Cu/Water Hybrid Nanofluid in Heat Transfer. *Exp. Therm. Fluid Sci.* **2012**, *38*, 54–60.

(232) Suresh, S.; Venkitaraj, K. P.; Hameed, M. S.; Sarangan, J. Turbulent Heat Transfer and Pressure Drop Characteristics of Dilute Water Based Al2O3-Cu Hybrid Nanofluids. *J. Nanosci. Nanotechnol.* **2014**, *14* (3), 2563–2572.

(233) Huang, D.; Wu, Z.; Sunden, B. Effects of Hybrid Nanofluid Mixture in Plate Heat Exchangers. *Exp. Therm. Fluid Sci.* 2016, 72, 190–196.

(234) Bhattad, A.; Sarkar, J.; Ghosh, P. Energetic and Exergetic Performances of Plate Heat Exchanger Using Brine-Based Hybrid Nanofluid for Milk Chilling Application. *Heat Transfer Eng.* **2020**, *41* (6–7), 522–535.

(235) Kumar, S.; Kothiyal, A. D.; Bisht, M. S.; Kumar, A. Effect of Nanofluid Flow and Protrusion Ribs on Performance in Square Channels: An Experimental Investigation. *J. Enhanc. Heat Transfer* **2019**, *26* (1), 75–100.

(236) Nazarzade, S.; Ghorbani, H. R.; Jafarpourgolroudbary, H. Synthesis, Preparation and the Experimental Study of Silver/Water Nanofluid to Optimize Convective Heat Transfer in a Shell and Tube Heat Exchanger. *Inorg. Nano-Metal Chem.* **2019**, *49* (6), 173–176.

(237) Said, Z.; Rahman, S. M. A.; El Haj Assad, M.; Alami, A. H. Heat Transfer Enhancement and Life Cycle Analysis of a Shell-and-Tube Heat Exchanger Using Stable CuO/Water Nanofluid. *Sustain. Energy Technol. Assessments* **2019**, *31*, 306–317.

(238) Somasekhar, K.; Malleswara Rao, K. N. D.; Sankararao, V.; Mohammed, R.; Veerendra, M.; Venkateswararao, T. A CFD Investigation of Heat Transfer Enhancement of Shell and Tube Heat Exchanger Using Al2o3-Water Nanofluid. *Mater. Today Proc.* **2018**, 5 (1), 1057–1062.

(239) Ullah, M. R.; Ishtiaq, T. M.; Mamun, M. A. H. Heat Transfer Enhancement in Shell and Tube Heat Exchanger by Using Al2O3/ Water and TiO2/Water Nanofluid. *AIP Conf. Proc.* **2018**, *2121* (July), 070018.

(240) Esfahani, M. R.; Languri, E. M. Exergy Analysis of a Shell-and-Tube Heat Exchanger Using Graphene Oxide Nanofluids. *Exp. Therm. Fluid Sci.* **2017**, *83*, 100–106.

(241) Lotfi, R.; Rashidi, A. M.; Amrollahi, A. Experimental Study on the Heat Transfer Enhancement of MWNT-Water Nanofluid in a Shell and Tube Heat Exchanger. *Int. Commun. Heat Mass Transfer* **2012**, 39 (1), 108–111.

(242) Said, Z.; Sharma, P.; Tiwari, A. K.; Le, V. V.; Huang, Z.; Bui, V. G.; Hoang, A. T. Application of Novel Framework Based on Ensemble Boosted Regression Trees and Gaussian Process Regression in Modelling Thermal Performance of Small-Scale Organic Rankine

Cycle (ORC) Using Hybrid Nanofluid. J. Clean. Prod. 2022, 360, 132194.

(243) Xie, G. N.; Wang, Q. W.; Zeng, M.; Luo, L. Q. Heat Transfer Analysis for Shell-and-Tube Heat Exchangers with Experimental Data by Artificial Neural Networks Approach. *Appl. Therm. Eng.* **2007**, *27* (5–6), 1096–1104.

(244) Wang, Q.; Xie, G.; Zeng, M.; Luo, L. Prediction of Heat Transfer Rates for Shell-and-Tube Heat Exchangers by Artificial Neural Networks Approach. J. Therm. Sci. 2006, 15 (3), 257–262.

(245) Xie, G.; Sunden, B.; Wang, Q.; Tang, L. Performance Predictions of Laminar and Turbulent Heat Transfer and Fluid Flow of Heat Exchangers Having Large Tube-Diameter and Large Tube-Row by Artificial Neural Networks. *Int. J. Heat Mass Transfer* **2009**, *52* (11–12), 2484–2497.

(246) Hojjat, M. Nanofluids as Coolant in a Shell and Tube Heat Exchanger: ANN Modeling and Multi-Objective Optimization. *Appl. Math. Comput.* **2020**, 365, 124710.

(247) Maddah, H.; Ghasemi, N.; Keyvani, B.; Cheraghali, R. Experimental and Numerical Study of Nanofluid in Heat Exchanger Fitted by Modified Twisted Tape: Exergy Analysis and ANN Prediction Model. *Heat Mass Transfer und Stoffuebertragung* **201**7, 53 (4), 1413–1423.

(248) Bahiraei, M.; Khosravi, R.; Heshmatian, S. Assessment and Optimization of Hydrothermal Characteristics for a Non-Newtonian Nanofluid Flow within Miniaturized Concentric-Tube Heat Exchanger Considering Designer's Viewpoint. *Appl. Therm. Eng.* **2017**, *123*, 266–276.

(249) Safikhani, H.; Abbassi, A.; Khalkhali, A.; Kalteh, M. Multi-Objective Optimization of Nanofluid Flow in Flat Tubes Using CFD, Artificial Neural Networks and Genetic Algorithms. *Adv. Powder Technol.* **2014**, 25 (5), 1608–1617.

(250) Zdaniuk, G. J.; Chamra, L. M.; Keith Walters, D. Correlating Heat Transfer and Friction in Helically-Finned Tubes Using Artificial Neural Networks. *Int. J. Heat Mass Transfer* **2007**, *50* (23–24), 4713– 4723.

(251) Wijayasekara, D.; Manic, M.; Sabharwall, P.; Utgikar, V. Optimal Artificial Neural Network Architecture Selection for Performance Prediction of Compact Heat Exchanger with the EBaLM-OTR Technique. *Nucl. Eng. Des.* **2011**, *241* (7), 2549–2557.

(252) Nižetić, S.; Jurčević, M.; Čoko, D.; Arıcı, M.; Hoang, A. T. Implementation of Phase Change Materials for Thermal Regulation of Photovoltaic Thermal Systems: Comprehensive Analysis of Design Approaches. *Energy* **2021**, *228*, 120546.

(253) Kumpanalaisatit, M.; Setthapun, W.; Sintuya, H.; Jansri, S. N. Efficiency Improvement of Ground-Mounted Solar Power Generation in Agrivoltaic System by Cultivation of Bok Choy (Brassica Rapa Subsp. Chinensis L.) Under the Panels. *Int. J. Renew. Energy Dev.* **2022**, *11* (1), 103–110.

(254) Chitsomboon, T.; Koonsrisook, A.; Le, T. H.; Hoang, A. T. Experimental Investigation of Solar Energy-Based Water Distillation Using Inclined Metal Tubes as Collector and Condenser. *Energy Sources, Part A Recover. Util. Environ. Eff.* **2021**, 1–17.

(255) Ahmed, O. K.; Daoud, R. W.; Bawa, S. M.; Ahmed, A. H. Optimization of PV/T Solar Water Collector Based on Fuzzy Logic Control. *Int. J. Renew. Energy Dev.* **2020**, *9* (2), 303–310.

(256) Saleh, U. A.; Johar, M. A.; Jumaat, S. A. B.; Rejab, M. N.; Wan Jamaludin, W. A. Evaluation of a PV-TEG Hybrid System Configuration for an Improved Energy Output: A Review. *Int. J. Renew. Energy Dev.* **2020**, *10* (2), 385–400.

(257) Guno, C. S.; Agaton, C. B.; Villanueva, R. O.; Villanueva, R. O. Optimal Investment Strategy for Solar PV Integration in Residential Buildings: A Case Study in The Philippines. *Int. J. Renew. Energy Dev* **2020**, *10*, 79–89.

(258) Senthil, R. Effect of Fluid Flow Direction on Charging of Multitube Thermal Energy Storage for Flat Plate Solar Collectors. *Int. J. Renew. Energy Dev.* **2020**, *10* (2), 365–371.

(259) Shahabuddin, M.; Alim, M. A.; Alam, T.; Mofijur, M.; Ahmed, S. F.; Perkins, G. A Critical Review on the Development and

Challenges of Concentrated Solar Power Technologies. Sustain. Energy Technol. Assessments 2021, 47, 101434.

(260) Wang, L.; Kisi, O.; Zounemat-Kermani, M.; Salazar, G. A.; Zhu, Z.; Gong, W. Solar Radiation Prediction Using Different Techniques: Model Evaluation and Comparison. *Renew. Sustain. Energy Rev.* **2016**, *61*, 384–397.

(261) Gupta, P. R.; Tiwari, A. K.; Said, Z. Solar Organic Rankine Cycle and Its Poly-Generation Applications - A Review. *Sustain. Energy Technol. Assessments* **2022**, *49*, 101732.

(262) Voyant, C.; Notton, G.; Kalogirou, S.; Nivet, M. L.; Paoli, C.; Motte, F.; Fouilloy, A. Machine Learning Methods for Solar Radiation Forecasting: A Review. *Renew. Energy* **2017**, *105*, 569–582.

(263) Ağbulut, Ü.; Gürel, A. E.; Biçen, Y. Prediction of Daily Global Solar Radiation Using Different Machine Learning Algorithms: Evaluation and Comparison. *Renew. Sustain. Energy Rev.* **2021**, *135*, 110114.

(264) Gürel, A. E.; Ağbulut, Ü.; Biçen, Y. Assessment of Machine Learning, Time Series, Response Surface Methodology and Empirical Models in Prediction of Global Solar Radiation. *J. Clean. Prod.* **2020**, 277, 122353.

(265) Meenal, R.; Selvakumar, A. I. Assessment of SVM, Empirical and ANN Based Solar Radiation Prediction Models with Most Influencing Input Parameters. *Renew. Energy* **2018**, *121*, 324–343.

(266) Fan, J.; Wu, L.; Zhang, F.; Cai, H.; Wang, X.; Lu, X.; Xiang, Y. Evaluating the Effect of Air Pollution on Global and Diffuse Solar Radiation Prediction Using Support Vector Machine Modeling Based on Sunshine Duration and Air Temperature. *Renew. Sustain. Energy Rev.* **2018**, *94*, 732–747.

(267) Eseye, A. T.; Zhang, J.; Zheng, D. Short-Term Photovoltaic Solar Power Forecasting Using a Hybrid Wavelet-PSO-SVM Model Based on SCADA and Meteorological Information. *Renew. Energy* **2018**, *118*, 357–367.

(268) Fan, J.; Wu, L.; Ma, X.; Zhou, H.; Zhang, F. Hybrid Support Vector Machines with Heuristic Algorithms for Prediction of Daily Diffuse Solar Radiation in Air-Polluted Regions. *Renew. Energy* **2020**, *145*, 2034–2045.

(269) Lee, J.; Wang, W.; Harrou, F.; Sun, Y. Reliable Solar Irradiance Prediction Using Ensemble Learning-Based Models: A Comparative Study. *Energy Convers. Manag.* **2020**, *208*, 112582.

(270) Cao, Y.; Kamrani, E.; Mirzaei, S.; Khandakar, A.; Vaferi, B. Electrical Efficiency of the Photovoltaic/Thermal Collectors Cooled by Nanofluids: Machine Learning Simulation and Optimization by Evolutionary Algorithm. *Energy Reports* **2022**, *8*, 24–36.

(271) Ranjan Parida, D.; Dani, N.; Basu, S. Data-Driven Analysis of Molten-Salt Nanofluids for Specific Heat Enhancement Using Unsupervised Machine Learning Methodologies. *Sol. Energy* **2021**, 227, 447–456.

(272) Ebrahimi-Moghadam, A.; Mohseni-Gharyehsafa, B.; Farzaneh-Gord, M. Using Artificial Neural Network and Quadratic Algorithm for Minimizing Entropy Generation of Al2O3-EG/W Nanofluid Flow inside Parabolic Trough Solar Collector. *Renew. Energy* **2018**, *129*, 473–485.

(273) Delfani, S.; Esmaeili, M.; Karami, M. Application of Artificial Neural Network for Performance Prediction of a Nanofluid-Based Direct Absorption Solar Collector. *Sustain. Energy Technol. Assessments* **2019**, *36*, 100559.

(274) Bahiraei, M.; Nazari, S.; Safarzadeh, H. Modeling of Energy Efficiency for a Solar Still Fitted with Thermoelectric Modules by ANFIS and PSO-Enhanced Neural Network: A Nanofluid Application. *Powder Technol.* **2021**, *385*, 185–198.

(275) Akella, A. K.; Saini, R. P.; Sharma, M. P. Social, Economical and Environmental Impacts of Renewable Energy Systems. *Renew. Energy* **2009**, *34* (2), 390–396.

(276) Hondo, H.; Moriizumi, Y. Employment Creation Potential of Renewable Power Generation Technologies: A Life Cycle Approach. *Renew. Sustain. Energy Rev.* **201**7, *79*, 128–136.

(277) Zheng, B.; Xu, J.; Ni, T.; Li, M. Geothermal Energy Utilization Trends from a Technological Paradigm Perspective. *Renew. Energy* **2015**, 77, 430–441. (278) Eyerer, S.; Schifflechner, C.; Hofbauer, S.; Bauer, W.; Wieland, C.; Spliethoff, H. Combined Heat and Power from Hydrothermal Geothermal Resources in Germany: An Assessment of the Potential. *Renew. Sustain. Energy Rev.* **2020**, *120* (\), 109661.

(279) Xia, L.; Zhang, Y. An Overview of World Geothermal Power Generation and a Case Study on China—The Resource and Market Perspective. *Renew. Sustain. Energy Rev.* **2019**, *112* (May), 411–423. (280) Dickson, M. H., Fanelli, M. *Geothermal Energy: Utilization and*

Technology, 1st ed.; Taylor & Francis Group, 2003. (281) Sircar, A.; Solanki, K.; Bist, N.; Yadav, K. Enhanced

Geothermal Systems-Promises and Challenges. Int. J. Renew. Energy Dev. 2022, 11 (2), 333–346.

(282) Zarrouk, S. J.; Moon, H. Efficiency of Geothermal Power Plants: A Worldwide Review. *Geothermics* **2014**, *51*, 142–153.

(283) Prananto, L. A.; Juangsa, F. B.; Iqbal, R. M.; Aziz, M.; Soelaiman, T. A. F. Dry Steam Cycle Application for Excess Steam Utilization: Kamojang Geothermal Power Plant Case Study. *Renew. Energy* **2018**, *117*, 157–165.

(284) Chen, H.; Goswami, D. Y.; Stefanakos, E. K. A Review of Thermodynamic Cycles and Working Fluids for the Conversion of Low-Grade Heat. *Renew. Sustain. Energy Rev.* **2010**, *14* (9), 3059–3067.

(285) Lee, I.; Tester, J. W.; You, F. Systems Analysis, Design, and Optimization of Geothermal Energy Systems for Power Production and Polygeneration: State-of-the-Art and Future Challenges. *Renew. Sustain. Energy Rev.* **2019**, *109* (April), 551–577.

(286) Hassan, F.; Jamil, F.; Hussain, A.; Ali, H. M.; Janjua, M. M.; Khushnood, S.; Farhan, M.; Altaf, K.; Said, Z.; Li, C. Recent Advancements in Latent Heat Phase Change Materials and Their Applications for Thermal Energy Storage and Buildings: A State of the Art Review. *Sustain. Energy Technol. Assessments* **2022**, *49*, 101646.

(287) Daneshipour, M.; Rafee, R. Nanofluids as the Circuit Fluids of the Geothermal Borehole Heat Exchangers. *Int. Commun. Heat Mass Transfer* **2017**, *81*, 34–41.

(288) Diglio, G.; Roselli, C.; Sasso, M.; Jawali Channabasappa, U. Borehole Heat Exchanger with Nanofluids as Heat Carrier. *Geothermics* **2018**, *72*, 112–123.

(289) Narei, H.; Ghasempour, R.; Noorollahi, Y. The Effect of Employing Nanofluid on Reducing the Bore Length of a Vertical Ground-Source Heat Pump. *Energy Convers. Manag.* **2016**, *123*, 581–591.

(290) Ramezanizadeh, M.; Alhuyi Nazari, M. Modeling Thermal Conductivity of Ag/Water Nanofluid by Applying a Mathematical Correlation and Artificial Neural Network. *Int. J. Low-Carbon Technol.* **2019**, *14* (4), 468–474.

(291) Hemmat Esfe, M.; Razi, P.; Hajmohammad, M. H.; Rostamian, S. H.; Sarsam, W. S.; Abbasian Arani, A. A.; Dahari, M. Optimization, Modeling and Accurate Prediction of Thermal Conductivity and Dynamic Viscosity of Stabilized Ethylene Glycol and Water Mixture Al2O3 Nanofluids by NSGA-II Using ANN. *Int. Commun. Heat Mass Transfer* **201**7, *82*, 154–160.

(292) Tiwari, A. K.; Javed, S.; Oztop, H. F.; Said, Z.; Pandya, N. S. Experimental and Numerical Investigation on the Thermal Performance of Triple Tube Heat Exchanger Equipped with Different Inserts with WO3/Water Nanofluid under Turbulent Condition. *Int. J. Therm. Sci.* **2021**, *164*, 106861.

(293) Devlin, J.; Li, K.; Higgins, P.; Foley, A. Gas Generation and Wind Power: A Review of Unlikely Allies in the United Kingdom and Ireland. *Renew. Sustain. Energy Rev.* **2017**, *70*, 757–768.

(294) Adem Cakmakcı, B.; Huner, E. Evaluation of Wind Energy Potential: A Case Study. *Energy Sources, Part A Recover. Util. Environ. Eff.* **2022**, 44 (1), 834–852.

(295) Wang, Y.; Hu, Q.; Li, L.; Foley, A. M.; Srinivasan, D. Approaches to Wind Power Curve Modeling: A Review and Discussion. *Renew. Sustain. Energy Rev.* **2019**, *116*, 109422.

(296) Hemmati, R.; Nosratabadi, S. M.; Mehrjerdi, H.; Bornapour, M. Coordination of Thermal/Wind Energies in Power-to-Gas Process for Cost/Pollution Abatement Considering Wind Energy Recovery.

Review

Energy Sources, Part A Recover. Util. Environ. Eff. 2022, 44 (1), 632–649.

(297) Chen, W.-H.; Wang, J.-S.; Chang, M.-H.; Tuan Hoang, A.; Shiung Lam, S.; Kwon, E. E.; Ashokkumar, V. Optimization of a Vertical Axis Wind Turbine with a Deflector under Unsteady Wind Conditions via Taguchi and Neural Network Applications. *Energy Convers. Manag.* **2022**, 254, 115209.

(298) Raj, C. P. C.; Surendran, S. A. A.; Amjathkhan, B.; Metilda, J. A. B.; Devaraj, S. E.; Aristotle, E. Nano Fluids for Improving Efficiency in Wind Turbine Cooling System. *Adv. Mater. Res.* **2014**, 984–985, 784–791.

(299) De Risi, A.; Milanese, M.; Colangelo, G.; Laforgia, D. High Efficiency Nanofluid Cooling System for Wind Turbines. *Therm. Sci.* **2014**, *18* (2), 543–554.

(300) Álvarez-Regueiro, E.; Vallejo, J. P.; Fernández-Seara, J.; Fernández, J.; Lugo, L. Experimental Convection Heat Transfer Analysis of a Nano-Enhanced Industrial Coolant. *Nanomaterials* **2019**, *9* (2), 267.

(301) Adun, H.; Kavaz, D.; Dagbasi, M.; Umar, H.; Wole-Osho, I. An Experimental Investigation of Thermal Conductivity and Dynamic Viscosity of Al2O3-ZnO-Fe3O4 Ternary Hybrid Nanofluid and Development of Machine Learning Model. *Powder Technol.* **2021**, 394, 1121–1140.

(302) Raizada, A.; Yadav, S. Hydrogen Rich Syngas Production from Food Waste via an Integrated Two-Stage Process of in-Situ Steam Gasification after Fast Pyrolysis. *Energy Sources, Part A Recover. Util. Environ. Eff.* **2022**, 44 (1), 1608–1619.

(303) Yustanti, E.; Muharman, A.; Mursito, A. T. The Effect of Wood Tar and Molasses Composition on Calorific Value and Compressive Strength in Bio-Coke Briquetting. *Int. J. Renew. Energy Dev.* **2022**, *11* (3), 600–607.

(304) Sharma, P.; Sharma, A. K. Application of Response Surface Methodology for Optimization of Fuel Injection Parameters of a Dual Fuel Engine Fuelled with Producer Gas- Biodiesel Blends. *Energy Sources, Part A Recover. Util. Environ. Eff.* **2021**, 00 (00), 1–18.

(305) Toklu, E. Biomass Energy Potential and Utilization in Turkey. *Renew. Energy* **2017**, *107*, 235–244.

(306) Handaya, H.; Susanto, H.; Indrawan, D.; Marimin, M. Supply and Demand Characteristics of Palm Kernel Shell as a Renewable Energy Source for Industries. *Int. J. Renew. Energy Dev.* **2022**, *11* (2), 481–490.

(307) Tuan Hoang, A.; Nizetic, S.; Olcer, A. I.; Chyuan Ong, H. Synthesis Pathway and Combustion Mechanism of a Sustainable Biofuel 2,5-Dimethylfuran: Progress and Prospective. *Fuel* **2021**, *286*, 119337.

(308) Hoang, A. T.; Huang, Z. H.; Nižetić, S.; Pandey, A.; Nguyen, X. P.; Luque, R.; Ong, H. C.; Said, Z.; Le, T. H.; Pham, V. V. Characteristics of Hydrogen Production from Steam Gasification of Plant-Originated Lignocellulosic Biomass and Its Prospects in Vietnam. *Int. J. Hydrogen Energy* **2022**, *47* (7), 4394–4425.

(309) Ahmed, S. F.; Mofijur, M.; Chowdhury, S. N.; Nahrin, M.; Rafa, N.; Chowdhury, A. T.; Nuzhat, S.; Ong, H. C. Pathways of Lignocellulosic Biomass Deconstruction for Biofuel and Value-Added Products Production. *Fuel* **2022**, *318*, 123618.

(310) Hoang, A. T.; Ong, H. C.; Fattah, I. M. R.; Chong, C. T.; Cheng, C. K.; Sakthivel, R.; Ok, Y. S. Progress on the Lignocellulosic Biomass Pyrolysis for Biofuel Production toward Environmental Sustainability. *Fuel Process. Technol.* **2021**, 223, 106997.

(311) Das, S.; Kashyap, D.; Kalita, P.; Kulkarni, V.; Itaya, Y. Clean Gaseous Fuel Application in Diesel Engine: A Sustainable Option for Rural Electrification in India. *Renewable and Sustainable Energy Reviews.* **2020**, *117*, 109485.

(312) Cen, K.; Zhang, J.; Chen, D.; Chen, F.; Zhang, Y.; Ma, H. Comparative Study of the Fuel Quality and Torrefaction Performance of Biomass and Its Molded Pellets: Effects of Temperature and Residence Time. *Energy Sources, Part A Recover. Util. Environ. Eff.* **2020**, 1-10.

(313) Ortiz-Alvarez, M.; Piloto-Rodríguez, R.; Pohl, S. Predicting Bio-Oil Yield Obtained from Lignocellulosic Biomass Pyrolysis Using Artificial Neural Networks. Energy Sources, Part A Recover. Util. Environ. Eff. 2022, 44 (1), 247–256.

(314) Chen, W.-H.; Hoang, A. T.; Nižetić, S.; Pandey, A.; Cheng, C. K.; Luque, R.; Ong, H. C.; Thomas, S.; Nguyen, X. P. Biomass-Derived Biochar: From Production to Application in Removing Heavy Metal-Contaminated Water. *Process Saf. Environ. Prot.* 2022, 160 (April), 704–733.

(315) Littlejohns, J. V.; Butler, J.; Luque, L.; Kannangara, M.; Totolo, S. Analysis of the Performance of an Integrated Small-Scale Biomass Gasification System in a Canadian Context. *Biomass Convers. Biorefinery* **2020**, *10* (2), 311–323.

(316) Sharma, P. Artificial Intelligence-based Model Prediction of Biodiesel-fueled Engine Performance and Emission Characteristics: A Comparative Evaluation of Gene Expression Programming and Artificial Neural Network. *Heat Transfer* **2021**, *50* (6), 5563–5587.

(317) Sharma, P. Gene Expression Programming-Based Model Prediction of Performance and Emission Characteristics of a Diesel Engine Fueled with Linseed Oil Biodiesel/Diesel Blends: An Artificial Intelligence Approach. *Energy Sources, Part A Recover. Util. Environ. Eff.* **2020**, 1.

(318) Safarian, S.; Ebrahimi Saryazdi, S. M.; Unnthorsson, R.; Richter, C. Artificial Neural Network Integrated with Thermodynamic Equilibrium Modeling of Downdraft Biomass Gasification-Power Production Plant. *Energy* **2020**, *213*, 118800.

(319) Mutlu, A. Y.; Yucel, O. An Artificial Intelligence Based Approach to Predicting Syngas Composition for Downdraft Biomass Gasification. *Energy* **2018**, *165*, 895–901.

(320) Baruah, D.; Baruah, D. C.; Hazarika, M. K. Artificial Neural Network Based Modeling of Biomass Gasification in Fixed Bed Downdraft Gasifiers. *Biomass and Bioenergy* **2017**, *98*, 264–271.

(321) Kargbo, H. O.; Zhang, J.; Phan, A. N. Optimisation of Two-Stage Biomass Gasification for Hydrogen Production via Artificial Neural Network. *Appl. Energy* **2021**, *302*, 117567.

(322) Aghbashlo, M.; Tabatabaei, M.; Nadian, M. H.; Davoodnia, V.; Soltanian, S. Prognostication of Lignocellulosic Biomass Pyrolysis Behavior Using ANFIS Model Tuned by PSO Algorithm. *Fuel* **2019**, 253, 189–198.

(323) Elmaz, F.; Yücel, Ö.; Mutlu, A. Y. Predictive Modeling of Biomass Gasification with Machine Learning-Based Regression Methods. *Energy* **2020**, *191*, 116541.

(324) Patel, J. R.; Rathod, M. K.; Elavarasan, R. M.; Said, Z. Influence of Longitudinal Fin Arrangement on the Melting and Solidification inside the Triplex Tube Latent Heat Thermal Storage System. *J. Energy Storage* **2022**, *46*, 103778.

(325) Masoumi, A.; Ghassem-zadeh, S.; Hosseini, S. H.; Ghavidel, B. Z. Application of Neural Network and Weighted Improved PSO for Uncertainty Modeling and Optimal Allocating of Renewable Energies along with Battery Energy Storage. *Appl. Soft Comput.* **2020**, *88*, 105979.

(326) Gayen, P. K.; Jana, A. An ANFIS Based Improved Control Action for Single Phase Utility or Micro-Grid Connected Battery Energy Storage System. J. Clean. Prod. 2017, 164, 1034–1049.

(327) Tang, Y.; Huang, Y.; Wang, H.; Wang, C.; Guo, q.; Yao, W. Framework for Artificial Intelligence Analysis in Large-Scale Power Grids Based on Digital Simulation. *CSEE J. Power Energy Syst.* **2018**, *4* (4), 459–468.

(328) Guo, Y.; Tan, Z.; Chen, H.; Li, G.; Wang, J.; Huang, R.; Liu, J.; Ahmad, T. Deep Learning-Based Fault Diagnosis of Variable Refrigerant Flow Air-Conditioning System for Building Energy Saving. *Appl. Energy* **2018**, *225*, 732–745.

(329) Zhao, Y.; Zhang, C.; Zhang, Y.; Wang, Z.; Li, J. A Review of Data Mining Technologies in Building Energy Systems: Load Prediction, Pattern Identification, Fault Detection and Diagnosis. *Energy Built Environ.* **2020**, *1* (2), 149–164.

(330) Rehman, H. ur; Korvola, T.; Abdurafikov, R.; Laakko, T.; Hasan, A.; Reda, F. Data Analysis of a Monitored Building Using Machine Learning and Optimization of Integrated Photovoltaic Panel, Battery and Electric Vehicles in a Central European Climatic Condition. *Energy Convers. Manag.* **2020**, *221*, 113206. (331) Ugwu, J.; Odo, K. C.; Oluka, L. O.; Salami, K. O. A Systematic Review on the Renewable Energy Development, Policies and Challenges in Nigeria with an International Perspective and Public Opinions. *Int. J. Renew. Energy Dev.* **2022**, *11* (1), 287–308.

pubs.acs.org/EF

(332) Nguyen, X. P.; Le, N. D.; Pham, V. V.; Huynh, T. T.; Dong, V. H.; Hoang, A. T. Mission, Challenges, and Prospects of Renewable Energy Development in Vietnam. *Energy Sources, Part A Recover. Util. Environ. Eff.* **2021**, 1–13.

(333) Yoro, K. O.; Daramola, M. O.; Sekoai, P. T.; Wilson, U. N.; Eterigho-Ikelegbe, O. Update on Current Approaches, Challenges, and Prospects of Modeling and Simulation in Renewable and Sustainable Energy Systems. *Renew. Sustain. Energy Rev.* 2021, 150, 111506.

(334) Franco, A.; Salza, P. Strategies for Optimal Penetration of Intermittent Renewables in Complex Energy Systems Based on Techno-Operational Objectives. *Renew. Energy* **2011**, *36* (2), 743–753.

(335) Sharma, P. Prediction-Optimization of the Effects of Di-Tert Butyl Peroxide-Biodiesel Blends on Engine Performance and Emissions Using Multi-Objective Response Surface Methodology (MORSM). J. Energy Resour. Technol. **2022**, 144, 072301.

(336) Penalba, M.; Ringwood, J. V. Systematic Complexity Reduction of Wave-to-Wire Models for Wave Energy System Design. *Ocean Eng.* **2020**, *217*, 107651.

(337) Sinha, S.; Chandel, S. S. Review of Software Tools for Hybrid Renewable Energy Systems. *Renew. Sustain. Energy Rev.* 2014, 32, 192–205.

(338) Kotzur, L.; Nolting, L.; Hoffmann, M.; Groß, T.; Smolenko, A.; Priesmann, J.; Büsing, H.; Beer, R.; Kullmann, F.; Singh, B.; Praktiknjo, A.; Stolten, D.; Robinius, M. A Modeler's Guide to Handle Complexity in Energy Systems Optimization. *Adv. Appl. Energy* **2021**, *4*, 100063.

(339) Astobiza, A. M.; Toboso, M.; Aparicio, M.; Lopez, D. AI Ethics for Sustainable Development Goals. *IEEE Technol. Soc. Mag.* **2021**, 40 (2), 66–71.

(340) Sharma, P.; Sharma, A. K. Statistical and Continuous Wavelet Transformation Based Analysis of Combustion Instabilities in a Biodiesel Fuelled Compression Ignition Engine. *J. Energy Resour. Technol.* **2022**, *144*, 032304.

Recommended by ACS

Experimental Investigation of Spout Deflection in a Rectangular Spouted Bed by the PIV Method

Yuanhe Yue, Yansong Shen, et al. JUNE 30, 2020 INDUSTRIAL & ENGINEERING CHEMISTRY RESEARCH

READ 🗹

CPFD Simulation of Hydrodynamics, Heat Transfer, and Reactions in a Downer Reactor for Coal Pyrolysis with Binary Particles

Yingya Wu, Xingying Lan, et al. NOVEMBER 19, 2019 ENERGY & FUELS

READ 🗹

Numerical Simulation of Combustion Characteristics in the Dome Zone of the COREX Melter-Gasifier: The Effect of Rising Gas

Yifan Hu, Yansong Shen, et al. DECEMBER 14, 2021 INDUSTRIAL & ENGINEERING CHEMISTRY RESEARCH

READ 🗹

Second International Symposium on Biomass/Wastes Energy and Environment (BEE 2019), Tianjin, China, May 23–26, 2019

Guanyi Chen and Robert S. Weber JANUARY 31, 2020 ENERGY & FUELS

READ 🗹

Get More Suggestions >